# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

## SCHEDULE 13D (Rule 13d-101)

## INFORMATION TO BE INCLUDED IN STATEMENTS PURSUANT TO RULE 13d-1(a) AND AMENDMENTS THERETO FILED PURSUANT TO RULE 13d-2(a)

Under the Securities Exchange Act of 1934 (Amendment No.)\*

## **FUSION FUEL GREEN PLC**

(Name of Issuer)

Class A Ordinary Shares, \$0.0001 par value per share (Title of Class of Securities)

> G3R25D 118 (CUSIP Number)

David Alan Miller, Esq. Jeffrey M. Gallant, Esq. Graubard Miller 405 Lexington Avenue, 11th Floor New York, NY 10174 (212) 818-8877

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

#### December 10, 2020 (Date of Event Which Requires Filing of This Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of Rule 13d-1(e), Rule 13d-1(f) or Rule 13d-1(g), check the following box.  $\Box$ 

*Note:* Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See Rule 13d-7(b) for other parties to whom copies are to be sent.

\* The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

	1		
1 Names of Reporting Persons			
	Fusion We		
2			ate Box if a Member of a Group
	(a) 🗆 (b) 🛛		
3	SEC Use (	Dnly	
4	Source of	Funds (Se	e Instructions)
	SC		
5	Check if d	isclosure	of legal proceedings is required pursuant to Items 2(d) or 2(e)
6		o or Place	of Organization
	Portugal		
NUMBE	R OF	7	Sole Voting Power
SHARES	5		3,187,500
BENEFICIALLY		8	Shared Voting Power
OWNED	BY		0
EACH		9	Sole Dispositive Power
REPORT			3,187,500
PERSON		10	Shared Dispositive Power
WITH			0
11	Aggregate Amount Beneficially Owned by Each Reporting Person		
	3,187,500		
12			
13 Percent of Class Represented by Amount in Row (11)		presented by Amount in Row (11)	
25.16%			
14 Type of Reporting Person		eporting P	erson
	СО		

## Item 1. Security and Issuer.

This statement on Schedule 13D (the "Schedule 13D") relates to the Class A ordinary shares, \$0.0001 par value per share (the "Class A Ordinary Shares") of Fusion Fuel Green plc, an Irish public limited company (the "Issuer") whose principal executive offices are located at 10 Earlsfort Terrace, Dublin 2, D02 T380, Ireland.

## Item 2. Identity and Background.

The Schedule 13D is being filed by Fusion Welcome, S.A. (the "Reporting Person"). The Reporting Person is a public limited company domiciled in Portugal, *sociedade anónima*. The business address of the Reporting Person is Ex-Siemens Facilities, Rua da Fábrica, S/N, Sabugo 2715-376, Almargem do Bispo, Portugal. There are four shareholders of the Reporting Person, none of which has voting or dispositive control over the securities held thereby. The voting and dispositive decisions regarding the portfolio securities of the Reporting Person require unanimous approval of shareholders of the Reporting Person. The Reporting Person is a Portuguese industrial group whose principal activities have been focused on solar concentrated photovoltaic technology since 2008.

During the last five years, the Reporting Person has not (i) been convicted in any criminal proceeding (excluding traffic violations or similar misdemeanors) or (ii) been a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

## Item 3. Source and Amount of Funds or Other Consideration.

On June 6, 2020, the Issuer entered into a business combination agreement (as amended and restated on August 25, 2020, the "Business Combination Agreement") with HL Acquisitions Corp., a British Virgin Islands business company ("HL"), Fusion Welcome – Fuel, S.A., a public limited company domiciled in Portugal, *sociedade anonima* ("Fusion Fuel"), Fusion Fuel Atlantic Limited, A British Virgin Islands business company and wholly-owned subsidiary of the Issuer ("Merger Sub"), and the shareholders of Fusion Fuel, including the Reporting Person ("Fusion Fuel Shareholders") pursuant to which (i) Merger Sub would merge with and into HL (the "Merger"), with HL being the surviving entity of the Merger and becoming a wholly-owned subsidiary of the Issuer, and (ii) the Issuer would acquire all the issued and outstanding shares of Fusion Fuel (the "Share Exchange," and together with the Merger, the "Transactions").

On December 10, 2020, the parties to the Business Combination Agreement consummated the Transactions in accordance with the Business Combination Agreement.

The Reporting Person held 37,500 ordinary shares of Fusion Fuel immediately prior to the Share Exchange. The Fusion Fuel ordinary shares were converted into an aggregate of 1,593,750 Class B ordinary shares of the Issuer (each Class B ordinary share is convertible at any time and from time to time into one Class A Ordinary Share) and 1,593,750 warrants of the Issuer (each warrant is exercisable for one Class A Ordinary Share at any time and from time to time at an initial exercise price of \$11.50 per share). Pursuant to the Business Combination Agreement, an aggregate of 159,374 Class B ordinary shares of the Issuer held by the Reporting Person are being held in escrow to serve as a source of funds for the indemnification obligations of the Reporting Person.

#### Item 4. Purpose of Transaction.

The Reporting Person acquired the securities described in this Schedule 13D in connection with the closing of the Transactions and intends to review its investment in the Issuer on a continuing basis. Any actions the Reporting Person might undertake may be made at any time and from time to time without prior notice and will be dependent upon the Reporting Person's review of numerous factors, including, but not limited to: an ongoing evaluation of the Issuer's business, financial condition, operations and prospects; price levels of the Issuer's securities; general market, industry and economic conditions; the relative attractiveness of alternative business and investment opportunities; and other future developments.



The Reporting Person may acquire additional securities of the Issuer, and, subject to the transfer restrictions described below in Item 6, retain or sell all or a portion of the securities then held, in the open market or in privately negotiated transactions.

Other than as described above, the Reporting Person does not currently have any plans or proposals that relate to, or would result in, any of the matters listed in Items 4(a)-(j) of Schedule 13D, although, depending on the factors discussed herein, the Reporting Person may change its purpose or formulate different plans or proposals with respect thereto at any time.

## Item 5. Interest in Securities of the Issuer.

(a) The Reporting Person beneficially owns 3,187,500 Class A Ordinary Shares. Such amount includes 1,593,750 Class A Ordinary Shares issuable upon the conversion of Class B Ordinary Shares. Such number of Class A Ordinary Shares represents 25.16% of the class of securities, based on 9,483,356 Class A Ordinary Shares outstanding, as reported in the issuer's 20FR12B filed on December 17, 2020.

(b) The number of shares as to which the person has:

- (i) Sole power to vote or direct the vote: 3,187,500
- (ii) Shared power to vote or direct the vote: 0
- (iii) Sole power to dispose or direct the disposition: 3,187,500
- (iv) Shared power to dispose or direct the disposition: 0

(c) Except as described in Item 4, during the past 60 days the Reporting Person has not effected any transactions in the Class A Ordinary Shares.

(d) None.

(e) Not applicable.

#### Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the Issuer.

#### Amended and Restated Registration Rights Agreement

On December 10, 2020, in connection with the closing of the Transactions, the Issuer entered into an amended and restated registration rights agreement ("Amended and Restated Registration Rights Agreement") providing certain of its shareholders, including the Reporting Person, with certain demand registration rights and piggy-back registration rights with respect to registration statements filed by the Issuer after the closing.

#### Transfer Restrictions

The Business Combination Agreement includes an agreement of the Fusion Fuel Shareholders, including the Reporting Person, pursuant to which they agreed, among other things not to transfer the Class B Ordinary Shares received by them in connection with the Share Exchange, except to certain permitted transferees, for a period ending on the one-year anniversary of the Closing Date, or earlier if, subsequent to the Closing Date, the Issuer consummates a liquidation, merger, stock exchange or other similar transaction which results in all of the Issuer's shareholders having the right to exchange their ordinary shares for cash, securities or other property.

#### Indemnification Escrow Agreement

The Business Combination Agreement provides for mutual indemnification by HL and the Fusion Fuel Shareholders for breaches of their respective representations, warranties, and covenants. Claims for indemnification may be asserted once damages exceed a €750,000 threshold and will be reimbursable to the full extent of the damages in excess of such threshold. Claims for indemnification must be brought before the tenth business day after the Issuer files its annual report for the fiscal year ending December 31, 2021. To provide a source of funds for the Fusion Fuel Shareholders' indemnification of HL, on December 10, 2020, the Reporting Person entered into an indemnification escrow agreement ("Indemnification Escrow Agreement") with the Issuer, HL, Fusion Fuel, Continental Stock Transfer & Trust Company, as escrow agent, and the other parties thereto, pursuant to which an aggregate of 159,374 of the Class B Ordinary Shares issued to the Reporting Person in the Transactions was placed into escrow.

The foregoing descriptions of the Amended and Restated Registration Rights Agreement, the Business Combination Agreement, and the Indemnification Escrow Agreement do not purport to be complete and are qualified in their entirety by reference to the full text of such agreements, each of which is attached as an exhibit to the Schedule 13D and is incorporated herein by reference.

Except as set forth herein, the Reporting Person does not have any contracts, arrangements, understandings or relationships (legal or otherwise) with any person with respect to any securities of the Issuer, including but not limited to any contracts, arrangements, understandings or relationships concerning the transfer or voting of such securities, finder's fees, joint ventures, loan or option arrangements, puts or calls, guarantees of profits, division of profits or losses, or the giving or withholding of proxies.

## Item 7. Materials to be Filed as Exhibits

Exhibit Number	Description
1	Amended and Restated Business Combination Agreement between HL Acquisitions Corp., Fusion Fuel Green plc, Fusion Welcome – Fuel, S.A., Fusion Fuel Atlantic Limited, and the former shareholders of Fusion Welcome – Fuel, S.A., dated August 25, 2020.
2	
2	<u>Amended and Restated Registration Rights Agreement between HL Acquisitions Corp., Fusion Fuel Green plc, certain former shareholders</u> of HL Acquisitions Corp., EarlyBirdCapital, Inc., and certain former shareholders of Fusion Welcome – Fuel, S.A., dated December 10, 2020.
3	Indemnification Escrow Agreement between Fusion Fuel Green plc, Fusion Welcome – Fuel, S.A., Fusion Welcome, S.A., HL Acquisitions Corp., Jeffrey Schwarz, and Continental Stock Transfer & Trust Company, dated December 10, 2020.
	5

## SIGNATURE

After reasonable inquiry and to the best of my knowledge and belief, I certify that the information set forth in this statement is true, complete and correct.

Date: December 22, 2020

## FUSION WELCOME, S.A.

By: /s/ João Wahnon

Name: João Wahnon Title: Director

## AMENDED AND RESTATED

## **BUSINESS COMBINATION AGREEMENT**

BY AND AMONG

HL ACQUISITIONS CORP.,

FUSION WELCOME – FUEL, S.A.,

FUSION FUEL GREEN LIMITED,

FUSION FUEL ATLANTIC LIMITED, and

THE FUSION FUEL SHAREHOLDERS

DATED AS OF AUGUST 25, 2020

## TABLE OF CONTENTS

		Page
Article I THE MERGER		
1.1	The Merger	2
1.2	Governing Documents	2
1.3	Effect on Securities	2
1.4	Tax Consequences	4
1.5	Taking of Necessary Action; Further Action	4
1.6	Sponsor Agreement	4
Article II TH	E SHARE EXCHANGE	4
2.1	The Share Exchange	4
2.2	Tax Consequences	6
2.3	Taking of Necessary Action; Further Action	6
2.4	Contingent Consideration	6
2.5	Indemnification Escrow	9
2.6	Closing	9
2.7	HL Representative	9
2.8	Company Shareholder Representative	9
A set of a TTE DT		10
Article III KI	PRESENTATIONS AND WARRANTIES REGARDING THE COMPANY	10
3.1	Organization and Qualification	10
3.2	Subsidiaries	10
3.3	Capitalization	11
3.4	<u>Authority Relative to this Agreement</u>	11
3.5	No Conflict; Required Filings and Consents	12
3.6	Compliance	13
3.7	Permits	13
3.8	Financial Matters	14
3.9	Absence of Certain Developments	15
3.10	Condition and Sufficiency of Assets	15
3.11	Litigation	15
3.12	Employment Matters	15
3.13	Restrictions on Business Activities	16
3.14	Title to Property	16
3.15	Taxes	17
3.16	Environmental Matters	18
3.17	Brokers	18
3.18	Intellectual Property	18
3.19	Agreements, Contracts and Commitments	19
3.20	Insurance	21
-		

i

3.21	Interested Party Transactions	21
3.22	Proxy Statement/Prospectus	21
3.23	Certain Business Practices	21
3.24	No Additional Representations and Warranties; No Reliance	22
3.25	Survival of Representations and Warranties	22
Article IV RE	PRESENTATIONS AND WARRANTIES OF THE COMPANY SHAREHOLDERS	23
4.1	Organization, Standing, and Corporate Power	23
4.2	Ownership of Company Ordinary Shares and Company Class A Shares	24
4.3	Investment Intent	24
4.4	Litigation	24
4.5	Certain Business Practices	24
4.6	No Additional Representations and Warranties; No Reliance	25
4.7	Survival of Representations and Warranties	25
Article V REI	PRESENTATIONS AND WARRANTIES OF HL	25
5.1	Organization and Qualification	25
5.2	Subsidiaries	26
5.3	<u>Capitalization</u>	26
5.4	Authority Relative to this Agreement	27
5.5	No Conflict; Required Filings and Consents	28
5.6	Compliance	28
5.7	HL SEC Reports and Financial Statements	29
5.8	No Undisclosed Liabilities	30
5.9	Absence of Certain Changes or Events	31
5.10	Litigation	31
5.11	Employee Benefit Plans	31
5.12	Labor Matters	31
5.13	Business Activities	31
5.14	<u>Title to Property</u>	31
5.15	Intellectual Property	32
5.16	Taxes	32
5.17	Environmental Matters	33
5.18	Brokers	33
5.19	Agreements, Contracts and Commitments	33
5.20	Insurance	34
5.21	Interested Party Transactions	34
5.22	<u>HL Listing</u>	34
5.23	Trust Fund	34
5.24	Proxy Statement/Prospectus	35
5.25	No Additional Representations and Warranties; No Reliance	35
5.26	Survival of Representations and Warranties	35

# ii

Artı	cle VI RE	PRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	35
	(1		25
	6.1 6.2	Organization and Qualification	35
	6.2	Subsidiaries Conitalization	36 36
	6.3 6.4	Capitalization Authority Relative to this Agreement	30
	6.5	<u>No Conflict; Required Filings and Consents</u>	37
			38
	6.6 6.7	Compliance	38
		Litigation Business Activities	38
	6.8	Business Activities	38 39
	6.9 6.10	Interested Party Transactions	39
			39
	6.11	Board Approval	
	6.12	Proxy Statement/Prospectus	39 39
	6.13	No Additional Representations and Warranties	
	6.14	Survival of Representations and Warranties	40
A			40
Arti	cie vii C	ONDUCT PRIOR TO CLOSING	40
	7 1	Contractor Designation III. do Company Designation IM and C. I.	40
	7.1	Conduct of Business by HL, the Company, Parent, and Merger Sub	40
	7.2	Confidentiality; Access to Information	43
	7.3	Exclusivity	45
	7.4	Certain Financial Information	45
	7.5	Company Audited Financial Statements	46
	7.6	Commercially Reasonable Efforts	46
A	1 1 1 1 1 1		47
Arti	cle VIII A	ADDITIONAL AGREEMENTS	47
	0.1	Erone E.A. On exist Martine	47
	8.1	Form F-4; Special Meeting	47
	8.2	Conversion of Parent to Public Limited Company; Directors and Officers of Parent After Transactions	48
	8.3	Intentionally Omitted	48
	8.4	Public Announcements	48
	8.5	Required Information	49
	8.6	No Securities Transactions	50
	8.7	No Claim Against Trust Fund	50
	8.8	Disclosure of Certain Matters	51
	8.9	Securities Listing	51
	8.10	Charter Protections; Directors' and Officers' Liability Insurance	51
	8.11	Insider Loans	53
	8.12	HL Borrowings	53
	8.13	Trust Fund Disbursement	53
	8.14	Intended Tax Treatment	54
	8.15	Incentive Equity Plan	54
	8.16	Employment Agreements	54
	8.17	Third Party Consents	54

iii

8.18	Amendment to the Stock Escrow Agreement	55
8.19	Amendment to the Warrant Agreement	55
8.20	Transfer Restrictions	55
8.21	Sublease Agreement.	55
8.22	Registration Rights Agreement	56
8.23	Shareholders Agreement	56
8.24	Reservation of Shares	56
8.25	Extension Proxy Statement	56
Article IX CC	NDITIONS TO THE TRANSACTION	57
9.1	Conditions to Obligations of Each Party to Effect the Transactions	57
9.2	Additional Conditions to Obligations of the Company, Company Shareholders, Parent and Merger Sub	59
9.3	Additional Conditions to the Obligations of HL	60
Article X TEI	RMINATION	61
10.1	Termination	61
10.2	Notice of Termination; Effect of Termination	63
10.3	Fees and Expenses	63
Article XI Inc	lemnification	63
11.1	Indemnification	63
11.2	Claims	65
11.3	Insurance Effect	65
11.4	Limitations on Indemnification	66
11.5	Adjustments to Consideration	66
11.6	Payments; Application of Escrow Fund	66
11.7	Exclusive Remedy	67
Article XII G	ENERAL PROVISIONS	67
12.1	Notices	67
12.2	Interpretation	68
12.3	Counterparts; Electronic Delivery	68
12.4	Entire Agreement; Third Party Beneficiaries	68
12.5	<u>Severability</u>	69
12.6	Other Remedies; Specific Performance	69
12.7	Governing Law	69
12.8	Consent to Jurisdiction; WAIVER OF TRIAL BY JURY	69
12.9	Rules of Construction	70
12.10	Assignment	70
12.11	Amendment	70
12.12	Extension; Waiver	70
12.13	Currency	70
12.14	<u>Schedules</u>	70

Exhibits

Exhibit A - Certain Definitions

iv

## Index to Defined Terms

This index is included	for convenience only an	d does not constitute a	part of the agreement

ApprovalsSection 3.1(a)AssetsSection 3.10Business DayExhibit ABVI Companies ActRecital ACertificationsSection 8.1(a)(iii)Chater Amendments ProposalsSection 8.1(a)(iii)Chater Amendments ProposalsSection 1.12ClaimSection 1.12ClosingSection 1.12Closing ConsiderationExhibit AClosing ConsiderationSection 2.6Closing Form 20-FSection 2.6Closing Form 20-FSection 3.8(a)(i)Company PanelSection 3.8(a)(i)Company Annual Financial StatementsSection 3.8(a)(i)Company Annual Financial StatementsSection 3.8(a)(i)Company ConsiderationExhibit ACompany ConsiderationSection 3.1(a)Company ConsiderationSection 3.8(a)(i)Company ToratesSection 3.8(a)(i)Company ToratesSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company Intellectual PropertyExhibit ACompany Intellectual PropertyExhibit ACompany Prinancial StatementsSection 3.8(a)(i)Company Stateholder RepresentativeSection 3.8(a)(i)Company Prinacial StatementsSection 3.8(a)(i)Company Prinacial Sta	Term	Section
AffiliateExhibit AAgreementPeambleAmended and Restated Articles of AssociationRecital EAmended Warrant AgreementSection 3.19ApprovalsSection 3.10Business DayExhibit ABusiness DayExhibit ABVI Companies ActSection 5.7(a)Charter Amendments ProposalsSection 5.7(a)Charter Amendments ProposalsSection 3.1(a)Charter DocumentsSection 1.12ClosingSection 2.6Closing ConsiderationExhibit AClosing ConsiderationSection 1.2Closing ParteSection 2.6Closing ParteSection 2.6Closing ParteSection 3.4(a)Closing ParteSection 3.4(a)Closing ParteSection 3.4(a)Closing ParteSection 3.4(a)Closing ParteSection 3.4(a)CodeRecital DCompanyPreambleCompany Annual Financial StatementsSection 3.4(a)Company Closing CertificateSection 3.4(a)Company Closing CertificateSection 3.4(a)Company Closing CertificateSection 3.4(a)Company NonsiderationExhibit ACompany Indernical StatementsSection 3.4(a)Company NonsiderationSection 3.4(a)Company NonsiderationSection 3.4(a)Company Indernical StatementsSection 3.4(a)Company NonsiderationSection 3.4(a)Company Indernical StatementsSection 3.4(a)Company Indernical StatementsSection 3.4(a) </td <td>Additional HL SEC Reports</td> <td>Section 7.1</td>	Additional HL SEC Reports	Section 7.1
Amended and Restated Articles of AssociationRecital EAmended Warrant AgreementSection 3.1 (a)ApprovalsSection 3.1 (a)AssetsSection 3.1 (a)Business DayExhibit ABVI Companies ActRecital ACrtificationsSection 7.7 (a)Charter Amendments ProposalsSection 7.7 (a)Charter DocumentsSection 1.1 (a)ClosingSection 1.1 (a)ClosingSection 1.1 (a)Closing ConsiderationSection 2.6Closing ConsiderationSection 2.6Closing Form 20-FSection 2.6Closing Form 20-FSection 3.8 (a)(a)Closing Form 20-FSection 8.8 (c)CodeRecital PCompany Annual Financial StatementsSection 7.5Company Annual Financial StatementsSection 7.5Company Closing CertificateSection 3.8 (a)(i)Company Closing CertificateSection 3.8 (a)(i)Company Closing CertificateSection 3.1 (a)Company Closing CertificateSection 7.5Company Financial StatementsSection 7.5 (a)Company Closing CertificateSection 3.1 (a)Company Closing CertificateSection 3.1 (a)Company Tinancial StatementsSection 3.1 (a)Company Tinancial StatementsSection 3.1 (a)Company Multer Financial StatementsSection 3.1 (a)Company Tinancial StatementsSection 3.1 (a)Company Tinancial StatementsSection 3.1 (a)Company Tinancial StatementsSection 3.1 (a)Company	*	Exhibit A
Amended Warrant AgreementSection 3.19ApprovalsSection 3.1(a)ApprovalsSection 3.1(a)Business DayExhibit ABVI Companies ActRevital ACertificationsSection 3.1(a)Charter Amendments ProposalsSection 3.1(a)Charter DocumentsSection 3.1(a)ClaimSection 3.1(a)ClosingSection 1.2Closing ConsiderationExhibit AClosing Press ReleaseSection 3.4(c)Closing Press ReleaseSection 3.4(c)Company Annual Financial StatementsSection 3.4(a)Company Annual Financial StatementsSection 3.4(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.4(a)CodeRecitial DCompany Annual Financial StatementsSection 3.4(a)Company Annual Financial StatementsSection 3.4(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.4(a)Company Financial StatementsSection 3.4(a)Company Financial StatementsSection 3.4(a)Company Financial StatementsSection 3.4(a)Company Financial StatementsSection 3.4(a)Company Financial	Agreement	Preamble
ApprovalsSection 3.1(a)AssetsSection 3.10Business DayExhibit ABVI Companies ActRecital ACertificationsSection 8.1(a)(iii)Chater Amendments ProposalsSection 8.1(a)(iii)Chater Amendments ProposalsSection 1.12ClaimSection 1.12ClosingSection 1.12Closing ConsiderationExhibit AClosing ConsiderationSection 2.6Closing Form 20-FSection 2.6Closing Form 20-FSection 3.8(a)(i)Company PanelSection 3.8(a)(i)Company Annual Financial StatementsSection 3.8(a)(i)Company Annual Financial StatementsSection 3.8(a)(i)Company ConsiderationExhibit ACompany ConsiderationSection 3.1(a)Company ConsiderationSection 3.8(a)(i)Company ToratesSection 3.8(a)(i)Company ToratesSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company Prinancial StatementsSection 3.8(a)(i)Company Intellectual PropertyExhibit ACompany Intellectual PropertyExhibit ACompany Prinancial StatementsSection 3.8(a)(i)Company Stateholder RepresentativeSection 3.8(a)(i)Company Prinacial StatementsSection 3.8(a)(i)Company Prinacial Sta	Amended and Restated Articles of Association	Recital E
AssetsSection 3.10Business DayExhibit ABVI Companies ActRecital ACertificationsSection 5.7(a)Charter Amements ProposalsSection 3.1(a)Charter Amements ProposalsSection 3.1(a)ClairSection 3.1(a)ClairSection 3.1(a)ClosingSection 2.6Closing ConsiderationExhibit AClosing PartsSection 2.6Closing Press ReleaseSection 2.6Closing Press ReleaseSection 3.4(c)CodeRecital DCompany Annual Financial StatementsSection 3.8(a)(ii)Company Closing CartificateSection 3.8(a)(ii)Company Closing CartificateSection 1.2Company Closing CartificateSection 3.8(a)(ii)Company Financial StatementsSection 3.8(a)(ii)Company Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesSection 1.1(b)Company ProductExhibit ACompany Ordinary SharesSection 3.8(a)(ii)Company ProductSchibit ACompany ProductSchibit ACompany ProductSchibit ACompany ProductSchibit ACompany ShareholdersSectio	Amended Warrant Agreement	Section 8.19
AssetsSection 3.10Business DayExhibit ABVI Companies ActRecital ACertificationsSection 5.7(a)Charter Amements ProposalsSection 3.1(a)Charter Amements ProposalsSection 3.1(a)ClairSection 3.1(a)ClairSection 3.1(a)ClosingSection 2.6Closing ConsiderationExhibit AClosing PartsSection 2.6Closing Press ReleaseSection 2.6Closing Press ReleaseSection 3.4(c)CodeRecital DCompany Annual Financial StatementsSection 3.8(a)(ii)Company Closing CartificateSection 3.8(a)(ii)Company Closing CartificateSection 1.2Company Closing CartificateSection 3.8(a)(ii)Company Financial StatementsSection 3.8(a)(ii)Company Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesSection 1.1(b)Company ProductExhibit ACompany Ordinary SharesSection 3.8(a)(ii)Company ProductSchibit ACompany ProductSchibit ACompany ProductSchibit ACompany ProductSchibit ACompany ShareholdersSectio	Approvals	Section 3.1(a)
BVI Companies Act Recital A   Certifications Section 5.7(a)   Chatter Amendments Proposals Section 8.1(a)(ii)   Chatter Documents Section 3.1(a)   Claim Section 2.6   Closing Consideration   Closing Consideration Exhibit A   Closing Form 20-F Section 2.6   Closing Press Release Section 2.6   Company Section 2.6 Section 3.4(c)   Code Recital D   Company Annual Financial Statements Section 3.4(c)   Company Udited Financial Statements Section 3.8(a)(i)   Company Consideration Exhibit A   Company Consideration Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Shares Exhibit A   Company Shares Section 3.8(a)(i)   Company Shedule Article IV Preamble <td></td> <td>Section 3.10</td>		Section 3.10
BVI Companies Act Recital A   Certifications Section 5.7(a)   Chatter Amendments Proposals Section 8.1(a)(ii)   Chatter Documents Section 3.1(a)   Claim Section 2.6   Closing Consideration   Closing Consideration Exhibit A   Closing Form 20-F Section 2.6   Closing Press Release Section 2.6   Company Section 2.6 Section 3.4(c)   Code Recital D   Company Annual Financial Statements Section 3.4(c)   Company Udited Financial Statements Section 3.8(a)(i)   Company Consideration Exhibit A   Company Consideration Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Indemnitees Section 3.8(a)(i)   Company Shares Exhibit A   Company Shares Section 3.8(a)(i)   Company Shedule Article IV Preamble <td>Business Day</td> <td>Exhibit A</td>	Business Day	Exhibit A
Charter Amendments ProposalsSection 8.1(a)(iii)Charter DocumentsSection 3.1(a)ClaimSection 11.2ClosingSection 11.2Closing ConsiderationExhibit AClosing DateSection 2.6Closing Form 20-FSection 8.4(c)CodeRecital DCompany Annual Financial StatementsSection 3.8(a)(i)Company Annual Financial StatementsSection 7.5Company Annual Financial StatementsSection 3.8(a)(i)Company Class A SharesExhibit ACompany ConsiderationExhibit ACompany ConsiderationSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company ConsiderationSection 3.8(a)(i)Company ConsiderationSection 9.3(a)Company ConsiderationSection 3.8(a)(i)Company IndemnicesSection 3.8(a)(i)Company IndemnicesSection 3.8(a)(i)Company Interim Financial StatementsSection 3.8(a)(i)Company Interim Financial StatementsSection 3.8(a)(ii)Company ProductExhibit ACompany Interim Financial StatementsSection 3.8(a)(ii)Company Interim Financial StatementsSection 3.8(a)(ii)Company ProductExhibit ACompany ProductExhibit ACompany ScheduleArticle IV PreambleCompany ScheduleSection 2.8Company SharesSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder Representa	BVI Companies Act	Recital A
Charter DocumentsSection 3.1(a)ClaimSection 1.2ClosingSection 2.6Closing ConsiderationExhibit AClosing DateSection 2.6Closing Press ReleaseSection 8.4(c)Company PreambleRecital DCompany Annual Financial StatementsSection 3.8(a)(i)Company Anulited Financial StatementsSection 7.5Company Closing CertificateSection 3.1(a)Company ConsiderationExhibit ACompany ConsiderationExhibit ACompany ConsiderationSection 3.1(a)Company Prinancial StatementsSection 3.1(a)Company ConsiderationExhibit ACompany ConsiderationExhibit ACompany ConsiderationSection 3.1(a)Company Financial StatementsSection 3.1(a)Company Prinancial StatementsSection 3.1(a)Company Prinancial StatementsSection 3.1(a)Company Intellectual PropertyExhibit ACompany Intellectual PropertyExhibit ACompany SharesExhibit ACompany SharesExhibit ACompany SharesExhibit ACompany ShareholderSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.6Contingent Consideration ShareholderSection 2.4(a)	Certifications	Section 5.7(a)
Charter DocumentsSection 3.1(a)ClaimSection 1.2ClosingSection 2.6Closing ConsiderationExhibit AClosing DateSection 2.6Closing Press ReleaseSection 8.4(c)Company PreambleRecital DCompany Annual Financial StatementsSection 3.8(a)(i)Company Anulited Financial StatementsSection 7.5Company Closing CertificateSection 3.1(a)Company ConsiderationExhibit ACompany ConsiderationExhibit ACompany ConsiderationSection 3.1(a)Company Prinancial StatementsSection 3.1(a)Company ConsiderationExhibit ACompany ConsiderationExhibit ACompany ConsiderationSection 3.1(a)Company Financial StatementsSection 3.1(a)Company Prinancial StatementsSection 3.1(a)Company Prinancial StatementsSection 3.1(a)Company Intellectual PropertyExhibit ACompany Intellectual PropertyExhibit ACompany SharesExhibit ACompany SharesExhibit ACompany SharesExhibit ACompany ShareholderSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.6Contingent Consideration ShareholderSection 2.4(a)	Charter Amendments Proposals	Section 8.1(a)(iii)
ClaimSection 11.2ClosingSection 2.6Closing ConsiderationExhibit AClosing DateSection 2.6Closing Form 20-FSection 8.4(c)CodeRecital DCompany StaresSection 8.4(c)Company Annual Financial StatementsSection 3.8(a)(i)Company ConsiderationExhibit ACompany ConsiderationSection 3.8(a)(i)Company ConsiderationExhibit ACompany ConsiderationExhibit ACompany IndemniteesSection 3.19(a)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Intellectual PropertyExhibit ACompany ProductExhibit ACompany Resistered Intellectual PropertyExhibit ACompany ShareholderSection 2.8Company Shareholder RepresentativeSection 7.2(a)Company Shareholder RepresentativeSection 7.2(a)Continent ConsiderationExhibit ACompany ShareholderSection 7.2(a)Continent ConsiderationSchibit ACompany ShareholderSection 7.2(a)Continent ConsiderationSchibit ACompany ShareholderSection 7.2(a)Continent Consideration<	*	
Closing ConsiderationExhibit AClosing DateSection 2.6Closing DateSection 2.6Closing Prom 20-FSection 8.4(c)CodeRecital DCompany PreambleCompanyCompany Annual Financial StatementsSection 3.8(a)(i)Company Anudie Financial StatementsSection 7.5Company Closing CertificateSection 3.8(a)(i)Company Closing CertificateSection 3.8(a)(i)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company IndemniteesSection 3.19(a)Company Intellectual PropertySection 11.1(b)Company Intellectual PropertyExhibit ACompany Ordinary StaresSection 3.8(a)(ii)Company ProductExhibit ACompany ProductExhibit ACompany SecturitesSection 2.8(a)(iii)Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.5Contingent Consideration ShareholderSection 2.4(a)	Claim	
Closing ConsiderationExhibit AClosing DateSection 2.6Closing DateSection 2.6Closing Prom 20-FSection 8.4(c)CodeRecital DCompany PreambleCompanyCompany Annual Financial StatementsSection 3.8(a)(i)Company Anudie Financial StatementsSection 7.5Company Closing CertificateSection 3.8(a)(i)Company Closing CertificateSection 3.8(a)(i)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company IndemniteesSection 3.19(a)Company Intellectual PropertySection 11.1(b)Company Intellectual PropertyExhibit ACompany Ordinary StaresSection 3.8(a)(ii)Company ProductExhibit ACompany ProductExhibit ACompany SecturitesSection 2.8(a)(iii)Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 2.5Contingent Consideration ShareholderSection 2.4(a)	Closing	Section 2.6
Closing DateSection 2.6Closing Form 20-FSection 8.4(c)Colsing Press ReleaseSection 8.4(c)CodeRecital DCompanyPreambleCompany Annual Financial StatementsSection 7.5Company Class A SharesExhibit ACompany ConsiderationSection 3.3(a)(i)Company ConsiderationSection 3.3(a)Company ContractsSection 3.19(a)Company IndemnitesSection 3.19(a)Company IndemnitesSection 3.19(a)Company IndemnitesSection 3.8(a)(ii)Company IndemnitesSection 3.8(a)(iii)Company IndemnitesSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Registered Intellectual PropertyExhibit ACompany Registered Intellectual PropertyExhibit ACompany Shareholder RepresentativeSection 2.8Company Shareholder RepresentativeSection 7.2(a)Confidentiality AgreementSection 2.4(a)Contingent Consideration ShareholderSection 2.4(a)		Exhibit A
Closing Form 20-FSection 8.4(c)Closing Press ReleaseSection 8.4(c)CodeRecital DCompanyPreambleCompany Annual Financial StatementsSection 3.8(a)(i)Company Andited Financial StatementsSection 7.5Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company IndemnitesSection 3.8(a)(ii)Company IndemnitesSection 3.8(a)(ii)Company Interim Financial StatementsSection 3.8(a)(ii)Company ScheduleExhibit ACompany ScheduleArticle IV PreambleCompany ScheduleSection 2.8Company SchedulersPreambleCompany Shareholder RepresentativeSection 2.5Contingent ConsiderationExhibit AContingent ConsiderationSection 2.4(a)		Section 2.6
Closing Press ReleaseSection 8.4(c)CodeRecital DCompanyPreambleCompany Annual Financial StatementsSection 3.8(a)(i)Company Audited Financial StatementsSection 7.5Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany InterimentsSection 3.19(a)Company InternetsSection 3.8(a)(ii)Company InternetsSection 3.8(a)(ii)Company InternetsSection 3.8(a)(iii)Company InternetsSection 3.8(a)(iii)Company InternetsSection 3.8(a)(iii)Company InternetsSection 3.8(a)(iii)Company Internet Financial StatementsSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Ordinary SharesExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 2.2(a)Confidentiality AgreementSection 2.5Contingent ConsiderationExhibit AConsiderationExhibit AConsiderationSection 2.4(a)		Section 8.4(c)
CodeRecital DCompanyPreambleCompany Anual Financial StatementsSection 3.8(a)(i)Company Audited Financial StatementsSection 7.5Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 1.1(b)Company Intellectual PropertyExhibit ACompany Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany Shareholder RepresentativeSection 2.8(a)(iii)Company Shareholder RepresentativeSection 2.8Company Shareholder SourcePreambleContingent ConsiderationSection 7.2(a)Contingent ConsiderationSection 7.2(a)Contingent Consideration ShareholderSection 2.4(a)		Section 8.4(c)
CompanyPreambleCompany Annual Financial StatementsSection 3.8(a)(i)Company Audited Financial StatementsSection 7.5Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company OntractsSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Interim Financial StatementsSection 3.8(a)(iii)Company Ordinary SharesExhibit ACompany ProductExhibit ACompany ScheduleArticle IV PreambleCompany ScheduleSection 2.8Company Shareholder RepresentativeSection 7.2(a)Confidentiality AgreementSection 7.2(a)Contingent Consideration ShareholderSection 2.4(a)	Code	Recital D
Company Annual Financial StatementsSection 3.8(a)(i)Company Audited Financial StatementsSection 7.5Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 1.1(b)Company Intellectual PropertyExhibit ACompany Ordinary SharesSection 3.8(a)(ii)Company ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany SecuritiesSection 2.8(a)(a)Company Shareholder RepresentativeSection 2.8Company ShareholdersPreambleContingent ConsiderationSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Preamble
Company Audited Financial StatementsSection 7.5Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ConstructsSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Ordinary SharesSection 3.8(a)(ii)Company Registered Intellectual PropertyExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleContingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Section 3.8(a)(i)
Company Class A SharesExhibit ACompany Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ConsiderationSection 3.19(a)Company ContractsSection 3.19(a)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Ordinary SharesSection 3.8(a)(ii)Company ProductExhibit ACompany ScheduleArticle IV PreambleCompany ScheduleArticle IV PreambleCompany Shareholder RepresentativeSection 2.8Company ShareholdersSection 7.2(a)Contingent Consideration ShareholderExhibit AContingent Consideration ShareholderSection 7.2(a)		
Company Closing CertificateSection 9.3(a)Company ConsiderationExhibit ACompany ContractsSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Ordinary SharesSection 3.8(a)(ii)Company Registered Intellectual PropertyExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)Contingent Consideration ShareholderExhibit AContingent Consideration ShareholderSection 2.4(a)		Exhibit A
Company ConsiderationExhibit ACompany ContractsSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Interim Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesSection 3.8(a)(ii)Company ProductExhibit ACompany ScheduleCompany ScheduleCompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleCompany ShareholdersSection 7.2(a)Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Section 9.3(a)
Company ContractsSection 3.19(a)Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Interim Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany ScheduleSection 2.8Company Shareholder RepresentativeSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Exhibit A
Company Financial StatementsSection 3.8(a)(ii)Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Interim Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleContingent ConsiderationSection 2.5Contingent Consideration ShareholderSection 2.4(a)		Section 3.19(a)
Company IndemniteesSection 11.1(b)Company Intellectual PropertyExhibit ACompany Interim Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany ScheduleSection 2.8Company Shareholder RepresentativeSection 7.2(a)Confidentiality AgreementSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Section 3.8(a)(ii)
Company Intellectual PropertyExhibit ACompany Interim Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		
Company Interim Financial StatementsSection 3.8(a)(ii)Company Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Exhibit A
Company Ordinary SharesExhibit ACompany ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Section 3.8(a)(ii)
Company ProductExhibit ACompany Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		
Company Registered Intellectual PropertyExhibit ACompany ScheduleArticle IV PreambleCompany SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Exhibit A
Company ScheduleArticle IV PreambleCompany SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Exhibit A
Company SecuritiesExhibit ACompany Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Article IV Preamble
Company Shareholder RepresentativeSection 2.8Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		
Company ShareholdersPreambleConfidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		Section 2.8
Confidentiality AgreementSection 7.2(a)ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		
ContinentalSection 2.5Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		
Contingent ConsiderationExhibit AContingent Consideration ShareholderSection 2.4(a)		
Contingent Consideration Shareholder Section 2.4(a)		
	Converting Shareholder	Section 1.3(a)

V

Term	Section
Corporate Law	Recital A
Copyrights	Exhibit A
Director Proposal	Section 8.1(a)(ii)
D&O Indemnified Person	Section 8.10(b)
Earnout Condition	Section 2.4(a)
EBC	Section 1.3(d)
Employment Agreements	Section 8.16
Environmental Laws	Exhibit A
Escrow Fund	Section 2.5
Escrow Termination Date	Section 2.5
Exchange Act	Section 3.8(b)
Exercise Notice	Section 2.4(f)
Export Control Laws	Exhibit A
Extension Proposal	Section 8.25(a)
Extension Proxy Statement	Section 8.25(a)
Extension Shareholders' Meeting	Section 8.25(c)
Final Prospectus	Section 8.7
Effective Time	Section 1.1
Form F-4	Exhibit A
Governmental Entity	Exhibit A
Hazardous Substances	Exhibit A
HL	Preamble
HL Audited Financial Statements	Section 5.7(b)
HL Borrowings	Section 8.12
HL Closing Certificate	Section 9.2(a)
HL Contracts	Section 5.19(a)
HL Convertible Securities	Section 5.3(b)
HL Extension Approval	Section 8.25(c)
HL Financial Statements	Section 5.7(b)
HL Indemnitees	Section 11.1(a)
HL Merger Consideration	Exhibit A
HL Merger Shares	Section 1.3(a)
HL Ordinary Shares	Exhibit A
HL Parent Warrant	Section 1.3(b)
HL Preferred Shares	Exhibit A
HL Representative	Section 2.7
HL Rights	Section 1.3(c)
HL Schedule	Article V Preamble
HL SEC Reports	Section 5.7(a)
HL SEC Reports HL Securities	Exhibit A
HL Special Meeting	Section 8.1(a)
HL Special Meeting HL Shareholder Matters	Section 8.1(a)
HL Shareholders	Exhibit A
HL Stock Options	Section 5.3(b)
HL Unaudited Financial Statements	
	Section 5.7(b)

vi

II. UPOs Exhibit A   III. Warrants Exhibit A   IRRS Exhibit A   Indemnification Pool Exhibit A   Indemnification Security Laws Section 1.1 (b)   Indemnification Escow Agreement Section 2.4 (c)   Indiadmification Escow Agreement Exhibit A   Insulance Policies Exhibit A   Insulance Policies Exhibit A   Interim Period Exhibit A   Interim Period Section 7.1   Interim Period Section 7.1   Lien Exhibit A   Material Company Contracts Section 3.19(a)   Me	Term	Section
IFRSExhibit Aimmediate familyExhibit Aindemnification PoolExhibit AIndemnitesSection 11.1(b)IndemnitesSection 12IndemnitesSection 2.4(e)IndemnitesSection 2.5IndemnitesExhibit AIndemnitesExhibit AIndemnitesExhibit AInstarterExhibit AInstarterExhibit AInstarterSection 2.5Information Privacy and Security LawsExhibit AInstarterExhibit AInstarterSection 7.1Intellectual PropertyExhibit AIntellectual PropertySection 7.1Irish Companies ActRecital ALegal RequirementsExhibit ALegal RequirementsSection 7.1LesseSection 7.1Material Company ContractsSection 7.1Material Adverse EffectSection 7.14(a)Material Adverse EffectSection 7.14(a)Merger ProposalSection 7.14(a)Merger ProposalSection 7.14(a)Most Recent Balance SheetSection 7.14(a)Most Recent Balance Sheet DateSection 7.22(a)Most Recent Balance Sheet DateSe	HL UPOs	Exhibit A
Immediate familyExhibit AIndemnification PoolExhibit AIndemnification PoolSection 11.1(b)IndemnitorsSection 11.2Independent ExpertSection 2.4(c)Indemnification Escrow AgreementExhibit AIndemnification Escrow AgreementExhibit AInsurance PoliciesExhibit AInsurance PoliciesExhibit AInterim PeriodSection 7.1Interim PeriodSection 7.1Interim PeriodExhibit AInterim PeriodExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALossesSection 11.1(c)Material Company ContractsSection 3.19(a)Merger ProposalSection 3.19(a)Merger SubSection 3.10(a)Merger SubSection 3.8(a)(i)Most Recent Balance Sheet DateSection 3.8(a)(i)Most Recent Balance Sheet DateSection 3.8(a)(i)Naterial Company SontractsSection 1.12Outside DateSection 1.2Most Recent Balance Sheet DateSection 3.8(a)(i)Nost Recent Balance Sheet DateSection 3.8(a)(i)Nost Recent Balance Sheet DateSection 3.8(a)(i)Naterial Company SontractsSection 1.12Outside DateSection 3.8(a)(i)Naterial Company SharesExhibit AParent Class A Ordinary Shares	HL Warrants	Section 1.3(b)
Indemnification PoolExhibit AIndemnification PoolSection 11.1(b)Indemnification ExportSection 11.2Independent ExportSection 2.4(c)Indemnification Escow AgreementSection 2.4Indemnification Escow AgreementExhibit AInsiderExhibit AInsiderExhibit AInstance PoliciesExhibit AIntellectual PropertyExhibit AInterim PeriodExhibit AInterim PeriodExhibit AIde RequirementsExhibit ALegal RequirementsSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Meterger ProposalSection 3.19(a)Merger ProposalSection 3.4(a)(i)Merger ProposalSection 3.4(a)(ii)Most Recent Balance SheetSection 3.4(a)(ii)Most Recent Balance Sheet DateSection 3.2(a)(ii)Nost Recent Balance Sheet DateSection 3.4(a)(ii)Nost Recent Balance Sheet DateSection 3.4(a)(ii)Nost Recent Balance Sheet DateSection 3.4(a)(ii)Nost Recent Balance Sheet DateSection 1.2Nost Recent Balance Sheet DateSection 1.2Nost Recent Balance Sheet Date <td< td=""><td>IFRS</td><td>Exhibit A</td></td<>	IFRS	Exhibit A
IndemnitesSection 11 (b)IndemnitorsSection 11 2IndemnitorsSection 2.4(e)Indemnification Escrow AgreementSection 2.5Indimition Privacy and Security LawsExhibit AInsurance PoliciesExhibit AInsurance PoliciesExhibit AInterim PeriodExhibit AInterim PeriodSection 7.1Interim PeriodRecital AKnowledgeExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALessesSection 1.11(c)Material Adverse EffectExhibit ALossesSection 3.19(a)Metarial Adverse EffectSchibit AMetarial Company ContractsSection 3.19(a)Metarial Company ContractsSection 3.10(a)Metared EffectSchibit AMetared ContractsSccion 3.10(a)Metared EffectSchibit AMetared EffectSchibit AMetared EffectSchibit AMetared EffectSchibit AMetared EffectSccion 3.80(ii)Metared	immediate family	Exhibit A
IndemnitorsSection 11.2Independent ExpertSection 2.5Information Privacy and Security LawsExhibit AInsiderExhibit AInsurance PoliciesExhibit AInterim PeriodSection 2.5Interim PeriodSection 7.1Interim PeriodSection 7.1Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALosseSection 11.1(c)Material Adverse EffectSection 3.1(a)Material Adverse EffectSection 3.1(a)Material Company ContractsSection 3.1(a)Merger ProposalSection 3.1(a)Merger SubSection 3.1(a)Most Recent Balance Sheet DateSection 3.1(a)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 12.4(c)Motif Class A Ordinary SharesExhibit AParent Class A Ordinary SharesExhibit AParent Class A Ordinary SharesExhibit AParent MarantsExhibit AParent MarantsExhibit AParent MarantsExhibit AParent MarantsExhibit AParent Class A Ordinary SharesExhibit AParent MarantsExhibit AParent Class A Ordinary SharesExhibit AParent Class A Ordinary SharesExhibit AParent MarantsExhibit AParent MarantsExhibit AParent MarantsExhibit AParent	Indemnification Pool	Exhibit A
Independent ExpertSection 2.4(c)Indemnification Privacy and Security LawsExhibit AInsurance PoliciesExhibit AInsurance PoliciesExhibit AInteillectual PropertyExhibit AInteilnectual PropertyExhibit ALegal RequirementsExhibit ALossesSection 1.1 (c)Material Adverse EffectExhibit AMaterial Company ContractsSection 2.4(a)Material Company ContractsSection 3.19(a)Merger ProposalSection 3.19(a)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Notice of ClaimSection 1.2Notice of ClaimSection 1.2Notice of ClaimSection 1.2Notice of ClaimSection 1.2Parent Class A Ordinary SharesExhibit AParent PlaneExhibit AParent Plane	Indemnitees	Section 11.1(b)
Indemnification Escrow AgreementSection 2.5Information Privacy and Security LawsExhibit AInsiderExhibit AInsurace PoliciesExhibit AIntellectual PropertyExhibit AIntellectual PropertyRexital AIntellectual PropertyExhibit AIntellectual PropertyExhibit AIntellectual PropertyExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALossesSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Merger ProposalSection 3.19(a)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 1.2Notice of ClaimSection 1.2Outside DateSection 1.2Outside DateSection 1.2Parent Class A Ordinary SharesExhibit AParent ProposalSection 1.2Parent PreamblePreambleParent VartasExhibit AParent PreamblePreambleParent PreambleExhibit AParent PreambleExhibit AParent PreambleExhibit AParent PreambleExhibit AParent PreambleExhibit AParent PreamblePreambleParent PreamblePreambleParent PreambleExhibit AParent PreamblePreambleParent PreambleExhibit AParent	Indemnitors	Section 11.2
Information Privacy and Security LawsExhibit AInsiderExhibit AInsurance PoliciesExhibit AIntellectual PropertyExhibit AInterim PeriodSection 7.1Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALossesSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Material Company ContractsSection 3.19(a)MergerRecital AMerger ProposalSection 3.16(a)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Notice of ClaimSection 1.2Notice of ClaimSection 1.2Outside DateSection 1.2Parent UasSection 1.2Notice of ClaimSection 3.8(a)(ii)Parent Class A Ordinary SharesExhibit AParent UasSection 3.8(a)(ii)Parent VarantisExhibit AParent VarantisExhibit AParent VarantisExhibit AParent VarantisExhibit AParent VarantisExhibit AParent Class A Ordinary SharesExhibit AParent VarantisExhibit AParent VarantisSection 3.7Parent Parent Soliti ASection 3.7Parent MarantisExhibit AParent MarantisExhibit AParent MarantisSection 3.14(b)Parent Marantis <t< td=""><td>Independent Expert</td><td>Section 2.4(e)</td></t<>	Independent Expert	Section 2.4(e)
InsiderExhibit AInsurance PoliciesExhibit AInterim PeriodExhibit AInterim PeriodSection 7.1Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALegal RequirementsExhibit ALossesSection 1.1(c)Material Averse EffectExhibit AMaterial Averse EffectSection 2.1(a)Meterial Company ContractsSection 3.19(a)Meterial Company ContractsSection 3.19(a)Merger ProposalSection 3.19(a)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Nost Recent Balance Sheet DateSection 3.8(a)(ii)Notice of ClaimSection 1.2Outside DateSection 1.2Outside DateSection 3.8(a)(ii)Parent Class A Ordinary SharesExhibit AParent WarrantsExhibit AParent WarrantsExhibit AParent WarrantsExhibit AParent WarrantsExhibit AParent VarrantsExhibit AParent Parent Pa	Indemnification Escrow Agreement	Section 2.5
Insurance PoliciesExhibit AInterim ProjedExhibit AInterim ProjedSection 7.1Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALienExhibit ALossesSection 1.1(rc)Material Adverse EffectExhibit AMaterial Adverse EffectSection 3.19(a)Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 3.10(a)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Notice of ClaimSection 3.8(a)(ii)Notice of ClaimSection 1.12Outside DateSection 3.8(a)(ii)Parent ValueSection 3.8(a)(ii)Parent ValueSection 3.8(a)(ii)Parent ValueSection 3.8(a)(ii)Parent ValueSection 1.2Parent ValueSection 1.2Parent ValueSection 1.2Parent ValueSection 3.8(a)(ii)Parent ValueSection 3.8(a)Parent ValueSection 1.2Parent ValueSection 3.8(a)Parent ValueSection 3.8(a)Parent ValueSection 3.7Personal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Information Privacy and Security Laws	Exhibit A
Intellectual PropertyExhibit AInterim PeriodSection 7.1Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALossesSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Merger ProposalSection 2.4(a)Merger SubSection 3.1(a)(i)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(iii)Most Recent Balance Sheet DateSection 1.2Notice of ClaimSection 1.2Notice of ClaimSection 1.2Notice of ClaimSection 10.1(b)Parent Class A Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParent WarrantsExhibit AParent WarrantsPreambleParent WarrantsPreambleParent MarrantsExhibit AParent MarrantsExhibit A <t< td=""><td>Insider</td><td>Exhibit A</td></t<>	Insider	Exhibit A
Interim PeriodSection 7.1Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALienExhibit ALossesSection 1.1 (c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Measurement DateSection 3.19(a)Merger ProposalSection 8.1(a)(i)Merger SubRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance SheetSection 3.8(a)(ii)Nost Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 5.22Net Present ValueSection 5.22Note of ClaimSection 1.2Outside DateSection 1.2Outside DateSection 1.2Parent Class A Ordinary SharesExhibit AParent VarantsExhibit AParent WarantsExhibit AParent Mares B Ordinary SharesExhibit AParent Mares B Ordina	Insurance Policies	Exhibit A
Irish Companies ActRecital AKnowledgeExhibit ALegal RequirementsExhibit ALienExhibit ALossesSection 1.1 (c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Mesurement DateSection 3.19(a)Merger ProposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Not Recent Balance Sheet DateSection 1.2(c)Notice of ClaimSection 1.2Outside DateSection 1.1(b)ParentPreambleParentPreambleParent Class A Ordinary SharesExhibit AParent PlanExhibit AParent VariatsExhibit AParent VariatsExhibit AParent VariatsExhibit AParent DateExhibit AParent PlanExhibit AParent PlanExhibit AParent DateExhibit AParent DateExhibit AParent DateExhibit AParent DateExhibit AParent PlanExhibit AParent PlanExhibit AParent DateExhibit AParent DateExhibit AParent DateExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit A	Intellectual Property	Exhibit A
KnowledgeExhibit ALegal RequirementsExhibit ALeinenExhibit ALossesSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Measurement DateSection 2.4(a)Merger ProposalSection 8.10(i)Merger SubPreambleTransactionsRecital AMost Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Notise Of ClaimSection 1.1.2Outside DateSection 1.1.2Notise Of ClaimSection 1.1.2Outside DateSection 1.1.2Parent Class A Ordinary SharesExhibit AParent PlanPreambleParent PlanExhibit AParent PlanExhibit A<	Interim Period	Section 7.1
Legal RequirementsExhibit ALienExhibit ALossesSection 1.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 8.1(a)(i)Merger Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Nost Recent Balance Sheet DateSection 1.2Notide DateSection 1.2Outside DateSection 1.1.2Outside DateSection 1.1.2Parent Class A Ordinary SharesExhibit AParent PlanExhibit AParent PlanExhibit AParent VarantsExhibit AParent VarantsExhibit AParent VarantsExhibit AParent VarantsExhibit AParent PlanExhibit AParent PlanExhibit AParent DateSection 3.7PeremitsSection 3.7PermitsSection 3.7Personal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Irish Companies Act	Recital A
LienExhibit ALossesSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Measurement DateSection 2.4(a)MergerRecital AMerger FoposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Nota Recent Balance Sheet DateSection 1.2(c)Notice of ClaimSection 1.2Notice of ClaimSection 1.2Outside DateSection 1.1.2Parent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanSection 3.7Parent PlanSection 3.7PermitsSection 3.7Permited LienExhibit AParent SonExhibit APersonal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Knowledge	Exhibit A
LossesSection 11.1(c)Material Adverse EffectExhibit AMaterial Company ContractsSection 3.9(a)Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 3.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.2(a)Most Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 1.2Outside Of ClaimSection 1.2Outside DateSection 1.1Outside DateSection 1.1Parent Class A Ordinary SharesExhibit AParent PlanPreambleParent PlanExhibit AParent VarrantsExhibit AParent VarrantsSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Legal Requirements	Exhibit A
Material Adverse EffectExhibit AMaterial Company ContractsSection 3.19(a)Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 3.8(a)(ii)NasdaqSection 2.4(c)Notice of ClaimSection 1.2Outside DateSection 1.2Outside DateSection 1.1.2Outside DateSection 1.1.2Outside DateSection 1.1.2Parent PreamblePreambleParent Class A Ordinary SharesExhibit AParent Class A Ordinary SharesExhibit AParent VaurantsExhibit AParent VaurantsExhibit AParent VaurantsExhibit AParent VaurantsExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent VarrantsExhibit AParent SSection 3.7.0Permited LienExhibit APersonal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Lien	Exhibit A
Material Company ContractsSection 3.19(a)Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance Sheet DateSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 1.8(a)(i)NasdaqSection 5.22Net Present ValueSection 1.2Notice of ClaimSection 1.2Outside DateSection 1.1.2ParentPreambleParent Class A Ordinary SharesExhibit AParent ValueExhibit AParent ValueExhibit AParent PlanExhibit AParent WarrantsExhibit AParent VartuesSection 3.7PermitsSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Losses	Section 11.1(c)
Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 3.8(a)(ii)NasdaqSection 2.22Notice of ClaimSection 1.1.2Outside DateSection 11.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent PlanExhibit AParent VarantsExhibit AParent VarantsExhibit AParent VarantsSection 3.7PermitsSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Material Adverse Effect	Exhibit A
Measurement DateSection 2.4(a)MergerRecital AMerger ProposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 3.8(a)(ii)NasdaqSection 2.22Notice of ClaimSection 1.1.2Outside DateSection 11.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent PlanExhibit AParent VarantsExhibit AParent VarantsExhibit AParent VarantsSection 3.7PermitsSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Material Company Contracts	Section 3.19(a)
Merger ProposalSection 8.1(a)(i)Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 5.22Net Present ValueSection 1.2Notice of ClaimSection 1.2Outside DateSection 1.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanPreambleParent VartiesPreambleParent VartiesPreambleParent VartiesPreambleParent VartiesExhibit AParent PlanExhibit AParent PlanSection 3.7PermitsSection 3.7Permits LienExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Measurement Date	
Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 5.22Net Present ValueSection 2.4(c)Notice of ClaimSection 10.1(b)ParentSection 10.1(b)Parent Class A Ordinary SharesExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent VartusExhibit AParent VartusExhibit AParent SExhibit AParent SExhibit AParent SExhibit AParent SExhibit AParent PlanExhibit AParent PlanEx	Merger	Recital A
Merger SubPreambleTransactionsRecital AMost Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 5.22Net Present ValueSection 2.4(c)Notice of ClaimSection 10.1(b)ParentSection 10.1(b)Parent Class A Ordinary SharesExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent PlanExhibit AParent VartusExhibit AParent VartusExhibit AParent SExhibit AParent SExhibit AParent SExhibit AParent SExhibit AParent PlanExhibit AParent PlanEx	Merger Proposal	Section 8.1(a)(i)
Most Recent Balance SheetSection 3.8(a)(ii)Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 3.8(a)(ii)NasdaqSection 5.22Net Present ValueSection 2.4(c)Notice of ClaimSection 11.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent VarrantsExhibit AParent VarrantsPreamblePatentsSection 3.7Permited LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Merger Sub	Preamble
Most Recent Balance Sheet DateSection 3.8(a)(ii)NasdaqSection 5.22Net Present ValueSection 2.4(c)Notice of ClaimSection 11.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsSection 3.7Permited LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Transactions	Recital A
NasdaqSection 5.22Net Present ValueSection 2.4(c)Notice of ClaimSection 1.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsExhibit AParentsExhibit AParentsExhibit AParentsExhibit AParentsExhibit APermitsSection 3.7PermoneExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Most Recent Balance Sheet	Section 3.8(a)(ii)
Net Present ValueSection 2.4(c)Notice of ClaimSection 11.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParent VarrantsExhibit AParentsExhibit AParent SSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Most Recent Balance Sheet Date	Section 3.8(a)(ii)
Notice of ClaimSection 11.2Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsSection 3.7PermitsSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Nasdaq	Section 5.22
Outside DateSection 10.1(b)ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsExhibit APermitsSection 3.7Permited LienExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Net Present Value	Section 2.4(c)
ParentPreambleParent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParent VarrantsPreamblePatentsExhibit APermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Notice of Claim	Section 11.2
Parent Class A Ordinary SharesExhibit AParent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsExhibit APermitsSection 3.7Permited LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Outside Date	Section 10.1(b)
Parent Class B Ordinary SharesExhibit AParent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsExhibit APermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Parent	Preamble
Parent PlanExhibit AParent WarrantsExhibit AParty / PartiesPreamblePatentsExhibit APermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Parent Class A Ordinary Shares	Exhibit A
Parent WarrantsExhibit AParent WarrantsPreambleParty / PartiesPreamblePatentsExhibit APermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Parent Class B Ordinary Shares	Exhibit A
Party / PartiesPreamblePatentsExhibit APermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Parent Plan	Exhibit A
PatentsExhibit APermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Parent Warrants	Exhibit A
PermitsSection 3.7Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Party / Parties	Preamble
Permitted LienExhibit APersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Patents	Exhibit A
PersonExhibit APersonal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Permits	Section 3.7
Personal Confidential InformationExhibit APersonal PropertySection 3.14(b)	Permitted Lien	Exhibit A
Personal Property Section 3.14(b)	Person	Exhibit A
	Personal Confidential Information	Exhibit A
	Personal Property	Section 3.14(b)
		Section 9.1(i)

vii

Plan     Section 3 12(b)       Power Purchase Agreement     Exhibit A       Portuguese Companies Act     Recital A       Pro Rata Percentage of Closing Consideration     Exhibit A       Pro Rata Percentage of Contingent Consideration     Exhibit A       Pro Rata Portion of Company Consideration     Exhibit A       Pro Rata Portion of Company Consideration     Exhibit A       Pro Rata Portion of Consigent Consideration     Exhibit A       Proxy Statement/Prospectus     Exhibit A       Proxy Statement/Prospectus     Exhibit A       Qualifying Counterparty     Section 3 15(a)       Returns     Section 3 15(a)       Returns     Section 3 23(c)       SEC     Exhibit A       Section 4 a     Section 3 32(c)       SEC     Section 4 a       Section 4 a     Section 4 a       Securities Act     Section 8 4(a)       Signing Form 8-K     Section 8 4(a)       Signing Press Release     Section 1 6       Stark Act     Section 1 6       Stark Act     Section 1 6       Stark Act     Section 1 6       Stare Exchange     Sect	Term	Section
Portuguese Companies ActRecital APro Rata Percentage of Closing ConsiderationExhibit APro Rata Percentage of Contingent ConsiderationExhibit APro Rata Portion of ConsigerationExhibit AProget Contingent ConsiderationSchibit AProsy Statement/ProspectusExhibit AQualifying CounterpartySection 3.16(a)Reviewable DocumentSection 3.16(a)Reviewable DocumentSection 3.2(c)SECExhibit ASection 2.4(b)Section 3.23(c)SECSection 3.16(a)Securities ActSection 3.23(c)SECSection 4.3Share ExchangeSection 8.4(a)Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 1.6SponsorSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)TrasesExhibit ATrasesExhibit ATrasesSection 3.2(a)TrasesSection 3.2(a)Trase Survival PeriodSection 5.2(b)Trase Survival PeriodSection 5.2(b)TrasesSection 1.4(d)TrasesSection 5.2(b)TrasesSection 5.2(b)Trase Survival PeriodSection 5.2(b)Trase Survival PeriodSection	Plan	Section 3.12(b)
Pro Rata Percentage of Closing ConsiderationExhibit APro Rata Percentage of Contingent ConsiderationExhibit APro Rata Portion of Company ConsiderationExhibit APro Rata Portion of ConsigerationExhibit APro Rata Portion of ConsiderationExhibit APro Rata Portion of ConsiderationSection 2.4(a)Proy Statement/ProspectusExhibit AQualifying CounterpartySection 2.4(b)ReturnsSection 2.4(b)ReturnsSection 2.4(b)ReturnsSection 2.4(b)ReturnsSection 2.4(b)ReturnsSection 3.15(a)ReturnsSection 3.15(a)Section 3.20(c)Setion 3.20(c)SECExhibit ASecurities ActSection 8.23Signing Form 8-KSection 8.23Signing Prom 8-KSection 8.23Signing Prom 8-KSection 1.6Stock Escrow AgreementSection 1.6Stork IsoresSection 1.6Stork IsoresSection 1.6Stork IsoresSection 1.4(a)Surviving BVI CompanyRecital ATrasesSection 3.2(a)Surviving BVI CompanySection 1.4(d)TrasesSection 3.2(b)StaresSection 1.4(d)TrasesSection 5.2(b)TrasesSection 5.2(b)Surviving BVI CompanySection 5.2(b)TrasesSection 5.2(b)TrasesSection 5.2(b)TrasesSection 5.2(b)TrasesSection 5.2(b)TrasesSection 5.2(b)	Power Purchase Agreement	Exhibit A
Pro Rata Percentage of Contingent ConsiderationExhibit APro Rata Portion of Closing ConsiderationExhibit APro Rata Portion of Contingent ConsiderationExhibit APro Rata Portion of Contingent ConsiderationExhibit AProject Contingent ConsiderationSection 2.4(a)Proxy StatementProspectusExhibit AQualifying CounterpartySection 2.4(b)ReturnsSection 3.15(a)Reviewable DocumentSection 8.5(a)SanctionsSection 3.2(a)SEC Approval DateSection 3.2(a)Section 3.2(a)Section 3.2(a)Section 3.2(b)Section 3.2(a)Signing Prom 8-KSection 8.4(a)Signing From 8-KSection 8.4(a)Sponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)TrasesExhibit ATrasesExhibit ATrasesSection 1.4(a)Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)Traseston ExpensesExhibit ATraseston ExpensesExhibit ATraseston ExpensesExhibit ATraseston ExpensesSection 5.23(b)Trast Jisbursement SchaleSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)	Portuguese Companies Act	Recital A
Pro Rata Portion of Closing ConsiderationExhibit APro Rata Portion of Comingent ConsiderationExhibit APro Rata Portion of Contingent ConsiderationExhibit AProcy Et Contingent ConsiderationSection 2.4(a)Proxy Statement/ProspectusSection 3.15(a)ReturnsSection 3.15(a)ReturnsSection 3.23(c)SEC Optoval DateSection 3.23(c)SEC Approval DateSection 4.3Securities ActSection 4.3Shareb Class AgreementSection 8.4(a)Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 1.6Stock Escrow AgreementSection 3.2(a)Stock Escrow AgreementSection 3.2(a)Stock Escrow AgreementSection 8.4(b)Stock Escrow AgreementSection 3.2(a)Stock Escrow AgreementSection 3.2(a)Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 3.2(a)Stork Isers AgreementSection 1.6Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 1.4(a)Stork Isers AgreementSection 1.4(a)Stork Isers AgreementSection 1.4(d)Trak TaxesExhibit AThresholdSection 1.4(d)Transaction ExpensesExhibit ATransaction ExpensesExhibit ATransaction ExpensesSection 5.23(b)Transaction ExpensesSection 5.23(a)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 5.3(d)US GAAP <td>Pro Rata Percentage of Closing Consideration</td> <td>Exhibit A</td>	Pro Rata Percentage of Closing Consideration	Exhibit A
Pro Rata Portion of Company ConsiderationExhibit APro Rata Portion of Contingent ConsiderationExhibit AProject Contingent ConsiderationSection 2.4(a)Proxy Statement/ProspectusExhibit AQualifying CounterpartySection 3.15(a)ReturnsSection 3.15(a)Reviewable DocumentSection 3.15(a)SactionsSection 3.23(c)SECExhibit ASection 2.24(b)Section 4.3Share ExchangeRecital AShare ExchangeRecital AShare ExchangeSection 8.23Signing Form 8-KSection 1.6Sponsor AgreementSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)Traseston ExpensesExhibit ATraseston ExpensesSection 5.23(b)Trust Disbursement ScheduleSection 5.23(a)UPO Exchange AgreementSection 5.23(a) <tr< td=""><td>Pro Rata Percentage of Contingent Consideration</td><td>Exhibit A</td></tr<>	Pro Rata Percentage of Contingent Consideration	Exhibit A
Pro Rata Portion of Contingent ConsiderationExhibit AProject Contingent ConsiderationSection 2 4(a)Proxy Statement/ProspectusExhibit AQualifying CounterpartySection 2 4(b)ReturnsSection 3 15(a)Reviewable DocumentSection 3 23(c)SanctionsSection 3 23(c)SECExhibit ASEC Approval DateSection 8 16(b)Secter A 13Section 8 16(b)Share ExchangeRecital AShare ExchangeSection 8 23Signing Press ReleaseSection 8 4(a)Signing Press ReleaseSection 1.6Stock Escrow AgreementSection 1.6Survival PeriodSection 1.14(a)Survival PeriodSection 1.14(a)Survival ReviendSection 1.14(a)TaxesExhibit ATaxesExhibit ATrasetion ExpensesSection 5.23(b)Trast Disbursement ScheduleSection 5.23(b)Trust Disbursement ScheduleSection 5.23(a)UPD Exchange AgreementSection 5.23(a)Color ExpensesSection 5.23(a)Trast Disbursement ScheduleSection 5.23(a)UPD Exchange AgreementSection 5.23(a)Survival RepresesSection 5.23(a)Trust Disbursement ScheduleSection 5.23(a)UPD Exchange AgreementSection 5.23(a)UPD Exchange AgreementSection 5.23(a)UPD Exchange AgreementSection 5.23(a)Survival RepresesSection 5.23(a)Trust Disbursement ScheduleSection 5.23(a)Su	Pro Rata Portion of Closing Consideration	Exhibit A
Project Contingent ConsiderationSection 2.4(a)Proxy Statement/ProspectusExhibit AQualifying CounterpartySection 2.4(b)ReturnsSection 3.15(a)Reviewable DocumentSection 8.5(a)SanctionsSection 3.23(c)SECExhibit ASEC Approval DateSection 8.1(b)Securints SettSection 8.23(c)Share ExchangeRecital AShare ExchangeSection 8.23Signing Form 8-KSection 8.4(a)Signing Pross ReleaseSection 1.6Sponsor AgreementSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.6Survival RollSection 1.4(a)Survival RollSection 1.4(a)Survival RollSection 1.4(a)Survival RollSection 2.2(a)Survival RollSection 2.2(b)TrademarksExhibit ATrademarksSection 5.23(b)TrademarksSection 5.23(b)Trat Disbursement ScheduleSection 5.23(b)Trust FundSection 5.23(b)UPO Exchange AgreementSection 5.23(b)US GAAPSect	Pro Rata Portion of Company Consideration	Exhibit A
Proxy Statement/ProspectusExhibit AQualifying CounterpartySection 2.4(b)ReturnsSection 3.15(a)Reviewable DocumentSection 3.23(c)SanctionsSection 3.23(c)SECExhibit ASEC Approval DateSection 8.1(b)Section 2.12(c)Section 8.1(b)Sections 8.1(b)Section 8.1(b)Sections 8.1(b)Section 8.1(b)Sections 8.23Section 8.23Signing Form 8-KSection 8.23Signing Press ReleaseSection 8.4(a)Signing Press ReleaseSection 1.6Stock Escrow AgreementSection 1.6Stock Iscrow AgreementSection 1.6Stock Iscrow AgreementSection 1.4(a)Storiying BVI CompanyRecital ATax / TaxesSection 1.4(a)TrademarksSection 1.4(d)TrademarksSection 5.23(b)Trust Disbursement ScheduleSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)Suriying BVI CompanySection 5.23(a)Trust Disbursement ScheduleSection 5.23(a)Trust Disbursement ScheduleSection 5.23(a)Trust Disbursement ScheduleSection 5.23(a)Trust Guage AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(b)Suriy FundSection 5.23(a)UPO Exchange Agree	Pro Rata Portion of Contingent Consideration	Exhibit A
Qualifying CounterpartySection 2.4(b)ReturnsSection 3.15(a)Reviewable DocumentSection 8.5(a)SanctionsSection 3.23(c)SECExhibit ASEC Approval DateSection 3.23(c)Securities ActSection 4.3Share ExchangeRecital AShare ExchangeSection 8.23Signing Form 8-KSection 8.4(a)Signing Frees ReleaseSection 1.6Stock Excrow AgreementSection 1.6SubsidiariesSection 1.6Survival PeriodSection 1.4(a)Surviving BVI CompanyRecital ATraesExhibit ATransaction ExpensesExhibit ATransaction ExpensesSection 5.23(b)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)US, GAAPSection 5.7(b)	Project Contingent Consideration	Section 2.4(a)
ReturnsSection 3.15(a)Reviewable DocumentSection 8.5(a)SanctionsSection 3.23(c)SECExhibit ASEC Approval DateSection 8.1(b)Securities ActSection 8.1(b)Securities ActSection 8.1(b)Share ExchangeRecitial AShare ExchangeSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 1.6Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 1.4(a)Sponsor AgreementSection 1.4(a)Stork VargeementSection 1.4(a)Stork IsraesExhibit AStock Escrow AgreementSection 1.4(a)Struving BVI CompanyRecital ATrax/TaxesExhibit ATransaction ExpensesExhibit ATransaction ExpensesExhibit ATransaction ExpensesSection 5.23(b)Trust Disbursement ScheduleSection 5.23(b)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(b)Trust DisbursementsSection 5.23(a)Trust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(b)UPO Exchange AgreementSection 5.23(b)US GAAPSection 5.7(b)	Proxy Statement/Prospectus	Exhibit A
Reviewable DocumentSection 8.5(a)SanctionsSection 3.23(c)SECExhibit ASEC Approval DateSection 8.1(b)Securities ActSection 4.3Share ExchangeRecital AShareholders AgreementSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 1.6Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 1.4(a)Survival PeriodSection 1.4(a)Survival PeriodSection 11.4(a)TrakesExhibit ATrakesExhibit ATrakesExhibit ATrakesSection 5.23(b)Trust Disbursement ScheduleSection 5.23(b)Trust Disbursement ScheduleSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)Survival PeriodSection 5.23(b)Trust FundSection 5.23(a)Trust FundSection 5.23(b)Trust FundSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 5.23(b)US GAAPSection 5.7(b)	Qualifying Counterparty	Section 2.4(b)
SanctionsSection 3.23(c)SECExhibit ASEC Approval DateSection 4.3Securities ActSection 4.3Share ExchangeRecital AShare ExchangeSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)Sponsor bExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6Survival PeriodSection 1.14(a)Surviving BVI CompanyRecital ATrasesExhibit ATrasestion ExpensesExhibit ATrastorion ExpensesExhibit ATrastorion ExpensesExhibit ATrust DisbursementsSection 5.23(b)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 5.23(b)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.23(b)US GAAPSection 5.7(b)	Returns	Section 3.15(a)
SECExhibit ASEC Approval DateSection 8.1(b)Securities ActSection 4.3Share ExchangeRecital AShareholders AgreementSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6Survival PeriodSection 1.6Survival PeriodSection 1.4(a)Survival PeriodSection 1.4(a)Tax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 1.3(d)UPO Exchange AgreementSection 1.3(d)UPO Exchange AgreementSection 5.7(b)	Reviewable Document	Section 8.5(a)
SEC Approval DateSection 8.1(b)Securities ActSection 4.3Share ExchangeRecital AShareholders AgreementSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 1.6Survival PeriodSection 1.4(a)Surviving BVI CompanyRecital ATaxesExhibit AThresholdSection 1.4(d)TrademarksExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust Disbursement ScheduleSection 5.23(a)UPO Exchange AgreementSection 5.23(a)UPO Exchange AgreementSection 5.7(b)	Sanctions	Section 3.23(c)
Securities ActSection 4.3Share ExchangeRecital AShare ExchangeSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 3.2(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)Transaction ExpensesExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 5.7(b)	SEC	Exhibit A
Share ExchangeRecital AShareholders AgreementSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 3.2(a)Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 1.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 5.7(b)	SEC Approval Date	Section 8.1(b)
Shareholders AgreementSection 8.23Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 1.6Survival PeriodSection 1.14(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATratosesement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 1.3(d)US. GAAPSection 1.3(d)	Securities Act	Section 4.3
Signing Form 8-KSection 8.4(a)Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 1.6Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Share Exchange	Recital A
Signing Press ReleaseSection 8.4(b)SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 3.2(a)Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Shareholders Agreement	Section 8.23
SponsorExhibit ASponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 3.2(a)Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		Section 8.4(a)
Sponsor AgreementSection 1.6Stock Escrow AgreementSection 1.6SubsidiariesSection 3.2(a)Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Signing Press Release	Section 8.4(b)
Stock Escrow AgreementSection 1.6SubsidiariesSection 3.2(a)Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		
SubsidiariesSection 3.2(a)Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(a)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Sponsor Agreement	Section 1.6
Survival PeriodSection 11.4(a)Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(a)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~
Surviving BVI CompanyRecital ATax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		Section 3.2(a)
Tax / TaxesExhibit AThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		
ThresholdSection 11.4(d)TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		
TrademarksExhibit ATransaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		Exhibit A
Transaction ExpensesExhibit ATrust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Threshold	Section 11.4(d)
Trust Disbursement ScheduleSection 5.23(b)Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		Exhibit A
Trust DisbursementsSection 5.23(b)Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)		Exhibit A
Trust FundSection 5.23(a)UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Trust Disbursement Schedule	Section 5.23(b)
UPO Exchange AgreementSection 1.3(d)U.S. GAAPSection 5.7(b)	Trust Disbursements	Section 5.23(b)
U.S. GAAP Section 5.7(b)	Trust Fund	Section 5.23(a)
Warrant Agreement Section 1.3(b)		
	Warrant Agreement	Section 1.3(b)

viii

## AMENDED AND RESTATED BUSINESS COMBINATION AGREEMENT

THIS AMENDED AND RESTATED BUSINESS COMBINATION AGREEMENT is made and entered into as of August 25, 2020, by and among HL Acquisitions Corp., a British Virgin Islands business company ("<u>HL</u>"), Fusion Welcome – Fuel, S.A., a public limited company domiciled in Portugal, *sociedade anónima* (the "<u>Company</u>"), Fusion Fuel Green Limited, formerly known as Dolya Holdco 3 Limited, a private limited company domiciled in Ireland ("<u>Parent</u>"), Fusion Fuel Atlantic Limited, a British Virgin Islands business company and wholly owned subsidiary of Parent ("<u>Merger Sub</u>"), and the shareholders of the Company set forth on the signature pages hereto ("<u>Company Shareholders</u>"). The term "<u>Agreement</u>" as used herein refers to this Amended and Restated Business Combination Agreement, as the same may be amended from time to time, and all schedules hereto (including the Company Schedule and the HL Schedule, as defined in the preambles to <u>Articles III</u> and <u>V</u> hereof, respectively). Each of Parent, HL, Merger Sub, the Company, and the Company Shareholders are referred to herein, individually, as a "<u>Party</u>" and, collectively, as the "<u>Parties</u>". Except as otherwise indicated, capitalized terms used herein and not otherwise defined shall have the meanings ascribed to them in <u>Exhibit A</u>.

## RECITALS

A. The Parties entered into that certain Business Combination Agreement dated as of June 6, 2020 (the "<u>Original Agreement</u>") pursuant to which, upon the terms and subject to the conditions therein and in accordance with the Business Companies Act, 2004 of the British Virgin Islands (the "<u>BVI</u> <u>Companies Act</u>") the Commercial Companies Code of Portugal, Decree-Law No. 262/86 – Official Gazette No. 201/1986 ("<u>Portuguese Companies Act</u>"), and the Companies Act 2014 of Ireland (the "<u>the Irish Companies Act</u>" and together with the BVI Companies Act and the Portuguese Companies Act, the "<u>Corporate Law</u>"), the Parties intend to enter into a business combination transaction by which (i) Merger Sub will merge with and into HL (the "<u>Merger</u>") with HL being the surviving entity of the Merger and becoming a wholly-owned subsidiary of Parent ("<u>Surviving BVI Company</u>"), followed immediately by (ii) the acquisition by Parent of all of the issued and outstanding Company Ordinary Shares and Company Class A Shares (the "<u>Share Exchange</u>", and together with the Merger, the "<u>Transactions</u>"), on the terms and subject to the conditions set forth in this Agreement.

B. Parent is a newly formed entity formed for the purpose of consummating the Transactions, and upon completion of the Transactions and subject to the conditions set forth herein, Parent will become the sole direct owner of the Company and the Surviving BVI Company.

C. The boards of directors of each of HL and the Company has approved this Agreement and determined that this Agreement and the Transactions and other transactions contemplated hereby are fair to, and in the best interests of, their respective companies and their respective shareholders.

D. The Parties intend, for U.S. federal income tax purposes, that the Merger, taken together with the Share Exchange, shall constitute a transaction that qualifies as a reorganization governed by Section 368 of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>") or an exchange governed by Section 351 of the Code.

E. The Company intends to (i) amend and restate its articles of association to, among other things, create the Company Class A Shares as a new class of shares and set forth the rights and preferences of the Company Class A Shares (the "<u>Amended and Restated Articles of Association</u>") and (ii) issue an aggregate of 100 Company Class A Shares to certain Company Shareholders; the Company hereby requests the written consent of HL and a waiver from HL of any breach of Sections 7.1(d), (f), and (g) of the Original Agreement that would result therefrom; and HL desires to provide such consent and waiver;

F. The Parties desire to amend and restate the Original Agreement to implement the terms, conditions, agreements and understandings set forth herein.

NOW, THEREFORE, in consideration of the covenants, promises and representations set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

## ARTICLE I

#### THE MERGER

1.1 The Merger. On the Closing Date, and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the BVI Companies Act, Merger Sub shall merge with and into HL, the separate corporate existence of Merger Sub shall cease, and HL shall continue as the Surviving BVI Company after the Merger and as a wholly owned subsidiary of Parent. The Merger will be consummated immediately upon the filing of Articles of Merger with the Registrar of Companies of the British Virgin Islands, or at such other time as may be agreed by HL and the Company in writing and specified in such filings (the "Effective Time"). The effect of the Merger will be as provided in this Agreement, the Articles of Merger, and the applicable provisions of the BVI Companies Act. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, by virtue of the Merger and without any further action on the part of the Parties or the holders of any of the securities of HL, all of the property, rights, privileges, powers, franchises, debts, liabilities, and duties of HL and Merger Sub shall vest in the Surviving BVI Company.

1.2 <u>Governing Documents</u>. At the Effective Time, the Memorandum and Articles of Association of Merger Sub shall become the Memorandum and Articles of Association of the Surviving BVI Company.

1.3 Effect on Securities. Subject to the terms and conditions of this Agreement, at the Effective Time, by virtue of the Merger and without any further action on the part of Parties or the holders of any of the securities of HL, the following shall occur:

(a) <u>Conversion of HL Ordinary Shares</u>. Other than any shares to be canceled pursuant to <u>Section 1.3(g)</u> and any HL Ordinary Shares owned by HL Shareholders who have validly elected to receive a portion of the proceeds held in HL's Trust Fund in exchange for the surrender of HL Ordinary Shares (each, a "<u>Converting Shareholder</u>"), each HL Ordinary Share issued and outstanding immediately prior to the Effective Time will be automatically converted, at the Effective Time, into one Parent Class A Ordinary Share (the "<u>HL Merger Shares</u>"). From and after the Effective Time, each certificate or book entry position that evidenced HL Ordinary Shares immediately prior to the Merger shall entitle the holder to the applicable number of HL Merger Shares into which such certificate or book entry position is convertible according to this <u>Section 1.3(a)</u> and all HL Ordinary Shares shall no longer be outstanding and shall automatically cease to exist; provided, however, that each HL certificate or book entry position owned by a Converting Shareholder to receive only such portion of the Trust Fund as provided for in HL's Charter Documents.

(b) <u>Adjustment of HL Warrants</u>. At the Effective Time, each outstanding warrant to purchase one HL Ordinary Share (each, an "<u>HL</u> <u>Warrant</u>") shall remain outstanding but shall be automatically adjusted to become a warrant to purchase one Parent Class A Ordinary Share (each, an "<u>HL</u> <u>Parent Warrant</u>"). Each such HL Parent Warrant will continue to have, and be subject to, the same terms and conditions set forth in the Warrant Agreement between HL and Continental Stock Transfer & Trust Company as the warrant agent ("<u>Warrant Agreement</u>") immediately prior to the Effective Time (including any repurchase rights and cashless exercise provisions), except that each HL Parent Warrant will be exercisable (or will become exercisable in accordance with its terms) for that number of whole Parent Class A Ordinary Shares equal to the number of HL Ordinary Shares that were issuable upon exercise of such HL Warrant immediately prior to the Effective Time.

(c) <u>Conversion of HL Rights</u>. Immediately prior to the Effective Time, each outstanding right exchangeable for one tenth of one HL Ordinary Share (each, an "<u>HL Right</u>") will be automatically exchanged for one tenth of one HL Ordinary Share in accordance with the terms of the HL Rights. At the Effective Time, each HL Ordinary Share issued upon the exchange of an HL Right shall be automatically converted into one Parent Class A Ordinary Share according to <u>Section 1.3(a)</u>.

(d) <u>Exchange of UPO</u>. Simultaneously with the execution of this Agreement, HL and EarlyBirdCapital, Inc. ("<u>EBC</u>") shall enter into an agreement (the "<u>UPO Exchange Agreement</u>") pursuant to which EBC shall agree, on behalf of itself and the other holders of the HL UPOs, to exchange such HL UPOs for an aggregate of 50,000 HL Ordinary Shares. At the Effective Time, each HL Ordinary Share issued upon exchange of an HL UPO shall be automatically converted into Parent Class A Ordinary Shares according to <u>Section 1.3(a)</u>.

(e) <u>Adjustments to Merger Consideration</u>. The number of Parent Class A Ordinary Shares issuable pursuant to this <u>Section 1.3</u> shall be equitably adjusted to reflect appropriately the effect of any stock split, reverse stock split, stock dividend (including any dividend or distribution of securities convertible into Parent Class A Ordinary Shares), extraordinary cash dividend, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to Parent Class A Ordinary Shares occurring on or after the date hereof but at or prior to the Effective Time.

(f) <u>Fractional Shares</u>. No fraction of a Parent Class A Ordinary Share will be issued by virtue of the Merger or by virtue of the exchange of HL Rights provided for in <u>Section 1.3</u>, and each holder of HL Ordinary Shares or HL Rights who would otherwise be entitled to a fraction of a Parent Class A Ordinary Share at any time Parent Class A Ordinary Shares are distributed to any such Person pursuant to this Agreement (after aggregating all fractional shares that otherwise would be received by such holder in connection with such distribution) shall receive from Parent, in lieu of such fractional share, one (1) Parent Class A Ordinary Share.

(g) <u>Cancellation of Treasury and Parent-Owned Stock</u>. Each HL Ordinary Share, HL Warrant, and HL Right held by the Company, Parent, Merger Sub, or HL or any direct or indirect wholly owned subsidiary of any of the foregoing immediately prior to the Effective Time shall be canceled and extinguished without any conversion or payment in respect thereof.

(h) <u>Conversion of Merger Sub Stock into Stock of Surviving BVI Corporation</u>. Each share of capital stock of Merger Sub outstanding immediately prior to the Effective Time shall be converted into and become one share of common stock of the Surviving BVI Corporation with the same rights, powers and privileges as the shares so converted. From and after the Effective Time, each share of capital stock of Merger Sub shall no longer be outstanding and shall automatically be cancelled and cease to exist.

1.4 <u>Tax Consequences</u>. It is intended by the Parties hereto that the Transactions together will be treated as a reorganization governed by Section 368 of the Code or an integrated exchange governed by Section 351 of the Code.

1.5 <u>Taking of Necessary Action</u>; <u>Further Action</u>. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest Surviving BVI Company with full right, title and possession to all assets, property, rights, privileges, powers, and franchises of HL and Merger Sub, the officers and directors of Parent, HL and Merger Sub will take all such lawful and necessary action.

1.6 <u>Sponsor Agreement</u>. As soon as practicable following the execution of this Agreement, and in any event prior to the SEC Approval Date, Parent, HL and the Sponsors shall enter into an agreement (the "<u>Sponsor Agreement</u>") providing that, immediately prior to the Effective Time, the Sponsors will forfeit, and HL will terminate and cancel, pro rata, an aggregate of 125,000 HL Ordinary Shares and 125,000 HL Warrants. Immediately prior to the Effective Time, HL and the Sponsors shall jointly notify Continental, HL's transfer agent and the escrow agent acting pursuant to the escrow agreement entered into with HL and the Sponsors on June 27, 2018 ("<u>Stock Escrow Agreement</u>"), of the forfeiture and cancellation of HL Ordinary Shares and HL Warrants pursuant to this <u>Section 1.6</u>.

#### ARTICLE II

#### THE SHARE EXCHANGE

#### 2.1 The Share Exchange.

(a) <u>The Share Exchange</u>. Immediately after the Effective Time, and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the Portuguese Companies Act and the Irish Companies Act, each Company Shareholder shall assign, transfer, and deliver, free and clear of all Liens, all of the Company Ordinary Shares and Company Class A Shares issued and outstanding on the Closing Date, to Parent. In consideration of the sale and transfer of the Company Ordinary Shares to Parent, on the Closing Date, upon surrender by a Company Shareholder, and such Company Shareholder shall be entitled to receive, its Pro Rata Portion of Closing Consideration (subject to the Parent Class B Ordinary Shares to Parent, on the Closing Date, upon surrender by a Company Shareholder, and such Company Shareholder shall be entitled to receive, its Pro Rata Portion of Closing Consideration (subject to the Parent Class B Ordinary Shares comprising the Escrow Fund being placed into escrow in accordance with the terms of <u>Section 2.5</u> and the Indemnification Escrow Agreement). In consideration of the sale and transfer of the Company Class A Shares, each such Contingent Consideration Shareholder to Parent of the certificates (if any) representing the Company Class A Shares, each such Contingent Consideration Shareholder to Parent of the certificates (if any) representing the Company Class A Shares, each such Contingent Consideration Shareholder to Parent of the certificates (if any) representing the Company Class A Shares, each such Contingent Consideration Shareholder shall be entitled to receive its Pro Rata Portion of Contingent Consideration Shareholder to Parent of the certificates (if any) representing the Company Class A Shares, each such Contingent Consideration Shareholder shall be entitled to receive its Pro Rata Portion of Contingent Consideration in accordance with the terms of <u>Section 2.4</u>.

(b) <u>Adjustments to Company Consideration</u>. The Company Consideration issuable pursuant to this <u>Article II</u> shall be equitably adjusted to reflect appropriately the effect of any stock split, reverse stock split, stock dividend (including any dividend or distribution of securities convertible into Parent Class A Ordinary Shares), extraordinary cash dividends, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to Parent Class A Ordinary Shares or Parent Class B Ordinary Shares occurring on or after the date hereof but at or prior to the Effective Time (or, as it relates to the Escrow Fund or Contingent Consideration, prior to the date of respective issuance).

(c) <u>Fractional Shares</u>. No fraction of a Parent Class A Ordinary Share, Parent Class B Ordinary Share or Parent Warrant will be issued by virtue of the Share Exchange (including the Contingent Consideration), and each holder of Company Ordinary Shares and/or Company Class A Shares who would otherwise be entitled to a fraction of a Parent Class A Ordinary Share, Parent Class B Ordinary Share or a Parent Warrant at any time Company Consideration is distributed to any such Person pursuant to this Agreement (after aggregating all fractional shares or warrants that otherwise would be received by such holder in connection with such distribution) shall receive from Parent, in lieu of such fractional share or warrant, one (1) Parent Class A Ordinary Share or one (1) Parent Warrant, as applicable.

(d) <u>Cancellation of Treasury; Parent-Owned Securities and Parent Ordinary Shares</u>. Each Company Ordinary Share and Company Class A Share held by the Company, Parent, HL, or Merger Sub or any direct or indirect wholly owned subsidiary of any of the foregoing immediately prior to the Effective Time and each Parent Class A Ordinary Share and Parent Class B Ordinary Share held by the Company, Parent, HL, or Merger Sub or any direct or indirect wholly owned subsidiary of any of the foregoing or any nominee or representative thereof shall be canceled and extinguished without any conversion or payment in respect thereof.

(e) <u>No Further Ownership Rights</u>. Until assigned, transferred and delivered as contemplated by <u>Section 2.1(a)</u>, the Company Ordinary Shares and Company Class A Shares shall be deemed, from and after the Effective Time, to represent only the right to receive a Pro Rata Portion of Closing Consideration or a Pro Rata Portion of Contingent Consideration, respectively, and any dividends or other distributions as contemplated by <u>Section 2.1(b)</u>. If, after the Effective Time, certificates representing Company Ordinary Shares and/or Company Class A Shares are presented to Parent or the Company for any reason, such Company Ordinary Shares and/or Company Class A Shares shall be exchanged as provided in <u>Section 2.1(a)</u>.

(f) Lost, Stolen, or Destroyed Certificates. In the event that any Company Shareholder's certificates representing Company Ordinary Shares shall have been lost, stolen, or destroyed, upon the making of an affidavit of that fact by the holder thereof in a form satisfactory to Parent, Parent shall issue to each such Company Shareholder its Pro Rata Portion of Closing Consideration into which the Company Ordinary Shares formerly represented by such certificates were exchanged and any dividends or distributions payable pursuant to <u>Section 2.1(b)</u>; provided, however, that, as a condition precedent to the delivery of each such Pro Rata Portion of Closing Consideration, such Company Shareholder shall indemnify Parent against any claim that may be made against Parent or the Company with respect to the certificates alleged to have been lost, stolen, or destroyed. In the event that any Contingent Consideration Shareholder's certificates representing Company Class A Shares shall have been lost, stolen, or destroyed, upon the making of an affidavit of that fact by the holder thereof in a form satisfactory to Parent, Parent shall allot to each such Contingent Consideration Shareholder its right to receive its Pro Rata Portion of Contingent Consideration pursuant to <u>Section 2.4</u> into which the Company Class A Shares formerly represented by such certificates were exchanged, provided, however, that, as a condition precedent to the allotment of the right to receive each such Pro Rata Portion of Contingent Consideration Shareholder is a condition precedent to the allotment of the right to receive each such Pro Rata Portion of Contingent Consideration Shareholder is a condition precedent to the allotment of the right to receive each such Pro Rata Portion of Contingent Consideration Shareholder shall indemnify Parent against any claim that may be made against Parent or the Company with respect to the certificates alleged to have been lost, stolen, or destroyed.

2.2 <u>Tax Consequences</u>. It is intended by the Parties hereto that the Transactions together will be treated as a reorganization governed by Section 368 of the Code or an integrated exchange governed by Section 351 of the Code.

2.3 <u>Taking of Necessary Action</u>; <u>Further Action</u>. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement, the officers and directors of Parent, the Company, the Company Shareholders, and the Company Shareholder Representative will take all such lawful and necessary action.

## 2.4 Contingent Consideration.

(a) The Company Shareholders holding Company Class A Shares (the "<u>Contingent Consideration Shareholders</u>") shall be issued their Pro Rata Portion of Contingent Consideration earnable with respect to each Power Purchase Agreement entered into between the Company (or an Affiliate of the Company) and a Qualifying Counterparty on or before June 30, 2022 (the Contingent Consideration earned in respect of each Power Purchase Agreement, the "<u>Project Contingent Consideration</u>"), upon (x) the satisfaction of the conditions below (each condition an "<u>Earnout Condition</u>" and the date the Earnout Condition is satisfied, the "<u>Measurement Date</u>") and (y) delivery of an Exercise Notice by such Contingent Consideration Shareholder pursuant to Section 2.4(f):

(i) two-fifths of the Project Contingent Consideration upon the signing of the Power Purchase Agreement;

and

(ii) one-fifth of the Project Contingent Consideration upon commencement of operations under the Power Purchase Agreement;

capacity;

(iii) two-fifths of the Project Contingent Consideration after ninety days of operation at ninety-five percent (95%) of nameplate

<u>provided</u>, that, following the achievement of all three Earnout Conditions with respect to any one Power Purchase Agreement, all Project Contingent Consideration will be deemed earned and payable for each other subsequent Power Purchase Agreement upon the signing of the respective Power Purchase Agreement (and will be immediately payable with respect to any Power Purchase Agreement already signed as of such date).

(b) As used herein, a "<u>Qualifying Counterparty</u>" means an entity listed in Schedule 2.4(b), which schedule may be modified or amended by mutual consent of the Company and HL from time to time up to the Closing.

(c) The aggregate number of Parent Class A Ordinary Shares and Parent Warrants earnable as Project Contingent Consideration with respect to each project with a Qualifying Counterparty shall be equal to the *quotient* of (i) twenty percent (20%) of the Net Present Value of the Power Purchase Agreement divided by (ii)  $\in$ 10.73, representing the aggregate agreed value of one Parent Class A Ordinary Share and one Parent Warrant. The "<u>Net Present Value</u>" of a Power Purchase Agreement shall be equal to (x) the sum of the projected unlevered free cash flows of the project each year, using a discount rate of seven percent (7%), *less* (y) the projected initial investment for the project, assuming a two percent (2%) management fee and no contingency.

(d) The aggregate Contingent Consideration issuable hereunder shall not exceed 1,137,000 Parent Class A Ordinary Shares and 1,137,000 Parent Warrants, which the Parties agree represents an aggregate of €61 million of Net Present Value of Power Purchase Agreements.

(e) At least five Business Days prior to the execution of a Power Purchase Agreement with a Qualifying Counterparty, Parent shall provide its calculation of Project Contingent Consideration with respect thereto to the HL Representative, which calculation shall be conclusive and binding upon the Parties unless, within ten Business Days after its receipt of such calculation, the HL Representative notifies Parent in writing that it disputes any of the amounts set forth therein, specifying the nature of the dispute and the basis therefor. Parent and the HL Representative shall in good faith attempt to resolve any dispute and amend the aggregate Project Contingent Consideration to the extent necessary to reflect the resolution of the dispute. If Parent and the HL Representative do not reach agreement in resolving the dispute within ten Business Days after the dispute notice is given to Parent by the HL Representative, Parent and the HL Representative shall submit the dispute to Grant Thornton LLP or, if Grant Thornton LLP is unable to serve, Parent and the HL Representative shall appoint by mutual agreement an independent valuation firm with expertise in the energy sector (the "Independent Expert"). Within thirty 30 days of such submission, the Independent Expert shall determine (it being understood that in making such determination, the Independent Expert shall be functioning as an expert and not as an arbitrator), based solely on written submissions by Parent and the HL Representative, and not by independent review, only those issues in dispute and shall render a written report as to the resolution of the dispute and the resulting Project Contingent Consideration which shall be conclusive and binding. All expenses relating to the engagement of the Independent Expert shall be borne by Parent. Parent and the HL Representative shall make available to the Independent Expert such books, records, work papers and other information as the Independent Expert may request and that are available to that party or its agents and shall be afforded the opportunity to present to the Independent Expert any material relating to the disputed issues and to discuss issues with the Independent Expert (provided, that Parent and the HL Representative shall not, and shall each cause its representatives not to, engage in any ex parte communications with the Independent Expert during the term of its engagement).

(f) Parent shall notify the Contingent Consideration Shareholders of Parent's intention to enter into its first Power Purchase Agreement with a Qualifying Counterparty at least five Business Days prior to the execution thereof, and each Contingent Consideration Shareholder shall deliver a notice to Parent exercising its right to be issued its Pro Rata Portion of the aggregate Contingent Consideration which may become payable with respect to all power Purchase Agreements in accordance with this Section 2.4 (each, an "Exercise Notice"). If an Earnout Condition has been satisfied and a Contingent Consideration Shareholder has delivered an Exercise Notice, the Project Contingent Consideration earned with respect to such Earnout Condition will be issued by Parent to such Contingent Consideration Shareholder after the applicable Measurement Date (or, if an Exercise Notice is delivered by such Contingent Consideration Shareholder after the applicable Measurement Date, then within five (5) Business Days of receipt by Parent of an Exercise Notice; it being understood that no Contingent Consideration Shareholder shall lose the right to receive any earned Contingent Consideration is earnable, Parent shall keep available for issuance a sufficient number of unissued Parent Class A Ordinary Shares (including Parent Class A Ordinary Shares underlying the Parent Warrants forming a part of the Contingent Consideration) to permit Parent to satisfy its issuance obligations set forth in this Section 2.4 and shall take all actions required to increase the authorized number of Parent Class A Ordinary Shares if at any time there shall be insufficient authorized unissued shares to permit such reservation.

(g) The Parties understand and agree that (i) the right to receive any Contingent Consideration shall not be represented by any form of certificate or other instrument, but shall be allotted by Parent on Closing and Parent shall enter the name of each Contingent Consideration Shareholder allotted such right in a register maintained by Parent, (ii) the right to receive any Contingent Consideration is not transferable except by operation of Legal Requirements relating to descent and distribution, divorce and community property, and does not constitute an equity or ownership interest in Parent, and (iii) the Contingent Consideration Shareholders shall not have any rights as a securityholder of Parent as a result of the Contingent Consideration Shareholders' right to receive any Contingent Consideration hereunder.

2.5 Indemnification Escrow. As the sole remedy of the Indemnitees for the indemnity obligations of the Company Shareholders set forth in <u>Article</u> <u>XI</u>, at the Closing, Parent shall deposit in escrow an aggregate of 212,500 Parent Class B Ordinary Shares (the "<u>Escrow Fund</u>"), which will be allocated among the Company Shareholders in the same proportions as the total Closing Consideration is allocated among them, all in accordance with the terms and conditions of an escrow agreement to be entered into at the Closing between Parent, the Company, the Company Shareholder Representative, the Surviving BVI Company, the HL Representative, and Continental Stock Transfer & Trust Company ("<u>Continental</u>") as escrow agent (the "<u>Indemnification Escrow Agreement</u>"). The Indemnification Escrow Agreement will provide that, on the tenth (10th) Business Day after Parent files its annual report for the year ending December 31, 2021 (the "<u>Escrow Termination Date</u>"), subject to any holdback for unresolved claims as provided in the Indemnification Escrow Agreement, Continental will release the Escrow Fund, less that portion of the Escrow Fund applied in satisfaction of or reserved with respect to indemnification claims made prior to such date, to the Company Shareholders in the same proportions as originally deposited into escrow.

2.6 <u>Closing</u>. Unless this Agreement shall have been terminated pursuant to <u>Section 10.1</u>, the consummation of the Transactions (the "<u>Closing</u>"), other than the filing of the Articles of Merger with the Registrar of Companies of the British Virgin Islands with respect to the Merger, shall take place at the offices of Graubard Miller, counsel to HL, The Chrysler Building, 405 Lexington Avenue, 11th Floor, New York, New York 10174-1901 at a time and date to be specified by HL and the Company, which shall be no later than the fifth (5<sup>th</sup>) Business Day after the satisfaction or waiver of the conditions set forth in <u>Article IX</u>, or at such other time, date and location as HL and the Company hereto agree in writing (the "<u>Closing Date</u>"). Closing signatures may be transmitted by facsimile or by email.pdf files.

2.7 <u>HL Representative</u>. HL has designated Jeffrey Schwarz as the initial representative (the "<u>HL Representative</u>") to represent the interests of the HL Shareholders after the Closing for purposes of approving amendments to this Agreement and the Indemnification Escrow Agreement, giving consents and approvals hereunder and thereunder, and making those determinations hereunder and thereunder that are specifically reserved to the HL Representative by the terms hereof and thereof. If such Person ceases to serve in such capacity, for any reason, HL (or, following the Closing, those members of the board of directors of Parent who were members of the board of directors of HL prior to the Closing) shall appoint its successor. The HL Representative shall (i) have no liability to Parent, HL, any Subsidiary or Affiliate of the foregoing or any equityholder of any of the foregoing (including any HL Shareholder) with respect to actions taken or omitted to be taken in its capacity as the HL Representative, and (ii) be entitled to indemnification by Parent against any loss, liability, or expenses arising out of actions taken or omitted to be taken in its capacity as the HL Representative.

2.8 <u>Company Shareholder Representative</u>. The Company Shareholders have designated Fusion Welcome, S.A., a public limited company domiciled in Portugal, as the initial representative (the "<u>Company Shareholder Representative</u>") to represent the interests of the Company Shareholders after the Closing for purposes of approving amendments to this Agreement and the Indemnification Escrow Agreement, giving consents and approvals hereunder and thereunder, and making those determinations hereunder and thereunder that are specifically reserved to the Company Shareholder Representative by the terms hereof and thereof. If such Person ceases to serve in such capacity, for any reason, the Company Shareholders (or, following the Closing, those members of the board of directors of Parent who were shareholders of the Company prior to the Closing) shall appoint its successor. The Company Shareholder Representative shall (i) have no liability to Parent, the Company Shareholders, any Subsidiary or Affiliate of the foregoing or any equityholder of any of the foregoing (including any Company Shareholder) with respect to actions taken or omitted to be taken in its capacity as the Company Shareholder Representative, and (ii) be entitled to indemnification by Parent against any loss, liability, or expenses arising out of actions taken or omitted to be taken in its capacity as the Company Shareholder Representative.

## ARTICLE III

#### REPRESENTATIONS AND WARRANTIES REGARDING THE COMPANY

Subject to the exceptions set forth in <u>Schedule 3</u> attached hereto (the "<u>Company Schedule</u>"), the Company hereby represents and warrants to HL as follows:

## 3.1 Organization and Qualification.

(a) The Company is a public limited company, *sociedade anónima*, duly formed, validly existing and in good standing under the laws of Portugal and has the requisite power and authority to own, lease, and operate its assets and properties and to carry on its business as it is now being conducted. The Company is in possession of all franchises, grants, authorizations, licenses, permits, easements, consents, certificates, approvals and orders of or from any Governmental Entity ("<u>Approvals</u>") necessary to own, lease, and operate the properties it purports to own, operate, or lease and to carry on its business as it is now being conducted, except where the failure to have such Approvals would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. Complete and correct copies of the Articles of Association (or other comparable governing instruments with different names) (collectively referred to herein as "<u>Charter Documents</u>") of the Company, as amended and currently in effect, have been made available to HL or HL's counsel.

(b) The Company is duly qualified or licensed to do business as a foreign company and is in good standing in each jurisdiction where the character of the properties owned, leased, or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. Each jurisdiction in which the Company is so qualified or licensed is listed in <u>Schedule 3.1(b)</u>.

#### 3.2 Subsidiaries.

(a) The Company has no direct or indirect subsidiaries other than those listed in <u>Schedule 3.2</u> (the "<u>Subsidiaries</u>"). Except as set forth in <u>Schedule 3.2</u>, the Company owns all of the outstanding equity securities of the Subsidiaries, free and clear of all Liens other than Permitted Liens, either directly or indirectly through one or more other Subsidiaries. Except with respect to the Subsidiaries, the Company does not own, directly or indirectly, any equity or voting interest in any Person and does not have any agreement or commitment to purchase any such interest, and has not agreed and is not obligated to make nor is bound by any written or oral agreement, contract, subcontract, lease, binding understanding, instrument, note, option, warranty, purchase order, license, sublicense, insurance policy, benefit plan, commitment or undertaking of any nature, as of the date hereof or as may hereafter be in effect, under which it may become obligated to make any future investment in or capital contribution to any other entity.

(b) Each Subsidiary that is a corporation is duly incorporated, validly existing and in good standing (or the equivalent thereof) under the laws of its jurisdiction of incorporation (as listed in <u>Schedule 3.2</u>) and has the requisite corporate power and authority to own, lease, and operate its assets and properties and to carry on its business as it is now being conducted. Each Subsidiary that is a limited liability company, *sociedade unipessoal por quotas*, is duly organized or formed, validly existing, and in good standing (or the equivalent thereof) under the laws of its jurisdiction of organization or formation (as listed in <u>Schedule 3.2</u>) and has the requisite limited liability company power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being conducted. Each Subsidiary is in possession of all Approvals necessary to own, lease, and operate the properties it purports to own, operate, or lease and to carry on its business as it is now being conducted to have a Material Adverse Effect on the Company. Complete and correct copies of the Charter Documents of each Subsidiary, as amended and currently in effect, have been made available to HL or HL's counsel.

(c) Each Subsidiary is duly qualified or licensed to do business as a foreign corporation or foreign limited liability company and is in good standing in each jurisdiction where the character of the properties owned, leased, or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company.

## 3.3 Capitalization.

(a) The authorized capital stock of the Company as of the date of this Agreement consists of 50,000 Company Ordinary Shares, all of which Company Ordinary Shares are issued and outstanding, and 100 Company Class A Shares, all of which Company Class A Shares are issued and outstanding. All of the outstanding Company Ordinary Shares and Company Class A Shares are validly issued and fully paid. All of the outstanding Company Ordinary Shares are held by the Company Shareholders. Other than the Company Ordinary Shares and the Company Class A Shares, no Company Securities are issued or outstanding. Other than Company Ordinary Shares and the Company Class A Shares, the Company has no class or series of securities authorized by its Charter Documents.

(b) As of the date of this Agreement, there are no subscriptions, options, warrants, convertible notes, derivative securities, equity securities, or other ownership interests, calls, rights (including preemptive rights), commitments or agreements of any character to which the Company is a party or by which it is bound obligating the Company to issue, deliver, or sell, or cause to be issued, delivered, or sold, or repurchase, redeem, or otherwise acquire, or cause the repurchase, redeemption, or acquisition of, any shares of capital stock or other ownership interests of the Company or obligating the Company to grant, extend, accelerate the vesting of, or enter into any such subscription, option, warrant, equity security, call, right, commitment or agreement.

(c) Neither the Company nor any Subsidiary has any outstanding bonds, debentures, notes or other obligations the holders of which have the right to vote (or which are convertible into or exercisable or exchangeable for securities having the right to vote) with the Company Shareholders on any matter.

(d) Except as contemplated by this Agreement, there are no registration rights, and there is no voting trust, proxy, rights plan, antitakeover plan, or other agreements or understandings, to which the Company is a party or by which the Company is bound with respect to any equity security of the Company.

(e) Except as contemplated by this Agreement, as a result of the consummation of the Transactions, no shares of capital stock, warrants, options, or other securities of the Company are issuable and no rights in connection with any shares, warrants, options or other securities of the Company accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility, or otherwise).

(f) No outstanding Company Ordinary Shares or Company Class A Shares are unvested or subjected to a repurchase option, risk of forfeiture, or other condition under any applicable agreement with the Company.

## 3.4 Authority Relative to this Agreement.

(a) The Company has all necessary power and authority to: (i) execute, deliver and perform this Agreement and each ancillary document that the Company has executed or delivered or is to execute or deliver pursuant to this Agreement, and (ii) carry out the Company's obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby (including the Share Exchange). The execution and delivery of this Agreement by the Company and the consummation by the Company of the transactions contemplated hereby (including the Share Exchange) have been, or will be, duly and validly authorized by all necessary corporate action on the part of the Company (including the approval by its board of directors), and no other proceedings on the part of the Company are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Company and, assuming the due authorization, execution, and delivery thereof by the other Parties, constitutes the legal and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity.

(b) The Company's board of directors has, by unanimous resolution, approved this Agreement, the Share Exchange, and the other transactions contemplated by this Agreement and determined that this Agreement, the Share Exchange, and the other transactions contemplated by this Agreement are advisable and in the best interests of the Company and the Company Shareholders.

## 3.5 No Conflict; Required Filings and Consents. Except as set forth in Schedule 3.5 hereto:

(a) The execution and delivery of this Agreement by the Company do not, and the performance of this Agreement by the Company shall not, (i) conflict with or violate the Charter Documents of the Company or any of its Subsidiaries, (ii) conflict with or violate any applicable Legal Requirements, (iii) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or impair the Company's or any of its Subsidiaries' rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a Lien on any of the properties or assets of the Company or any of its Subsidiaries (other than Permitted Liens) pursuant to, any Company Contracts, or (iv) result in the triggering, acceleration or increase of any payment to any Person pursuant to any Company Contract, including any "change in control" or similar provision of any Company Contract; *except*, with respect to clauses (ii), (iii) and (iv), for any such conflicts, violations, breaches, defaults, impairments, alterations, triggerings, accelerations, increases or other occurrences that would not, individually and in the aggregate, have a Material Adverse Effect on the Company.

(b) The execution and delivery of this Agreement by the Company does not, and the performance of its obligations hereunder will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or other third party (including, without limitation, lenders and lessors), except (i) the consents, approvals, authorizations, and permits described in <u>Schedule 3.5(b)</u>, and (ii) where the failure to obtain such consents, approvals, authorizations, or permits, or to make such filings or notifications, would not, individually or in the aggregate, reasonably be expected to (x) have a Material Adverse Effect on the Company or, after the Closing, Parent, or (y) prevent the consummation of the Transactions or otherwise prevent the Company from performing its obligations under this Agreement on a timely basis.

3.6 <u>Compliance</u>. The Company and each of its Subsidiaries has complied with all, and is not in violation of any, and is conducting its business in compliance with all, applicable Legal Requirements. Neither the Company nor any of its Subsidiaries is in default or violation in any material respect of any term, condition or provision of any applicable Charter Documents. No written notice of non-compliance with any applicable Legal Requirements has been received by the Company or any of its Subsidiaries (and the Company has no knowledge of any such notice delivered to any other Person).

3.7 <u>Permits</u>. (i) The Company and each Subsidiary, as applicable, collectively hold all permits necessary to lawfully conduct in all material respects the business of the Company and each Subsidiary as presently conducted, and to own, lease and operate its assets and properties (collectively, the "<u>Permits</u>"), (ii) all such Permits are in full force and effect, and no suspension or cancellation of any of the Permits is pending or, to the Company's knowledge, threatened, and (iii) all Permits are renewable by their terms in the ordinary course of business. The Company has made available to HL or HL's coursel true, correct and complete copies of all Permits, all of which are listed on <u>Schedule 3.7</u>. Neither the Company nor any Subsidiary, as applicable, is in violation of the terms of any Permit.

## 3.8 Financial Matters.

(a) Financial Statements. HL or its counsel has been furnished with each of the following:

(i) the unaudited balance sheets of Company as of December 31, 2019 and the related unaudited statements of income, cash flow and changes in shareholders' equity of Company for the fiscal years then ended, accompanied by any notes thereto (collectively, the "<u>Company</u> <u>Annual Financial Statements</u>"); and

(ii) the unaudited consolidated balance sheet of the Company for the three month period ended as of March 31, 2020 (the "<u>Most</u> <u>Recent Balance Sheet</u>" and the date thereof, the "<u>Most Recent Balance Sheet Date</u>") and the related unaudited consolidated statement of income of the Company for the quarter then ended (the "<u>Company Interim Financial Statements</u>" and, together with the Company Annual Financial Statements, the "<u>Company Financial Statements</u>").

(b) <u>Compliance with IFRS</u>. The Company Financial Statements (including any notes thereto) (i) accurately reflect in all material respects, (ii) have been prepared, in all material respects, in accordance with IFRS consistently applied, and (iii) fairly present, in all material respects, the consolidated financial position and results of operations of the Company on the dates and for the periods specified herein, all in accordance with IFRS (subject, in the case of the Company Interim Financial Statements, to the absence of statements of cash flows and shareholders' equity and footnotes and, in each case, to year-end and periodic reclassifications and adjustments that are not expected to have a Material Adverse Effect). The Company and its Subsidiaries have never been subject to the reporting requirements of Sections 13(a) and 15(d) of the Securities Exchange Act of 1934, as amended ("Exchange Act").

(c) <u>Absence of Undisclosed Liabilities</u>. Neither the Company nor its Subsidiaries has any liabilities which are of a nature required by IFRS to be reflected in a balance sheet or the notes thereto except for (i) liabilities included in the Most Recent Balance Sheet, (ii) liabilities incurred in the ordinary course of business since the Most Recent Balance Sheet Date or in contemplation of the Transactions or with respect thereto and (iii) liabilities that do not constitute a Material Adverse Effect.

(d) <u>Projections</u>. All financial projections with respect to the Company and its Subsidiaries that were delivered by or on behalf of the Company to HL or its representatives, were prepared by the Company or its representatives in good faith using assumptions that the Company believes to be reasonable.

(e) <u>Auditor</u>. The Company has engaged Marcum LLP as its independent registered public accountants to audit the Company Annual Financial Statements for inclusion in the Proxy Statement/Prospectus. To the Company's knowledge, Marcum LLP has at all required times since the date of enactment of the Sarbanes-Oxley Act been: (i) a registered public accounting firm (as defined in Section 2(a)(12) of the Sarbanes-Oxley Act); (ii) "independent" with respect to the Company and its Subsidiaries within the meaning of Regulation S-X under the Exchange Act; and (iii) in compliance with subsections (g) through (l) of Section 10A of the Exchange Act and the rules and regulations promulgated by the SEC and the Public Company Accounting Oversight Board thereunder.

(f) <u>Controls</u>. The Company has established and maintained a system of internal accounting controls. Such internal controls are sufficient to provide reasonable assurance regarding the reliability of the Company's financial reporting and the preparation of the Company Financial Statements for external purposes in accordance with IFRS.

3.9 <u>Absence of Certain Developments</u>. From the Most Recent Balance Sheet Date to the date hereof, (a) there has not been any change, development, condition or event that constitutes a Material Adverse Effect, (b) the business has been conducted in the ordinary course of business (aside from steps taken in contemplation of the Transactions), and (c) neither the Company nor its Subsidiaries has taken any action that would have required the prior written consent of HL under <u>Section 7.1</u> if such action had been taken after the date hereof and prior to the Closing.

3.10 Condition and Sufficiency of Assets. The Company or one of its Subsidiaries has good and valid title to, or a valid leasehold interest in, or adequate rights to use, all buildings, machinery, equipment, and other tangible assets which are necessary for the conduct of its or their business as currently conducted and are shown on the Interim Financial Statement or acquired after the Most Recent Balance Sheet Date (the "Assets"). The Assets are free and clear of all Liens, except for Permitted Liens, except for Assets disposed of in the ordinary course of business since the Most Recent Balance Sheet Date and except in the case of any non-owned Asset, for Liens contained in the Company Contract to use such Asset. Each Asset has been maintained in the ordinary course of business, is in good operating condition, subject to normal wear and tear, and is suitable for the purposes for which it is currently used.

3.11 Litigation. There have been, and are, no claims, suits, actions or proceedings pending or, to the Company's knowledge, threatened against the Company or any of its Subsidiaries before any court, governmental department, commission, agency, instrumentality or authority, or any arbitrator.

## 3.12 Employment Matters.

(a) As of the date of this Agreement, the Company does not have, and has never had, any employees. Neither the Company nor any of its Subsidiaries is a party to any collective bargaining agreement or other labor union contract applicable to persons employed by the Company or its Subsidiaries nor does the Company have knowledge of any activities or proceedings of any labor union to organize any such employees.

(b) As of the date of this Agreement, the Company does not have any "employee benefit plan" as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended, or any other employee compensation, deferred compensation, incentive, fringe, severance, change in control, retirement, death, disability, medical, or employee benefit plan, program, policy or other arrangement (collectively, "Plans").

(c) Neither the execution and delivery of this Agreement nor the consummation of the Transactions will (i) result in any payment (including severance, bonus or otherwise) becoming due to any shareholder, director, officer or employee of the Company or its Subsidiaries under any Plan or otherwise, (ii) materially increase any benefits otherwise payable under any Plan, or (iii) result in the acceleration of the time of payment or vesting of any such benefits.

(d) The Company and its Subsidiaries are in compliance with all Legal Requirements respecting hiring, employment, termination of employment, employment practices, terms and conditions of employment, employment discrimination, harassment, retaliation, reasonable accommodation, wages and hours, and employee health and safety and are not liable for any arrears of wages or penalties with respect thereto.

(e) There are no pending, or to the Company's knowledge, threatened or reasonably anticipated claims or actions against the Company or any of its Subsidiaries by any former employee in connection with such employee's employment or termination of employment by the Company or any of its Subsidiaries.

3.13 <u>Restrictions on Business Activities</u>. Except as disclosed in <u>Schedule 3.13</u> hereto, there is no agreement, commitment, judgment, injunction, order or decree binding upon the Company or its Subsidiaries or their respective assets or to which the Company or any of its Subsidiaries is a party which has had or would reasonably be expected to have the effect of prohibiting or materially impairing any business practice of the Company or its Subsidiaries, any acquisition of property by the Company or its Subsidiaries or the conduct of business by the Company or its Subsidiaries as currently conducted other than such effects, individually or in the aggregate, which have not had and would not reasonably be expected to have a Material Adverse Effect on the Company.

## 3.14 Title to Property.

(a) Except as set forth in <u>Schedule 3.14(a)</u>, neither the Company nor any Subsidiary owns or leases any real property and there are no options or other contracts under which the Company or any Subsidiary has a right or obligation to acquire or lease any interest in real property.

(b) All material personal property and other material property and assets of the Company and its Subsidiaries owned, used or held for use in connection with the business of the Company and its Subsidiaries (the "<u>Personal Property</u>"), are shown or reflected on the Most Recent Balance Sheet, to the extent required by IFRS applied on a consistent basis in accordance with past practice, other than those entered into or acquired on or after the Most Recent Balance Sheet Date in the ordinary course of business. The Company and its Subsidiaries have good and marketable title to the Personal Property owned by them, and all such Personal Property is in each case held free and clear of all Liens, except for Permitted Liens or Liens disclosed in the Company Financial Statements, none of which Liens would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on such property or on the present use of such property in the businesses of the Company and its Subsidiaries. The Personal Property is structurally sound, in good operating condition, ordinary wear and tear excepted, and is suitable for the uses to which it is being put.

(c) All material leases pursuant to which the Company and/or one of its Subsidiaries leases from others real property or Personal Property are valid and effective in accordance with their respective terms, and there is not, under any of such leases, any existing material default or event of default of the Company or its Subsidiaries or, to the Company's knowledge, any other party (or any event which with notice or lapse of time, or both, would constitute a material default), except where the lack of such validity and effectiveness or the existence of such default or event of default would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the Company.

(d) Each of the Company and its Subsidiaries is in possession of, or has valid and effective rights to, all properties, assets and rights (other than Intellectual Property, which is governed exclusively by <u>Section 3.18</u>) required, in all material respects, for the effective conduct of its business, as it is currently operated and expected to be operated in the future, in the ordinary course.

3.15 Taxes. Except as set forth in Schedule 3.15 hereto:

(a) The Company and its Subsidiaries have timely filed all returns, estimates, information statements and reports relating to Taxes ("<u>Returns</u>") required to be filed by them with any Tax authority prior to the date hereof. All such Returns are true, correct and complete. The Company and its Subsidiaries have paid all Taxes due and payable on such Returns.

(b) All Taxes that the Company and its Subsidiaries are required by law to withhold or collect have been duly withheld or collected and have been timely paid over to the proper Governmental Entity to the extent due and payable.

(c) The Company and its Subsidiaries have not been delinquent in the payment of any Tax nor is there any Tax deficiency outstanding, proposed or assessed by a taxing authority against the Company or any of its Subsidiaries, nor has the Company or any of its Subsidiaries executed any unexpired waiver of any statute of limitations on or extending the period for the assessment or collection of any Tax. The Company and its Subsidiaries have complied in all material respects with all applicable Legal Requirements regarding withholding Taxes and timely has paid over in full to the proper taxing authorities all amounts required to be so withheld and paid over for all periods.

(d) No audit or other examination of any Return of the Company or any of its Subsidiaries by any Tax authority is presently in progress, nor has the Company or any of its Subsidiaries been notified of any request for such an audit or other examination.

(e) No adjustment relating to any Returns filed by the Company or any of its Subsidiaries has been proposed, formally or informally, by any Tax authority to the Company or any of its Subsidiaries or any representative thereof.

(f) Neither the Company nor any of its Subsidiaries has any liability for any unpaid Taxes which has not been accrued for or reserved on the Company's balance sheets included in the Company Financial Statements for the most recent fiscal year ended, other than any liability for unpaid Taxes that may have accrued since the date of such Company Financial Statements in connection with the operation of the business of the Company and its Subsidiaries in the ordinary course of business or any liability for unpaid Taxes incurred in connection with the Transactions.

(g) Neither the Company nor any of its Subsidiaries has taken, intends to take, or has agreed to take any action or is aware of any fact or circumstance that, to the Company's knowledge, would prevent or impede, or would reasonably be expected to prevent or impede, the Transactions from qualifying as a reorganization governed by Section 368 of the Code or an integrated exchange within the meaning of Section 351 of the Code.

3.16 Environmental Matters. Except as would not reasonably be expected, individually or in the aggregate, to result in a material liability of the Company and its Subsidiaries, taken as a whole: (i) the Company and its Subsidiaries have complied in all material respects with applicable Environmental Laws; (ii) none of the Company or its Subsidiaries or, the knowledge of the Company, any third party has caused any properties currently owned, leased or operated by the Company or its Subsidiaries to be contaminated with any Hazardous Substances; (iii) the properties formerly owned, leased or operated by the Company or its Subsidiaries were not contaminated with Hazardous Substances during the period of ownership, leasing or operation by the Company or its Subsidiaries; (iv) as of the date hereof, none of the Company or its Subsidiaries has received notice that it is potentially liable for any Hazardous Substance disposal or contamination on any third party or public property (whether above, on or below ground or in the atmosphere or water); (v) as of the date hereof, none of the Company or its Subsidiaries has received any written notice, demand, letter, claim or request for information alleging that the Company or any Subsidiary may be in material violation of or have material liability under any Environmental Law; and (vi) none of the Company or its Subsidiaries is subject to any orders, decrees, injunctions or other arrangements with any Governmental Entity or subject to any contractual indemnity or other agreement with any third party relating to a material liability under any Environmental Law, including in relation to Hazardous Substances.

3.17 Brokers. Except as set forth in <u>Schedule 3.17</u> hereto, neither the Company nor any of its Subsidiaries has incurred, nor will it incur, and has not entered into any contract, agreement, understanding, arrangement, or commitment pursuant to which Parent or any of its direct or in indirect subsidiaries could incur, directly or indirectly, any liability for brokerage, finders' fees, agent's commissions, or any similar charges in connection with this Agreement or any transactions contemplated hereby.

## 3.18 Intellectual Property.

(a) Schedule 3.18 hereto contains a description of the material Company Registered Intellectual Property.

(b) The Company owns or has enforceable rights to use all Intellectual Property used or held for use in, or necessary for, the conduct of its business as presently conducted. Except as disclosed in <u>Schedule 3.18</u> hereto, no Company Intellectual Property or Company Product is subject to any material proceeding or outstanding decree, order, judgment or stipulation restricting in any manner the use, transfer or licensing thereof by the Company, or which would reasonably be expected to negatively affect the validity or enforceability of such Company Intellectual Property.

(c) Except as disclosed in <u>Schedule 3.18</u> hereto, the Company owns and has good and exclusive title to each item of Company Intellectual Property owned by it free and clear of any Liens (excluding non-exclusive licenses and related restrictions granted by it in the ordinary course of business).

(d) (i) the operation of the business of the Company and its Subsidiaries as such business currently is conducted, including the Company's use of any product, device or process, has not and does not infringe or misappropriate the Intellectual Property of any third party or constitute unfair competition or trade practices, (ii) the Company has not received any claims or threats from third parties alleging any such infringement, misappropriation or unfair competition or trade practices and (iii) to the Company's knowledge, no third party has materially infringed or misappropriate any Company Intellectual Property.

(e) With respect to the collection, use, processing, storage, transfer, and security of Personal Confidential Information, the Company and each of its Subsidiaries have, in the conduct of their businesses, materially complied with applicable Information Privacy and Security Laws. No action is pending or, to the Company's knowledge threatened in writing against the Company or any of its Subsidiaries relating to the processing or security of Personal Confidential Information. The Company and its Subsidiaries have not: (y) experienced any actual, alleged, or suspected material data breach or other security incident that resulted in the unauthorized acquisition of unencrypted computerized data that compromised the security, confidentiality, or integrity of Personal Confidential Information in their possession or control; or (z) been subject to or received any notice of any audit, investigation, complaint, or other action by any Governmental Entity or other Person concerning the Company's or any Subsidiary's collection, use, processing, storage, transfer, or protection of Personal Confidential Information or actual, alleged, or suspected violation of any Information Privacy and Security Laws, and to the Company's knowledge, there are no facts or circumstances that could reasonably be expected to give rise to any such action, in each case except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

## 3.19 Agreements, Contracts and Commitments.

(a) <u>Schedule 3.19</u> sets forth a complete and accurate list of all Material Company Contracts in effect on the date of this Agreement, specifying the parties thereto. As used herein, the term "<u>Company Contracts</u>" means all legally binding contracts, agreements, leases, mortgages, indentures, notes, and bonds, whether written or oral, to which the Company or any of its Subsidiaries is a party or by or to which any of the properties or assets of the Company or any of its Subsidiaries may be bound (including without limitation notes for borrowed money payable to the Company or any of its Subsidiaries), and the term "<u>Material Company Contracts</u>" means each of the following Company Contracts:

(i) any Company Contract (or group of related Company Contracts) for the sale of products or services or for the purchase of products or services which will, by its terms, extend over a period of more than one (1) year after the date hereof;

(ii) any Company Contract pursuant to which a partnership or joint venture was established;

(iii) any Company Contract made other than in the ordinary course of business (x) providing for the grant of any preferential rights of first offer or first refusal to purchase or lease any Asset or (y) providing for any exclusive right to sell or distribute, or otherwise relating to the sale or distribution of, any product or service of the Company or a Subsidiary;

(iv) any Company Contract under which the Company or a Subsidiary has permitted any Asset to become, or to become subject to, a Lien (other than a Permitted Lien);

(v) any Company Contract providing for the employment or consultancy of any Person on a full-time, part-time, consulting or other basis or otherwise providing compensation or other benefits in excess of \$100,000 per year;

(vi) any collective bargaining agreement with any labor union;

(vii) any Company Contract containing covenants that (A) restrict the Company or any Subsidiary from any solicitation, hiring or engagement of any Person or the solicitation of any customer or (B) limit the freedom of the Company, a Subsidiary, or any Affiliate thereof to engage in any line of business or compete with any Person;

(viii) any outstanding general or special powers of attorney executed by or on behalf of the Company or a Subsidiary;

(ix) any Company Contract under which the Company or a Subsidiary has advanced or loaned an amount to, or received a loan, note, or other instrument, agreement, or arrangement for or relating to the borrowing of money from, any of its Affiliates, members, officers, managers, members of the board of managers, or employees, other than in the ordinary course of business;

(x) any Company Contract (or group of related Company Contracts) the performance of which mandates payment of consideration in excess of \$100,000 per annum over the remaining life of such Company Contract, other than (A) any Company Contract that is terminable by the Company or a Subsidiary at will without material liability and on less than ninety (90) days' notice and (B) purchase orders received in the ordinary course of business;

(xi) any guaranty by the Company, a Subsidiary, or any Affiliate of the foregoing, of any obligation of a third party in excess of \$100,000; and

(xii) any obligation to register any Company Securities with any Governmental Authority.

(b) The Company has made available to HL copies of each Material Company Contract that are accurate and complete, in each case, as amended or otherwise modified and in effect.

(c) Each Material Company Contract is in full force and effect and is enforceable against each party to such Material Company Contract. Neither the Company, a Subsidiary, nor, to the Company's knowledge, any other party to any Material Company Contract is in material breach or violation of, or default under, or has repudiated any provision of, any Material Company Contract.

3.20 Insurance. As of the date of this Agreement, neither the Company nor any of its Subsidiaries maintains any insurance policies.

3.21 Interested Party Transactions. Except as set forth in the Schedule 3.21 hereto, (a) no Insider or a member of his or her immediate family is indebted to the Company or any of its Subsidiaries, nor is the Company or any of its Subsidiaries indebted (or committed to make loans or extend or guarantee credit) to any of such Persons, other than (i) for payment of salary for services rendered, (ii) reimbursement for reasonable expenses incurred on behalf of the Company or any of its Subsidiaries, (iii) for other employee benefits made generally available to all employees, and (iv) arms' length relationships between the Company or any of its Subsidiaries, on the one hand, and an Affiliate of an Insider, on the other hand and (b) to the Company's knowledge, no Insider is, directly or indirectly, interested in any Company Contracts (other than such contracts as relate to the acquisition of capital stock or other securities of the Company or any Company Contract of employment).

3.22 Proxy Statement/Prospectus. None of the information relating to the Company or its Subsidiaries to be supplied by the Company, or by any other Person acting on behalf of the Company at its direction, in writing specifically for inclusion in the Proxy Statement/Prospectus will, as of the date the Proxy Statement/Prospectus (or any amendment or supplement thereto) is first mailed to the HL Shareholders, at the time of the HL Special Meeting, or at the Effective Time, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

#### 3.23 Certain Business Practices.

(a) Neither the Company nor any of its Subsidiaries or Affiliates acting on its behalf has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the Foreign Corrupt Practices Act of 1977 or similar law of any other jurisdiction, or (iii) made any other unlawful payment. Neither the Company nor any of its Subsidiaries or Affiliates acting on its behalf has directly or indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder the Company or assist the Company in connection with any actual or proposed transaction.

(b) The operations of the Company are and have been conducted at all times in compliance with money laundering statutes in all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Entity in all material respects, and no action involving the Company with respect to the any of the foregoing is pending or, to the knowledge of the Company, threatened.

(c) Neither the Company nor any of its directors or officers, or, to the knowledge of the Company, any other Person acting on behalf of the Company is currently identified on the specially designated nationals or other blocked person list or otherwise currently subject to any U.S. sanctions administered by or enforced by the United States (including any administered or enforced by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the U.S. Department of State, or the Bureau of Industry and Security of the U.S. Department of Commerce), the United Nations Security Council, the European Union or any member state of the European Union (collectively, "<u>Sanctions</u>"), or is located, organized, or resident in a country or territory that is, or whose government is, the subject of Sanctions that broadly prohibit dealings with that country or territory, or will, directly, or indirectly, use the proceeds of any offering of securities of the Company, Parent, or the Surviving BVI Company, or lend, contribute, or otherwise make available such proceeds to any other Person in any manner that would result in a violation of any Sanctions or result in the imposition of Sanctions against such Person.

(d) The Company is currently in compliance with, and has complied with, all Export Control Laws applicable to it. Without limiting the foregoing: (i) the Company has obtained all material export licenses and other material approvals required for its exports of products required by any Export Control Law and all such approvals and licenses are in full force and effect; (ii) the Company is in material compliance with the terms of such applicable export licenses or other approvals; and (ii) there are no claims pending or threatened in writing against the Company with respect to such export licenses or other approvals.

3.24 No Additional Representations and Warranties; No Reliance. Except as provided in this Article III (as modified by the Company Schedule), neither the Company, any Subsidiary, any of their respective Affiliates, nor any of their respective directors, officers, employees, shareholders, or representatives has made, or is making, any representation or warranty whatsoever to HL or its Affiliates, and no such Party shall be liable in respect of the accuracy or completeness of any information provided to HL or its Affiliates. The Company acknowledges and agrees (on its own behalf and on behalf of its Affiliates and its Representatives) that: (i) it has conducted its own independent investigation of the financial condition, results of operations, assets, liabilities, properties and projected operations of HL; (ii) it has been afforded satisfactory access to the books and records, facilities and personnel of HL for purposes of conducting such investigation; and (iii) except for the representations and warranties set forth in <u>Article V</u> (as modified by the HL Schedule), it is not relying on any representations and warranties from any Person in connection with the transactions contemplated hereby.

3.25 <u>Survival of Representations and Warranties</u>. The representations and warranties of the Company set forth in this Agreement shall survive the Closing as set forth in <u>Section 11.4</u>.

# ARTICLE IV

#### REPRESENTATIONS AND WARRANTIES OF THE COMPANY SHAREHOLDERS

Each Company Shareholder hereby represents and warrants, severally and not jointly, solely with respect to himself or itself in respect of his or its Company Securities, to HL as follows:

# 4.1 Organization, Standing, and Corporate Power.

(a) Each Company Shareholder that is not an individual is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction in which it was formed and has all requisite legal power and authority to carry on its business as now being conducted, except as would not, individually or in the aggregate, reasonably be expected to prevent, delay, or impair the ability of the Company Shareholders to consummate the Transactions.

(b) Each Company Shareholder that is not an individual has all requisite corporate or other legal entity power and authority and has taken all corporate or other legal entity action necessary in order to execute, deliver and perform its obligations under this Agreement and the ancillary agreements to which it is a party and to consummate the Transactions. The execution, delivery and performance by each Company Shareholder that is not an individual of this Agreement and the ancillary agreements to which it is a party, and the consummation by it of the Transactions, have been duly and validly authorized by all necessary corporate consent and authorizations on the part of such Company Shareholder, and no other corporate actions on the part of such Company Shareholder are necessary to authorize the execution and delivery by such Company Shareholder of this Agreement, the ancillary agreements to which it is a party and the consummation by it of the Transactions. This Agreement has been duly executed and delivered by each Company Shareholder and, assuming due authorization, execution and delivery hereof by the other parties, is a legal, valid and binding obligation of each Company Shareholder, enforceable against each Company Shareholder in accordance with its terms (subject to applicable bankruptcy, solvency, fraudulent transfer, reorganization, moratorium and other Laws affecting creditors' rights generally from time to time in effect and by general principles of equity).

(c) The execution, delivery and performance of this Agreement and the ancillary agreements to which each Company Shareholder is a party, and the consummation of the Transactions, do not, and will not, constitute or result in (i) a breach or violation of, or a default under, the Charter Documents of any such Company Shareholder who is not an individual, or (ii) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) of or default under, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of such Company Shareholder pursuant to, any material agreement to which any such Company Shareholder is a party or, assuming (solely with respect to performance of this Agreement and consummation of the Transactions) compliance with the matters referred to in Section 5.3(a), under any Law to which any such Company Shareholder is subject (except Laws that are applicable due to the business, Contracts or licenses of the Company or its Subsidiaries).

4.2 <u>Ownership of Company Ordinary Shares and Company Class A Shares</u>. Each Company Shareholder is the sole legal and beneficial owner of the Company Ordinary Shares and Company Class A Shares attributed to such Company Shareholder in <u>Schedule 4.2</u>. Each Company Shareholder owns the Company Ordinary Shares and Company Class A Shares held by it free and clear of any Liens and does not have outstanding any option, right, or agreement to sell such Company Ordinary Shares to any other Person.

4.3 Investment Intent. Each Company Shareholder is acquiring securities of Parent solely for its own account for investment purposes and not with a view to, or for offer or sale in connection with, any distribution thereof. Each Company Shareholder acknowledges that the Parent Class A Ordinary Shares, Parent Class B Ordinary Shares and Parent Warrants to be issued to the Company Shareholder in the Share Exchange (including the Contingent Consideration) are not registered under the Securities Act of 1933, as amended ("Securities Act") or any state securities laws, and that such securities may not be transferred or sold except pursuant to (i) the registration provisions of the Securities Act or pursuant to an applicable exemption therefrom and subject to state securities laws, as applicable, (ii) the expiration of the lockup period as set forth in Section 8.20 and (iii) the Shareholders Agreement. Each Company Shareholder qualifies as an "accredited investor" as such term is defined in Rule 501(a) promulgated pursuant to the Securities Act. Each Company Shareholder is able to bear the economic risk of holding Parent securities for an indefinite period, including total loss of investment, and has sufficient knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of such investment.

4.4 <u>Litigation</u>. There have been, and are, no claims, suits, actions or proceedings pending or, to the knowledge of such Company Shareholder, threatened against the Company Shareholder which in any manner seek to restrain, enjoin, prohibit, make illegal, or materially delay the Transactions or the performance by the Company Shareholder of its obligations under this Agreement or, when executed, any ancillary agreement to which such Company Shareholder is a party.

### 4.5 Certain Business Practices.

(a) No Company Shareholder or any of its Affiliates acting on its behalf has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the Foreign Corrupt Practices Act of 1977 or similar law of any other jurisdiction, or (iii) made any other unlawful payment. No Company Shareholder or any of its Affiliates acting on its behalf has directly or indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder the Company or assist the Company in connection with any actual or proposed transaction.

(b) The operations of each Company Shareholder and, to the knowledge of any such Company Shareholder, the Company, are and have been conducted at all times in compliance with money laundering statutes in all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Entity in all material respects, and no action involving the Company Shareholder or Company with respect to the any of the foregoing is pending or, to the knowledge of the Company Shareholder, threatened. (c) No Company Shareholder or, if the Company Shareholder is not an individual, any of its directors or officers, or, to the knowledge of the Company Shareholder, any other Person acting on the Company Shareholder's behalf is currently identified on the specially designated nationals or other blocked person list or otherwise currently subject to any Sanctions, or is located, organized, or resident in a country or territory that is, or whose government is, the subject of Sanctions that broadly prohibit dealings with that country or territory, or will, directly, or indirectly, use the proceeds of any offering of securities of the Company or Parent, or lend, contribute, or otherwise make available such proceeds to any other Person in any manner that would result in a violation of any Sanctions or result in the imposition of Sanctions against such Person.

4.6 No Additional Representations and Warranties; No Reliance. Except as provided in this <u>Article IV</u>, neither the Company Shareholders, any of their respective Affiliates, nor any of their respective directors, officers, employees, shareholders, or representatives has made, or is making, any representation or warranty whatsoever to HL or its Affiliates, and no such Party shall be liable in respect of the accuracy or completeness of any information provided to HL or its Affiliates. The Company Shareholders acknowledge and agree (each on its own behalf and on behalf of its Affiliates and its Representatives) that: (i) it has conducted its own independent investigation of the financial condition, results of operations, assets, liabilities, properties and projected operations of HL; (ii) it has been afforded satisfactory access to the books and records, facilities and personnel of HL for purposes of conducting such investigation; and (iii) except for the representations and warranties set forth in <u>Article V</u> (as modified by the HL Schedule), it is not relying on any representations and warranties from any Person in connection with the transactions contemplated hereby.

4.7 <u>Survival of Representations and Warranties</u>. The representations and warranties of the Company Shareholders set forth in this Agreement shall survive the Closing as set forth in <u>Section 11.4</u>.

# ARTICLE V

## **REPRESENTATIONS AND WARRANTIES OF HL**

Subject to the exceptions set forth in Schedule 5 attached hereto (the "HL Schedule"), HL represents and warrants to the Company as follows:

### 5.1 Organization and Qualification.

(a) HL is a corporation duly incorporated, validly existing and in good standing under the laws the British Virgin Islands and has the requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being conducted. HL is in possession of all Approvals necessary to own, lease and operate the properties it purports to own, operate or lease and to carry on its business as it is now being conducted, except where the failure to have such Approvals would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on HL. Complete and correct copies of the Charter Documents of HL, as amended and currently in effect, have been made available to the Company or Company's counsel.

(b) HL is duly qualified or licensed to do business as a foreign corporation and is in good standing in each jurisdiction where the character of the properties owned, leased, or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on HL. Each jurisdiction in which HL is so qualified or licensed is listed in <u>Schedule 5.1(b)</u>.

5.2 <u>Subsidiaries</u>. HL has no, and has never had any, direct or indirect subsidiaries or participations in joint ventures or other entities. HL does not own, directly or indirectly, any equity or voting interest in any Person or has any agreement or commitment to purchase any such interest, and has not agreed and is not obligated to make nor is bound by any written or oral agreement, contract, subcontract, lease, binding understanding, instrument, note, option, warranty, purchase order, license, sublicense, insurance policy, benefit plan, commitment or undertaking of any nature, as of the date hereof or as may hereafter be in effect under which it may become obligated to make, any future investment in or capital contribution to any other Person.

#### 5.3 Capitalization.

(a) As of the date of this Agreement, the authorized capital stock of HL consists of (i) 100,000,000 HL Ordinary Shares, of which 6,560,751 HL Ordinary Shares are issued and outstanding, all of which are validly issued, fully paid and nonassessable and (ii) 1,000,000 HL Preferred Shares, of which no HL Preferred Shares are issued and outstanding.

(b) Except as provided for in this Agreement or as set forth in <u>Schedule 5.3(b)</u>, (i) no HL Ordinary Shares are reserved for issuance upon the exercise of outstanding options to purchase HL Securities granted to employees of HL or other parties ("<u>HL Stock Options</u>") and there are no outstanding HL Stock Options; (ii) 7,875,000 HL Ordinary Shares are reserved for issuance upon the exercise of HL Warrants; (iii) 550,000 HL Ordinary Shares are reserved for issuance upon the exercise of outstanding HL UPOs, including 250,000 HL Ordinary Shares underlying HL Warrants included in the HL UPOs; and (v) no HL Ordinary Shares are reserved for issuance upon the conversion of the HL Preferred Shares or any outstanding convertible notes, debentures, or other securities (collectively, "<u>HL Convertible Securities</u>") and there are no outstanding HL Ordinary Shares subject to issuance, upon issuance on the terms and conditions specified in the instrument pursuant to which they are issuable, will be duly authorized, validly issued, fully paid, and nonassessable. All outstanding HL Ordinary Shares, HL Warrants, and HL Rights have been issued and granted in compliance with (x) all applicable securities laws and (in all material respects) other applicable Legal Requirements, and (y) all requirements set forth in any applicable HL Contracts and Charter Documents.

(c) Except as provided for in this Agreement or as set forth in <u>Schedule 5.3(c)</u> hereto, there are no subscriptions, options, warrants, equity securities, or other ownership interests, calls, rights (including preemptive rights), commitments or agreements of any character to which HL is a party or by which it is bound obligating HL to issue, deliver or sell, or cause to be issued, delivered or sold, or repurchase, redeem or otherwise acquire, or cause the repurchase, redemption or acquisition of, any shares of capital stock or other ownership interests of HL or obligating HL to grant, extend, accelerate the vesting of or enter into any such subscription, option, warrant, equity security, call, right, commitment, or agreement. HL does not have any outstanding bonds, debentures, notes or other obligations the holders of which have or upon the happening of certain events would have the right to vote (or which are convertible into or exercisable or exchangeable for securities having the right to vote) with the HL Shareholders on any matter.

(d) Except as set forth in <u>Schedule 5.3(d)</u> or as contemplated by this Agreement, there are no registration rights, and there is no voting trust, proxy, rights plan, anti-takeover plan, or other agreements or understandings to which HL is a party or by which HL is bound with respect to any HL Securities.

(e) Except as provided for in this Agreement or as set forth in <u>Schedule 5.3(e)</u>, as a result of the consummation of the Transactions, no shares of capital stock, warrants, options, or other securities of HL are issuable and no rights in connection with any shares, warrants, options, or other securities of HL accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility or otherwise).

(f) Except as provided for in this Agreement or as set forth in Schedule 5.3(f), no outstanding HL Securities are unvested or subjected to a repurchase option, risk of forfeiture, or other condition under any applicable agreement with HL.

(g) To HL's knowledge, fewer than 150 natural or legal persons in each member of the European Economic Area and the United Kingdom hold, as of record and/or beneficially, HL Ordinary Shares, HL Warrants, HL Rights, or HL UPOs other than qualified investors (within the meaning of Article 2(e) of Regulation (EU) 2017/1129 of the European Parliament and of the Counsel of 14 June 2017).

### 5.4 Authority Relative to this Agreement.

(a) HL has all necessary power and authority to: (i) execute, deliver and perform this Agreement, and each ancillary document that HL has executed or delivered or is to execute or deliver pursuant to this Agreement, and (ii) carry out HL's obligations hereunder and thereunder and, to consummate the transactions contemplated hereby (including the HL Merger). The execution and delivery of this Agreement by HL and the consummation by HL of the transactions contemplated hereby (including the HL Merger) have been duly and validly authorized by all necessary corporate action on the part of HL (including the approval by its board of directors), and no other corporate proceedings on the part of HL Shareholders required under HL's Charter Documents and the BVI Companies Act. This Agreement has been duly and validly executed and delivered by HL and, assuming the due authorization, execution and delivery thereof by the other Parties, constitutes the legal and binding obligation of HL, enforceable against HL in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity.

(b) The board of directors of HL has, as of the date of this Agreement, unanimously (i) declared the advisability of the HL Merger and approved this Agreement and the transactions contemplated hereby in accordance with the Charter Documents of HL, (ii) determined that the HL Merger is in the best interests of the HL Shareholders, and (iii) determined that the fair market value of the Company is equal to at least 80% of the balance in the Trust Fund (excluding taxes payable).

# 5.5 No Conflict; Required Filings and Consents.

(a) The execution and delivery of this Agreement by HL do not, and the performance of this Agreement by HL shall not: (i) conflict with or violate HL's Charter Documents, (ii) conflict with or violate any applicable Legal Requirements, or (iii) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or materially impair HL's rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a Lien on any of the properties or assets of HL (other than Permitted Liens) pursuant to, any HL Contracts, except, with respect to clauses (ii) and (iii), for any such conflicts, violations, breaches, defaults, impairments, alterations or other occurrences that would not, individually and in the aggregate, have a Material Adverse Effect on HL.

(b) The execution and delivery of this Agreement by HL do not, and the performance of their respective obligations hereunder will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity, except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act, state securities laws, and the rules and regulations thereunder, and appropriate documents with the relevant authorities of other jurisdictions in which HL is qualified to do business, and (ii) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, reasonably be expected to (x) have a Material Adverse Effect on HL, or (y) prevent the consummation of the Transactions or otherwise prevent HL from performing its material obligations under this Agreement on a timely basis.

5.6 <u>Compliance</u>. HL has complied with all, and is not in violation of any, applicable Legal Requirements with respect to the conduct of its business, or the ownership or operation of its business, except for failures to comply or violations which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on HL. The businesses and activities of HL have not been and are not being conducted in violation of any applicable Legal Requirements, except for violations which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on HL. The businesses and activities of HL have not been and are not being conducted in violation of any applicable Legal Requirements, except for violations which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on HL. HL is not in default or violation in any material respect of any term, condition or provision of any applicable Charter Documents. Except as set forth in <u>Schedule 5.6</u>, no written notice of non-compliance with any applicable Legal Requirements has been received by HL (and HL has no knowledge of any such notice delivered to any other Person).

# 5.7 HL SEC Reports and Financial Statements.

(a) HL has timely filed all required registration statements, reports, schedules, forms, statements and other documents filed by HL with the SEC since its formation (collectively, as they have been amended since the time of their filing and including all exhibits thereto, the "HL SEC Reports"). All HL SEC Reports, any correspondence from or to the SEC (other than such correspondence in connection with the initial public offering of HL or such other correspondence with is not required by the rules of the SEC to be publicly disclosed), and all certifications and statements required by: (i) Rule 13a-14 or 15d-14 under the Exchange Act; or (ii) 18 U.S.C. §1350 (Section 906) of the Sarbanes-Oxley Act with respect to any of the foregoing (collectively, the "Certifications"), as of their respective dates, are available on the SEC's Electronic Data-Gathering, Analysis and Retrieval system (EDGAR) in full without redaction (except for redaction permitted by SEC rules and regulations). HL has heretofore furnished to the Company true and correct copies of all amendments and modifications that have not been filed by HL with the SEC to all agreements, documents and other instruments that previously had been filed by HL with the SEC and are currently in effect. The HL SEC Reports were prepared in accordance with the requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act, as the case may be, and the rules and regulations thereunder. None of the HL SEC Reports, as of their respective dates (or, if amended or superseded by a filing prior to the date of this Agreement or the Closing Date, then on the date of such filing), contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. The Certifications are each true and correct. To HL's knowledge, each director and executive officer of HL has filed with the SEC on a timely basis all statements required with respect to HL by Section 16(a) of the Exchange Act and the rules and regulations thereunder. As used in this Section 5.7(a), the term "file" shall be broadly construed to include any manner in which a document or information is furnished, supplied, or otherwise made available to the SEC or Nasdaq.

(b) The audited financial statements of HL ("<u>HL Audited Financial Statements</u>") and unaudited interim financial statements of HL ("<u>HL Unaudited Financial Statements</u>") and unaudited interim financial statements of HL ("<u>HL Unaudited Financial Statements</u>") (including, in each case, the notes and schedules thereto) included in the HL SEC Reports complied as to form in all material respects with the published rules and regulations of the SEC with respect thereto, were prepared in accordance with generally accepted accounting principles in the United States ("<u>U.S. GAAP</u>") applied on a consistent basis in accordance with past practice during the periods involved (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by Form 10-Q of the SEC) and Regulation S-X or Regulation S-K, as applicable, and fairly present (subject, in the case of the unaudited interim financial statements included therein, to normal year-end adjustments, the effect of which are not, individually or in the aggregate, material, and the absence of complete footnotes to the extent permitted by Regulation S-X or Regulation S-K, as applicable) in all material respects the financial position of HL as of the respective dates thereof and the results of their operations and cash flows for the respective periods then ended.

(c) HL has established and maintains disclosure controls and procedures (as defined in Rule 13a-15 or 15d-15(e) under the Exchange Act). Such disclosure controls and procedures are designed to ensure that material information relating to HL is made known to HL's principal executive officer and its principal financial officer, particularly during the periods in which the periodic reports required under the Exchange Act are being prepared. To HL's knowledge, such disclosure controls and procedures are effective in timely alerting HL's principal executive officer and principal financial officer to material information required to be included in HL's periodic reports required under the Exchange Act.

(d) HL has established and maintained a system of internal controls. To HL's knowledge, such internal controls are effective and sufficient to provide reasonable assurance regarding the reliability of HL's financial reporting and the preparation of the HL Financial Statements for external purposes in accordance with U.S. GAAP.

(e) There are no outstanding loans or other extensions of credit made by HL to any executive officer (as defined in Rule 3b-7 under the Exchange Act) or director of HL. HL has not taken any action prohibited by Section 402 of the Sarbanes-Oxley Act.

(f) The books of account, minute books and transfer ledgers and other similar books and records of HL have been maintained in accordance with good business practice, are complete and correct in all material respects and there have been no material transactions that are required to be set forth therein and which have not been so set forth.

(g) Except as otherwise noted in the HL Financial Statements, the accounts and notes receivable of HL reflected in the HL Financial Statements: (i) arose from bona fide sales transactions in the ordinary course of business and are payable on ordinary trade terms, (ii) are legal, valid and binding obligations of the respective debtors enforceable in accordance with their terms, except as such may be limited by bankruptcy, insolvency, reorganization, or other similar laws affecting creditors' rights generally, and by general equitable principles, (iii) are not subject to any valid set-off or counterclaim to which HL has been notified in writing as of the date hereof except to the extent set forth in such balance sheet contained therein, and (iv) are not the subject of any actions or proceedings brought by or on behalf of HL as of the date hereof.

(h) As of the date hereof, there are no outstanding written SEC comments from the SEC with respect to the HL SEC Reports. To the knowledge of HL, none of the HL SEC Reports filed on or prior to the date hereof is subject to ongoing SEC review or investigation as of the date hereof.

5.8 No Undisclosed Liabilities. Except as set forth in Schedule 5.8 hereto, HL has no liabilities (absolute, accrued, contingent or otherwise) of a nature required under U.S. GAAP, applied on a consistent basis in accordance with past practice, to be disclosed on a balance sheet or in the related notes to HL Financial Statements that are, individually or in the aggregate, material to the business, results of operations or financial condition of HL, except: (i) liabilities provided for in or otherwise disclosed in the balance sheet included in the most recent HL Financial Statements or in the notes to the most recent HL Financial Statements, and (ii) such liabilities arising in the ordinary course of HL's business since the date of the most recent HL Financial Statement, none of which, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on HL.

5.9 Absence of Certain Changes or Events. Except as contemplated by this Agreement, since the date of the most recent HL Financial Statement to the date of this Agreement, there has not been: (i) any Material Adverse Effect on HL, (ii) any declaration, setting aside or payment of any dividend on, or other distribution (whether in cash, stock or property) in respect of, any of HL Securities, or any purchase, redemption or other acquisition by HL of any of HL Securities, (iii) any split, combination or reclassification of any HL Securities, (iv) any granting by HL of any increase in compensation or fringe benefits, or any payment by HL of any bonus, or any granting by HL of any increase in severance or termination pay or any entry by HL into any currently effective employment, severance, termination or indemnification agreement or any agreement the benefits of which are contingent or the terms of which are materially altered upon the occurrence of a transaction involving HL of the nature contemplated hereby, (v) entry by HL into any licensing or other agreement with regard to the acquisition or disposition of any material Intellectual Property other than licenses and services agreements in the ordinary course of business consistent with past practice or any amendment or consent with respect to any licensing or services agreement, (vi) any material change by HL in its accounting methods, principles or practices, except as required by concurrent changes in U.S. GAAP, (vii) any change in the auditors of HL, or (viii) any material revaluation by HL of any of its assets, including, without limitation, writing down the value of capitalized inventory or writing off notes or accounts receivable or any sale of assets of HL.

5.10 Litigation. There are no, and have never been any, claims, suits, actions or proceedings pending or to HL's knowledge, threatened against HL, before any court, governmental department, commission, agency, instrumentality or authority, or any arbitrator.

5.11 Employee Benefit Plans. HL does not maintain, and has no liability under, any Plan, and neither the execution and delivery of this Agreement nor the consummation of the Transactions will (i) result in any payment (including severance, unemployment compensation, golden parachute, bonus, or otherwise) becoming due to any shareholder, director, or employee of HL, or (ii) result in the acceleration of the time of payment or vesting of any such benefits.

5.12 Labor Matters. HL is not a party to any collective bargaining agreement or other labor union contract applicable to persons employed by HL and HL does not know of any activities or proceedings of any labor union to organize any such employees. Other than as described in the HL SEC Reports, HL has never had any employees.

5.13 <u>Business Activities</u>. Since its organization, HL has not conducted any business activities other than activities directed toward the accomplishment of a business combination. Except as set forth in the HL Charter Documents, there is no agreement, commitment, exclusive license, judgment, injunction, order, or decree binding upon HL or to which HL is a party which has or would reasonably be expected to have the effect of prohibiting or materially impairing any business practice of HL, any acquisition of property by HL, or the conduct of business by HL.

5.14 <u>Title to Property</u>. HL does not own or lease any real property or personal property. Except as set forth in <u>Schedule 5.14</u>, there are no options or other contracts under which HL has a right or obligation to acquire or lease any interest in real property or personal property.

5.15 Intellectual Property. HL does not own, license, or otherwise have any right, title or interest in any material Intellectual Property.

5.16 Taxes. Except as set forth in Schedule 5.16 hereto:

(a) HL has timely filed all Returns required to be filed by HL with any Tax authority prior to the date hereof. All such Returns are true, correct, and complete. HL has paid all Taxes shown to be due and payable on such Returns.

(b) All Taxes that HL is required by law to withhold or collect have been duly withheld or collected and have been timely paid over to the proper Governmental Entity to the extent due and payable.

(c) HL has not been delinquent in the payment of any Tax, nor is there any Tax deficiency outstanding, proposed or assessed by a taxing authority against HL, nor has HL executed any unexpired waiver of any statute of limitations on or extending the period for the assessment or collection of any Tax. HL has complied in all respects with all applicable Legal Requirements regarding withholding Taxes and has timely paid over in full to the proper taxing authorities all amounts required to be so withheld and paid over for all periods.

(d) No audit or other examination of any Return of HL by any Tax authority is presently in progress, nor has HL been notified of any request for such an audit or other examination.

(e) No adjustment relating to any Returns filed by HL has been proposed, formally or informally, by any Tax authority to HL or any representative thereof.

(f) HL has no liability for any unpaid Taxes which has not been accrued for or reserved on HL's balance sheets included in the HL Audited Financial Statements for the most recent fiscal year ended, whether asserted or unasserted, contingent or otherwise, other than any liability for unpaid Taxes that may have accrued since the date of such HL Financial Statements in connection with the operation of the business of HL in the ordinary course of business or any liability for unpaid Taxes incurred in connection with the Transactions.

(g) HL has not taken, intends to take, or has agreed to take any action or is aware of any fact or circumstance that would prevent or impede, or would reasonably be expected to prevent or impede, the Transactions from qualifying as a reorganization governed by Section 368 of the Code or an integrated exchange within the meaning of Section 351 of the Code.

5.17 Environmental Matters. Except for such matters that, individually or in the aggregate, would not reasonably be expected to result in a material liability of HL: (i) HL has complied in all material respects with applicable Environmental Laws; (ii) neither HL nor, the knowledge of HL, any third party, has caused any properties currently owned, leased or operated by HL to be contaminated with any Hazardous Substances; (iii) the properties formerly owned, leased or operated by HL were not contaminated with Hazardous Substances during the period of ownership, leasing or operation by HL; (iv) as of the date hereof, HL has not received notice that it is potentially liable for any Hazardous Substance disposal or contamination on any third party or public property (whether above, on, or below ground or in the atmosphere or water); (v) as of the date hereof, HL has not received any written notice, demand, letter, claim, or request for information alleging that HL may be in material violation of or have material liability under any Environmental Law; and (vii) HL is not subject to any orders, decrees, injunctions, or other arrangements with any Governmental Entity or subject to any contractual indemnity or other agreement with any third party relating to a material liability under any Environmental Law, including in relation to Hazardous Substances.

5.18 <u>Brokers</u>. Except as set forth in <u>Schedule 5.18</u>, HL has not incurred, and neither will incur, and has not entered into any contract, agreement, understanding, arrangement or commitment pursuant to which Parent or the Surviving BVI Company or any of its direct or indirect subsidiaries could incur, directly or indirectly, any liability for brokerage or finders' fees or agent's commissions or any similar charges in connection with this Agreement or any transaction contemplated hereby.

# 5.19 Agreements, Contracts and Commitments.

(a) Except as set forth in the HL SEC Reports filed prior to the date of this Agreement or as set forth on <u>Schedule 5.19(a)</u>, other than confidentiality and non-disclosure agreements, there are no contracts, agreements, leases, mortgages, indentures, notes, bonds, Liens, license, permit, franchise, purchase orders, sales orders or other understandings, commitments or obligations (including without limitation outstanding offers or proposals) of any kind, whether written or oral, to which HL a party or by or to which any of the properties or assets of HL may be bound, subject or affected, which may not be cancelled without penalty or liability by HL on less than 30 days' or less prior notice ("<u>HL Contracts</u>"). All HL Contracts are listed in <u>Schedule 5.19</u> other than those that are exhibits to the HL SEC Reports.

(b) Except as set forth in the HL SEC Reports filed prior to the date of this Agreement, each HL Contract was entered into at arms' length and in the ordinary course, is in full force and effect, and is valid and binding upon and enforceable against each of the parties thereto, except insofar as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, or similar laws affecting creditors' rights generally or by principles governing the availability of equitable remedies. True, correct, and complete copies of all HL Contracts (or written summaries in the case of oral HL Contracts) have been made available to the Company or Company counsel.

(c) Neither HL nor, to the knowledge of HL, any other party thereto is in breach of or in default under, and no event has occurred which with notice or lapse of time or both would become a breach of or default under, any HL Contract, and no party to any HL Contract has given any written notice of any claim of any such breach, default or event, which, individually or in the aggregate, are reasonably likely to have a Material Adverse Effect on HL. Each HL Contract that has not expired by its terms is in full force and effect, except where such failure to be in full force and effect is not reasonably likely to have a Material Adverse Effect on HL.

5.20 Insurance. Except for directors' and officers' liability insurance, HL does not maintain any Insurance Policy.

5.21 <u>Interested Party Transactions</u>. Except as set forth in the HL SEC Reports, no employee, officer, director, or shareholder of HL or a member of his or her immediate family is (a) indebted to HL nor is HL indebted (or committed to make loans or extend or guarantee credit) to any of them, other than reimbursement for reasonable expenses incurred on behalf of HL or (b) to HL's knowledge, directly or indirectly, interested in any material contract with HL (other than such contracts as relate to the acquisition of such shareholder's ownership of capital stock or other securities of HL).

5.22 <u>HL Listing</u>. The HL Ordinary Share, HL Warrants, HL Rights, and HL's units are listed for trading on the Nasdaq Capital Market ("<u>Nasdaq</u>"). Except as set forth in the HL SEC Reports, there is no, and there has never been any, action or proceeding pending or, to HL's knowledge, threatened against HL by Nasdaq with respect to any intention by such entity to prohibit or terminate the listing of HL Ordinary Shares on Nasdaq. None of HL or any of its Affiliates has taken any action in an attempt to terminate the registration of the HL Ordinary Shares under the Exchange Act.

### 5.23 Trust Fund.

(a) As of the date hereof, HL has no less than \$53.4 million invested in United States Government securities or money market funds meeting the conditions under Rule 2a-7(d) promulgated under the Investment Company Act of 1940, as amended, in a trust account administered by Continental (the "Trust Fund").

(b) <u>Schedule 5.23(b)</u> sets forth, as of the date hereof, HL's good faith estimate of all payments to be made out of the Trust Fund at or prior to the Closing as directed in the trust termination letter, including all amounts payable (i) to Converting Shareholders, (ii) for income Tax or other Tax obligations of HL prior to Closing, (iii) as repayment of HL Borrowings, if any, (iv) to EBC pursuant to the Business Combination Marketing Agreement dated June 27, 2018, and (v) to satisfy all Transaction Expenses of the Parties hereto in connection with each of its operations and efforts to effect the Transactions. Such schedule, together with the updated schedule required to be delivered pursuant to <u>Section 8.13</u> below, is referred to as the "<u>Trust Disbursement Schedule</u>" and the amounts referred to therein or otherwise made in respect thereof at the Closing are hereinafter referred to as the "<u>Trust Disbursements</u>".

(c) HL has performed all material obligations required to be performed by it to date under, and is not in material default, breach or delinquent in performance or any other respect (claimed or actual) in connection with the Trust Fund, and no event has occurred which, with due notice or lapse of time or both, would constitute such a default or breach thereunder. As of the date hereof, there are no claims or proceedings pending with respect to the Trust Fund. As of the Effective Time, the obligations of HL to dissolve or liquidate pursuant to the HL Charter Documents shall terminate, and as of the Effective Time, HL shall have no obligation whatsoever pursuant to the HL Charter Documents to dissolve and liquidate the assets of HL by reason of the consummation of the Transactions. To the knowledge of HL, as of the date hereof, following the Effective Time, no HL Shareholder shall be entitled to receive any amount from the Trust Fund except to the extent such HL Shareholder is a Converting Shareholder.

(d) Assuming the accuracy of the representations and warranties of the Company, the Parent and the Company Shareholders contained herein and the compliance by the Company, the Parent and the Company Shareholders with each of its respective obligations hereunder, HL believes that the conditions to the use of funds in the Trust Fund as set forth herein will be satisfied and funds available in the Trust Fund (less all Trust Disbursements) will be available to HL and the Company to fund on-going operations of the Company from and after the Closing Date.

5.24 <u>Proxy Statement/Prospectus</u>. None of the information relating to HL to be included in the Proxy Statement/Prospectus will, as of the date the Proxy Statement/Prospectus (or any amendment or supplement thereto) is first mailed to the HL Shareholders, at the time of the HL Special Meeting, or at the Effective Time, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

5.25 No Additional Representations and Warranties; No Reliance. Except as provided in this <u>Article V</u> (as modified by the HL Schedule), neither HL, any of its Affiliates, nor any of their respective directors, officers, employees, shareholders, partners, members or representatives has made, or is making, any representation or warranty whatsoever to the Company, any Subsidiary, or their Affiliates, and no such Party shall be liable in respect of the accuracy or completeness of any information provided to the Company, any Subsidiary or their Affiliates. HL acknowledges and agrees (on its own behalf and on behalf of its Affiliates and their respective Representatives) that: (i) it has conducted its own independent investigation of the financial condition, results of operations, assets, liabilities, properties and projected operations of the Company and its Subsidiaries; (ii) it has been afforded satisfactory access to the books and records, facilities and personnel of the Company and its Subsidiaries for purposes of conducting such investigation; and (iii) except for the representations and warranties set forth in <u>Article III</u> (as modified by the Company Schedule), <u>Article IV</u>, and <u>Article VI</u>, it is not relying on any representations and warranties from any Person in connection with the transactions contemplated hereby.

5.26 <u>Survival of Representations and Warranties</u>. The representations and warranties of HL forth in this Agreement shall survive the Closing as set forth in <u>Section 11.4</u>.

### ARTICLE VI

### REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Each of Parent and Merger Sub represents and warrants to HL and the Company (each with respect to itself only) as follows:

#### 6.1 Organization and Qualification.

(a) Each of Parent and Merger Sub is duly incorporated, validly existing and in good standing under the laws of the state or country of its incorporation and has the requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being conducted. Complete and correct copies of the Charter Documents of each of Parent and Merger Sub, as amended and currently in effect, have been made available to HL or HL's counsel.

(b) Each of Parent and Merger Sub is duly qualified or licensed to do business as a foreign corporation and is in good standing in each jurisdiction where the character of the properties owned, leased, or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Parent or Merger Sub.

6.2 <u>Subsidiaries</u>. Parent has no, and has never had any, direct or indirect subsidiaries or participations in joint ventures or other entities other than Merger Sub. Merger Sub does not have, and has never had any, direct or indirect subsidiaries or participations in joint ventures or other entities. Except for Parent's ownership of Merger Sub, neither Parent nor Merger Sub owns, directly or indirectly, any equity or voting interest in any Person or has any agreement or commitment to purchase any such interest, and has not agreed and is not obligated to make nor is bound by any written or oral agreement, contract, subcontract, lease, binding understanding, instrument, note, option, warranty, purchase order, license, sublicense, insurance policy, benefit plan, commitment or undertaking of any nature under which it may become obligated to make, any future investment in or capital contribution to any other Person. Parent and Merger Sub do not have any assets or properties of any kind, do not now conduct and have never conducted any business, and have and will have at the Closing no obligations or liabilities of any nature whatsoever, except for such obligations as are imposed under this Agreement.

# 6.3 Capitalization.

(a) The Parent has no limit in its authorized share capital and it has an issued and outstanding share capital of one ordinary share of EUR1.00. The Memorandum and Articles of Association will be amended prior to the Closing as set forth in <u>Section 8.2</u> hereunder. The authorized shares of stock of Merger Sub is 100 ordinary shares, par value \$0.01 per share, one (1) of which is issued and outstanding. Parent owns all of the outstanding equity securities of Merger Sub, free and clear of all Liens.

(b) Except for this Agreement, the Transactions, and the PIPE Subscription Agreements, as of the date hereof, there are no subscriptions, options, warrants, equity securities, or other ownership interests, calls, rights (including preemptive rights), commitments or agreements of any character to which Parent or Merger Sub is a party or by which it is bound obligating Parent or Merger Sub to issue, deliver or sell, or cause to be issued, delivered or sold, or repurchase, redeem or otherwise acquire, or cause the repurchase, redemption or acquisition of, any shares of capital stock or other ownership interests of Parent or Merger Sub or obligating any of them to grant, extend, accelerate the vesting of or enter into any such subscription, option, warrant, equity security, call, right, commitment, or agreement. Neither Parent nor Merger Sub has any outstanding bonds, debentures, notes or other obligations the holders of which have or upon the happening of certain events would have the right to vote (or which are convertible into or exercisable or exchangeable for securities having the right to vote) with the shareholders or shareholders of Parent or Merger Sub on any matter.

(c) Except contemplated by this Agreement and the PIPE Subscription Agreements, there are no registration rights, and there is no voting trust, proxy, rights plan, anti-takeover plan, or other agreements or understandings to which Parent or Merger Sub is a party or by which Parent or Merger Sub is bound with respect to any of their equity securities.

(d) Except as provided for in this Agreement and the PIPE Subscription Agreements, as a result of the consummation of the Transactions, no shares of capital stock, warrants, options, or other securities of Parent or Merger Sub are issuable and no rights in connection with any shares, warrants, options, or other securities of Parent or Merger Sub accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility or otherwise).

(e) No outstanding equity securities of Parent or Merger Sub are unvested or subjected to a repurchase option, risk of forfeiture, or other condition under any applicable agreement with Parent or Merger Sub.

(f) The Parent Class A Ordinary Shares, Parent Class B Ordinary Shares, HL Parent Warrants and Parent Warrants to be issued by Parent in connection with the Transactions, upon issuance in accordance with the terms of this Agreement, will be duly authorized and validly issued and such Parent Class A Ordinary Shares, Parent Class B Ordinary Shares, HL Parent Warrants and Parent Warrants will be fully paid and nonassessable, free and clear of all Liens.

6.4 <u>Authority Relative to this Agreement</u>. Each of Parent and Merger Sub has all necessary power and authority to: (i) execute, deliver and perform this Agreement, and each ancillary document that Parent and Merger Sub, respectively, has executed or delivered or is to execute or deliver pursuant to this Agreement, and (ii) carry out each of their obligations hereunder and thereunder and, to consummate the transactions contemplated hereby and thereby (including the Transactions). The execution and delivery of this Agreement by Parent and Merger Sub and the consummation by Parent and Merger Sub of the transactions contemplated hereby (including the Transactions) have been or will be duly and validly authorized by all necessary corporate action on the part of Parent and Merger Sub (including the approval by their respective boards of directors), and no other corporate proceedings on the part of Parent or Merger Sub are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been or will be duly and validly executed and delivered by Parent and Merger Sub and, assuming the due authorization, execution and delivery thereof by the Company and HL, constitutes the legal and binding obligation of Parent and Merger Sub, enforceable against Parent and Merger Sub in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity.

# 6.5 No Conflict; Required Filings and Consents.

(a) The execution, delivery, and performance of this Agreement by each of Parent and Merger Sub do not and shall not: (i) conflict with or violate any of their respective Charter Documents, (ii) conflict with or violate any applicable Legal Requirements, or (iii) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or materially impair any of their rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a Lien on any of the properties or assets of Parent or Merger Sub, *except*, with respect to clauses (ii) and (iii), for any such conflicts, violations, breaches, defaults, impairments, alterations or other occurrences that would not, individually and in the aggregate, have a Material Adverse Effect on Parent or Merger Sub.

(b) The execution and delivery of this Agreement by Parent and Merger Sub does not, and the performance of their respective obligations hereunder will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity, except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act, state securities laws, and the rules and regulations thereunder, and appropriate documents with the relevant authorities of other jurisdictions in which Parent is qualified to do business, (ii) the qualification of the Surviving BVI Company, the Company, or its Subsidiaries as a foreign corporation in those jurisdictions in which the business of such entities makes such qualification necessary, and (iii) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, reasonably be expected to (x) have a Material Adverse Effect on Parent or Merger Sub, or (y) prevent the consummation of the Transactions or otherwise prevent Parent or Merger Sub from performing their material obligations under this Agreement on a timely basis.

6.6 <u>Compliance</u>. Each of Parent and Merger Sub has complied with all, and is not in violation of any, applicable Legal Requirements with respect to the conduct of its business, or the ownership or operation of its business, except for failures to comply or violations which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on Parent or Merger Sub. The businesses and activities of Parent and Merger Sub have not been and are not being conducted in violation of any applicable Legal Requirements, except for violations which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on Parent or Merger Sub. Neither Parent nor Merger Sub is in default or violation in any material respect of any term, condition or provision of any applicable Charter Documents. No written notice of non-compliance with any applicable Legal Requirements has been received by Parent or Merger Sub.

6.7 Litigation. There are no, and there never have been any, claims, suits, actions or proceedings pending against Parent or Merger Sub before any court, governmental department, commission, agency, instrumentality, authority, or arbitrator.

6.8 <u>Business Activities</u>. Since its organization, neither Parent nor Merger Sub has conducted any business activities other than activities directed toward the accomplishment of the Transactions. Neither Parent nor Merger Sub has any employees. Neither Parent nor Merger Sub owns or leases any real property, personal property, or Intellectual Property. Parent and Merger Sub were each created for the purpose of facilitating the Transactions and have not conducted any prior business activities, other than any such activities incidental to consummating the Transactions. Except as set forth in the Charter Documents of each of Parent and Merger Sub, there is no agreement, commitment, exclusive license, judgment, injunction, order, or decree binding upon, or to which such Person is a party, which has or would reasonably be expected to have the effect of prohibiting or materially impairing any business practice of Parent or Merger Sub, or any acquisition of property by or the conduct of business by Parent or Merger Sub.

6.9 Brokers. Except as set forth in Schedule 6.9 hereto, neither Parent nor Merger Sub has incurred, and will not incur, and have not entered into any contract, agreement, understanding, arrangement or commitment pursuant to which the Parent, the Surviving BVI Company, the Company, or any of its or their direct or indirect subsidiaries could incur, directly or indirectly, any liability for brokerage or finders' fees or agent's commissions or any similar charges in connection with this Agreement or any transaction contemplated hereby.

6.10 Interested Party Transactions. No officer, director, or shareholder of Parent or Merger Sub or a member of his or her immediate family is (a) indebted to any of Parent or Merger Sub nor is Parent or Merger Sub indebted (or committed to make loans or extend or guarantee credit) to any of them, other than reimbursement for reasonable expenses incurred on behalf of Parent or Merger Sub; or (b) directly or indirectly, interested in any material contract with Parent or Merger Sub (other than such contracts as relate to the acquisition of such shareholder's or shareholder's ownership of capital stock or other securities).

6.11 <u>Board Approval</u>. The board of directors of each of Parent and Merger Sub has, as of the date of this Agreement, unanimously (i) declared the advisability of the Transactions and approved this Agreement in accordance with the Charter Documents of such entity, and (ii) determined that the Transactions are in the best interests of the shareholders of such entity.

6.12 <u>Proxy Statement/Prospectus</u>. None of the information relating to Parent or Merger Sub to be included in the Proxy Statement/Prospectus will, as of the date the Proxy Statement/Prospectus (or any amendment or supplement thereto) is first mailed to the HL Shareholders, at the time of the HL Special Meeting, or at the Effective Time, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

6.13 No Additional Representations and Warranties. Except as provided in this Article VI, neither Parent nor Merger Sub, any of their respective Affiliates, nor any of their respective directors, officers, shareholders, or representatives has made, or is making, any representation or warranty whatsoever to HL or its Affiliates, and no such Party shall be liable in respect of the accuracy or completeness of any information provided to HL or its Affiliates. Each of Parent and Merger Sub acknowledges and agrees (on its own behalf and on behalf of its Affiliates and its Representatives) that: (i) it has conducted its own independent investigation of the financial condition, results of operations, assets, liabilities, properties and projected operations of the Company and HL; (ii) it has been afforded satisfactory access to the books and records, facilities and personnel of each of the Company and HL for purposes of conducting such investigation; and (iii) except for the representations and warranties set forth in <u>Article V</u> (as modified by the HL Schedule), it is not relying on any representations and warranties from any Person in connection with the transactions contemplated hereby.

6.14 <u>Survival of Representations and Warranties</u>. The representations and warranties of Parent and Merger Sub set forth in this Agreement shall survive the Closing as set forth in <u>Section 11.4</u>.

#### ARTICLE VII

# CONDUCT PRIOR TO CLOSING

7.1 Conduct of Business by HL, the Company, Parent, and Merger Sub. During the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms and the Closing (the "Interim Period"), each of HL, the Company's Subsidiaries, Parent and Merger Sub shall, except to the extent that HL (in the case of a request by the Company) or the Company (in the case of a request by HL) shall otherwise consent in writing (which consent shall not be unreasonably withheld, conditioned or delayed) or as set forth in Schedule 7.1 hereto or as contemplated by this Agreement, carry on its business in the usual, regular and ordinary course consistent with past practices, in substantially the same manner as heretofore conducted and in compliance with all applicable Legal Requirements (except as expressly contemplated by Schedule 7.1 hereto) and use its commercially reasonable efforts consistent with past practices and policies to (i) preserve substantially intact its present business organization, (ii) keep available the services of its present key officers and employees, (iii) preserve its relationships with key customers, suppliers, distributors, licensors, licensees, and others with which it has significant business dealings. During the Interim Period, HL shall file all Certifications, registration statements, reports, schedules, forms, statements, and other documents required to be filed or furnished to the SEC ("Additional HL SEC Reports"), which Additional SEC Reports shall be prepared in accordance with the requirements of the Securities Act, the Exchange Act, and the Sarbanes-Oxley Act, as the case may be, and the rules and regulations thereunder, and which will not contain any untrue statement of a material fact or omission of a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. In addition, except as required or permitted by the terms of this Agreement, the PIPE Subscription Agreements, or as set forth in Schedule 7.1 hereto, without the prior written consent of HL (in the case of a request by the Company) or the Company (in the case of a request by HL), which consent shall not be unreasonably withheld, conditioned or delayed, during the Interim Period, HL, the Company, the Company's Subsidiaries, Parent, and Merger Sub shall not do any of the following:

(a) Waive any share repurchase rights, accelerate, amend or (except as specifically provided for herein) change the period of exercisability of options or restricted stock, or reprice options granted under any employee, consultant, director or other stock plans or authorize cash payments in exchange for any options granted under any of such plans;

(b) Grant any material severance or termination pay to (i) any officer or (ii) any employee, except pursuant to applicable Legal Requirements, written agreements outstanding, or Plans or policies existing on the date hereof and as previously or concurrently disclosed or made available to the other Party, or in the case of the Company and its Subsidiaries except in connection with the promotion, hiring or firing of any employee in the ordinary course of business consistent with past practice;

(c) Abandon, allow to lapse, transfer, sell, assign, or license to any Person or otherwise extend, amend or modify any material rights to any Intellectual Property or enter into grants to transfer or license to any Person future patent rights, other than in the ordinary course of business consistent with past practices;

(d) Declare, set aside or pay any dividends on or make any other distributions (whether in cash, stock, equity securities or property) in respect of any capital stock (other than any such dividend or distribution by a Subsidiary of the Company to the Company or another such Subsidiary), or split, combine or reclassify any capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for any capital stock;

(e) Purchase, redeem or otherwise acquire, directly or indirectly, any shares of capital stock or other equity securities or ownership interests, except with respect to Converting Shareholders;

(f) Issue, deliver, sell, authorize, pledge or otherwise encumber, or agree to any of the foregoing with respect to, any shares of capital stock or other equity securities or ownership interests or any securities convertible into or exchangeable for shares of capital stock or other equity securities or ownership interests, or subscriptions, rights, warrants or options to acquire any shares of capital stock or other equity securities or ownership interests or any securities convertible into or exchangeable for shares of capital stock or other equity securities or ownership interests, or enter into other agreements or commitments of any character obligating it to issue any such shares, equity securities or other ownership interests or convertible or exchangeable securities, except as contemplated by Section 8.2 hereunder;

(g) Amend its Charter Documents in any material respect, except as contemplated by Section 8.2 hereunder;

(h) Acquire or agree to acquire by merging or consolidating with, or by purchasing any equity interest in or a material portion of the assets of, or by any other manner, any business or any corporation, partnership, association, or other business organization or division thereof, or otherwise acquire or agree to acquire outside the ordinary course of business any assets which are material, individually or in the aggregate, to the business of such Party or enter into any joint ventures, strategic partnerships or alliances, or other arrangements that provide for exclusivity of territory or otherwise restrict such Party's ability to compete or to offer or sell any products or services to other Persons. For purposes of this paragraph, "material" includes the requirement that, as a result of such transaction, financial statements of the acquired, merged, or consolidated entity be included in the Proxy Statement/Prospectus;

(i) Sell, lease, license, encumber or otherwise dispose of any properties or assets, except (A) sales in the ordinary course of business consistent with past practice, and (B) the sale, lease or disposition of property or assets that are not material, individually or in the aggregate, to the business of such Party (measured with all of its Subsidiaries, taken as a whole);

(j) Except incurrences of indebtedness under the Company's existing credit facilities (and, in the case of the Company and its Subsidiaries, extensions of credit in the ordinary course with employees and among the Company and its Subsidiaries), incur any indebtedness for borrowed money or guarantee any such indebtedness of another Person or Persons (other than Affiliates), issue or sell any debt securities or options, warrants, calls or other rights to acquire any debt securities, enter into any "keep well" or other agreement to maintain any financial statement condition or enter into any arrangement having the economic effect of any of the foregoing;

(k) Except as contemplated by <u>Section 8.15</u> hereunder, or as otherwise required by applicable Legal Requirements or pursuant to an existing Plan, policy or Company Contract, (i) adopt or materially amend any Plan (including any Plan that provides for severance), or enter into any employment contract or collective bargaining agreement (other than in the ordinary course of business consistent with past practice), (ii) pay any special bonus or special remuneration to any director or employee, except in the ordinary course of business consistent with past practices, or (iii) materially increase the salaries or wage rates or fringe benefits (including rights to severance or indemnification) of its directors, officers, employees or consultants, except in the ordinary course of business consistent with past practices;

(1) (i) Pay, discharge, settle or satisfy any material claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), or litigation (whether or not commenced prior to the date of this Agreement) other than the payment, discharge, settlement or satisfaction of any claims, liabilities or obligations in the ordinary course of business consistent with past practices or in accordance with their terms, or recognized or disclosed in the most recent Company Financial Statements or in the HL SEC Reports, as applicable, or incurred since the date of such financial statements, (ii) settle any material litigation where the consideration given by the Party is other than monetary or to which an officer, director or employee of such Person is a party in his or her capacity as such, or (iii) waive the benefits of, agree to modify in any material manner, terminate, release any Person from or knowingly fail to enforce any material confidentiality or similar agreement to which the Company or any of its Subsidiaries is a beneficiary (other than with customers and other counterparties in the ordinary course of business consistent with past practices) or to which HL is a beneficiary, as applicable;

(m) Except in the ordinary course of business consistent with past practices, modify in any material respect or terminate (other than in accordance with its terms) any Company Material Contract or HL Contract, as applicable, or waive, delay the exercise of, release or assign any material rights or claims thereunder;

(n) Except as required by law, U.S. GAAP, or IFRS, revalue any of its assets in any material manner or make any material change in accounting methods, principles or practices;

(o) Except in the ordinary course of business consistent with past practices, incur or enter into any agreement, contract or commitment requiring such Party to pay in excess of \$1,000,000 in any 12-month period;

(p) Make or rescind any Tax elections that, individually or in the aggregate, would be reasonably likely to adversely affect the Tax liability or Tax attributes of such Party, settle or compromise any income Tax liability outside the ordinary course of business or, except as required by applicable Legal Requirements, change any method of accounting for Tax purposes or prepare or file any Return in a manner inconsistent with past practice;

(q) Form or establish any subsidiary except in the ordinary course of business consistent with prior practice or as contemplated by this

# Agreement;

(r) Permit any Person to exercise any of its discretionary rights under any Plan to provide for the automatic acceleration of any outstanding options, the termination of any outstanding repurchase rights or the termination of any cancellation rights issued pursuant to such Plans;

(s) Make capital expenditures in excess of \$500,000;

(t) Enter into any material transaction with or distribute or advance any assets or property to any of its officers, directors, partners, shareholders, managers, members or other Affiliates other than (i) the payment of salary and benefits and the advancement of expenses in the ordinary course of business consistent with prior practice or (ii) such distributions or advancements by a Subsidiary of the Company to the Company or another such Subsidiary; or

(u) Agree in writing or otherwise agree or commit to take any of the actions described in Section 7.1(a) through (t) above.

Notwithstanding the foregoing provisions of <u>Sections 7.1(d)</u>, (f), and (g) as set forth above, HL hereby consents to the Company's adoption of its Amended and Restated Articles of Association, the creation of the Company Class A Shares thereunder, and the issuance of an aggregate of 100 Company Class A Shares to certain Company Shareholders, and waives any noncompliance with such Sections.

Nothing in this Section 7.1 shall prohibit a Party from taking any action or omitting to take any action as required by this Agreement.

# 7.2 Confidentiality; Access to Information.

(a) <u>Confidentiality</u>. The Parties agree that they shall be bound by that certain Non-Disclosure Agreement, dated as of May 14, 2020 (the "<u>Confidentiality Agreement</u>"), by and between the Company and HL, with respect to all nonpublic information exchanged in connection with this Agreement and the negotiations related thereto. The terms of the Confidentiality Agreement are hereby incorporated herein by reference and shall continue in full force and effect until the Closing, at which time the Confidentiality Agreement shall terminate. If this Agreement is, for any reason, terminated prior to the Closing, the Confidentiality Agreement shall continue in full force and effect, subject to <u>Section 10.2(b)</u> hereof.

# (b) Access to Information.

(i) Subject to confidentiality obligations and similar restrictions that may be applicable to information furnished to the Company or its Subsidiaries by third parties that may be in the Company's or its Subsidiaries' possession from time to time, and except for any information which in the opinion of legal counsel of the Company would result in the loss of attorney-client privilege or other privilege from disclosure, the Company will afford HL and its financial advisors, accountants, counsel and other representatives reasonable access during normal business hours, upon reasonable notice, in such manner as to not interfere with the normal operation of the Company and its Subsidiaries, to the properties, books, records and management personnel of the Company during the period prior to the Closing to obtain all information concerning the business, including the status of business development efforts, properties, results of operations and personnel of the Company, as HL may reasonably request; provided, that such access shall not include any invasive or intrusive investigations or other testing, sampling or analysis of any properties, facilities or equipment of the Company or its Subsidiaries without the prior written consent of the Company. The Parties shall use commercially reasonable efforts to make alternative arrangements for such disclosure where the restrictions in the preceding sentence apply. No information or knowledge obtained by HL in any investigation pursuant to this <u>Section 7.2(b)(i)</u> will affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the Parties to consummate the Transactions.

(ii) Subject to confidentiality obligations and similar restrictions that may be applicable to information furnished to HL by third parties that may be in HL's possession from time to time, and except for any information which in the opinion of legal counsel of HL would result in the loss of attorney-client privilege or other privilege from disclosure, HL will afford the Company and its financial advisors, underwriters, accountants, counsel and other representatives reasonable access during normal business hours, upon reasonable notice, in such manner as to not interfere with the normal operation of HL, to the properties, books, records and personnel of HL during the period prior to the Closing to obtain all information concerning the business, including properties, results of operations and personnel of HL, as the Company may reasonably request. The Parties shall use commercially reasonable efforts to make alternative arrangements for such disclosure where the restrictions in the preceding sentence apply. No information or knowledge obtained by the Company in any investigation pursuant to this <u>Section 7.2(b)(ii)</u> will affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the Parties to consummate the Transactions.

7.3 Exclusivity. During the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Closing. (a) the Company Company Shareholders. Parent, and Merger Sub will not, and each will cause its respective Affiliates. employees, agents, officers, directors and representatives not to, directly or indirectly, solicit or enter into discussions or transactions with, or encourage, or provide any information to, any corporation, partnership or other entity or group (other than HL and its designees) concerning any merger, sale of ownership interests in the Company (other than shares of (i) any indebtedness for borrowed money to the extent not convertible into equity securities of the Company as permitted by this Agreement (including indebtedness evidenced by notes, bonds, debentures, credit agreement(s) or similar instruments or seller carryback financing in acquisitions) and (ii) any instruments convertible into equity securities of the Company upon the consummation of the Transactions which are permitted by this Agreement) and/or a material portion of the assets of the Company (other than immaterial assets or assets sold in the ordinary course of business), or a recapitalization or similar transaction involving the Company and (b) HL will not, and will cause its Affiliates, employees, agents, officers, directors and representatives not to, directly or indirectly, solicit or enter into discussions or transactions with, or encourage, or provide any information to, any corporation, partnership or other entity or group (other than the Company and its designees) concerning any merger, purchase of ownership interests and/or assets, recapitalization or similar business combination transaction. In addition, (x) the Company, Company Shareholders, Parent, and Merger Sub will, and will cause its respective Affiliates, employees, agents, officers, directors and representatives to, immediately cease any and all existing discussions or negotiations with any Person conducted heretofore with respect to any alternative merger, sale of ownership interests in and/or a material portion of the assets of the Company (other than immaterial assets or assets sold in the ordinary course of business) or recapitalization or similar transaction involving the Company and (y) HL will, and will cause its Affiliates, employees, agents, officers, directors and representatives to, immediately cease any and all existing discussions or negotiations with any Person conducted heretofore with respect to any alternative merger, purchase of ownership interests and/or assets, recapitalization or similar business combination transaction. The Company will promptly notify HL if it receives, or if any of its or its Affiliates' employees, agents, officers, directors or representatives receives any proposal, offer or submission with respect to a competing transaction after the date of this Agreement. Notwithstanding the foregoing, the Company may respond to any such proposal, offer or submission by indicating only that the Company is subject to an exclusivity agreement and is unable to provide any information related to the Company and its Subsidiaries or entertain any proposals or offers or engage in any negotiations or discussions concerning a competing transaction for as long as that exclusivity agreement remains in effect. The Parties agree that the rights and remedies for noncompliance with this Section 7.3 include specific performance, it being acknowledged and agreed that any breach or threatened breach will cause irreparable injury to the non-breaching Party and that money damages would not provide an adequate remedy for such injury.

### 7.4 Certain Financial Information.

(a) Within fifteen (15) Business Days after the end of each month between the date hereof and the earlier of the Closing Date and the date on which this Agreement is terminated, the Company shall deliver to HL unaudited consolidated financial statements for such month, including a balance sheet, statement of operations, statement of cash flows, and statement of shareholders' equity, that are certified as correct and complete by the Company's Chief Executive Officer and Chief Financial Officer, prepared in accordance with IFRS applied on a consistent basis to prior periods (except as may be indicated in the notes thereto), and fairly presenting in all material respects the financial position of the Company as of the date thereof and the results of operations and cash flows for the period indicated, except that such financial statements need not contain notes and may be subject to normal adjustments that are not expected to be material to the Company.

(b) During the period from the date of this Agreement and continuing until the earlier of the Closing Date and the date on which this Agreement is terminated, the Company shall, and shall cause its auditors to (i) continue to provide HL and its Representatives reasonable access to all of the financial information used in preparation of Company Financial Statements and the financial information furnished pursuant to <u>Section 7.4(a)</u>, and (ii) reasonably cooperate with any reviews performed by HL or its Representatives of any such Company Financial Statements or financial information.

7.5 <u>Company Audited Financial Statements</u>. Reasonably promptly after the date of this Agreement, and only to the extent required to be included in the Proxy Statement, the Company shall deliver any consolidated audited financial statements of the Company to HL for the same time periods as the Company Annual Financial Statements, together with an opinion of the Company's auditing firm and all notes thereto ("<u>Company Audited Financial Statements</u>"). The Company Audited Financial Statements shall comply as to form in all material respects, and be prepared in accordance, with IFRS (as modified by the rules and regulations of the SEC) applied on a consistent basis throughout the periods involved and shall fairly present in all material respects the consolidated financial position of the Company at the date thereof and the results of their operations and cash flows for the period therein indicated.

7.6 Commercially Reasonable Efforts. Upon the terms and subject to the conditions set forth in this Agreement, each of the Parties agrees to use its commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things reasonably necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Transactions, including using commercially reasonable efforts to accomplish the following: (i) the taking of such reasonable acts necessary to cause the conditions precedent set forth in <u>Article IX</u> to be satisfied, (ii) the obtaining of such reasonably necessary actions, waivers, consents, approvals, orders and authorizations from Governmental Entities and the making of such reasonably necessary registrations, declarations and filings (including registrations, declarations and filings with Governmental Entities, if any) and the taking of such reasonable steps as may be reasonably necessary to avoid any suit, claim, action, investigation or proceeding by any Governmental Entity, (iii) the obtaining of such material consents, approvals or waivers from third parties required as a result of the Transactions, including the consents referred to in <u>Schedules 3.5</u> and <u>5.5</u>, (iv) the defending of any suits, claims, actions, investigations or proceedings, whether judicial or administrative, challenging this Agreement or the consummation of the Transactions, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Entity vacated or reversed, and (v) the execution or delivery of any additional instruments reasonably necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement. Notwithstanding anything herein to the contrary, nothing in this Agreement shall be deemed to require HL or the Company to agree to any divestiture by itself or any of its Affiliates of shares of cap

# ARTICLE VIII

### ADDITIONAL AGREEMENTS

# 8.1 Form F-4; Special Meeting.

(a) As soon as is reasonably practicable after receipt by Parent from HL and the Company of all financial and other information relating to HL and the Company as is necessary for its preparation, Parent shall prepare and file with the SEC and with all other applicable regulatory bodies, the Form F-4 with respect to the Parent Class A Ordinary Shares and HL Parent Warrants to be issued in the Merger, which Form F-4 will contain the Proxy Statement/Prospectus to be used for the purpose of soliciting proxies from the HL Shareholders to vote in favor of (i) the adoption of this Agreement and the approval of the Merger, including the issuance of the HL Merger Consideration (the "Merger Proposal"), (ii) the election to the board of directors of Parent of the individuals identified on Schedule 8.2 for the class of director set forth opposite the name of such individual (the "Director Proposal"); (iii) the approval of certain material differences between HL's Charter Documents and Parent's Charter Documents, to be effective from and after the Closing (the "Charter Amendments Proposals"), and (iv) approval of any other proposals reasonably agreed by HL and the Company to be necessary or appropriate in connection with the transactions contemplated hereby (together with the Merger Proposal, Director Proposal, and Charter Amendments Proposals, the "HL Shareholder Matters") at a meeting of HL Shareholders to be called and held for such purpose (the "HL Special Meeting"). Without the prior written consent of the Company, the HL Shareholder Matters shall be the only matters (other than procedural matters) which HL shall propose to be acted on by the HL Shareholders at the HL Special Meeting. The Company and HL shall each furnish to Parent all information as is necessary in connection with the preparation of the Proxy Statement/Prospectus. The Company and HL, and each of their counsel shall be given a reasonable opportunity to review, comment on and approve in writing the preliminary Proxy Statement/Prospectus prior to its filing with the SEC and any other amendments or documents filed with the SEC and Parent shall not file any documents with the SEC without the prior written consent of the Company and HL, such consent not to be unreasonably withheld, conditioned or delayed. Parent shall consider the comments of the Company and HL in good faith. Parent, with the assistance and written approval of the Company and HL, shall promptly respond to any SEC comments on the Proxy Statement/Prospectus and shall otherwise use reasonable best efforts to cause the Proxy Statement/Prospectus to be approved by the SEC as promptly as practicable. Parent shall also take any and all actions required to satisfy the requirements of the Securities Act and the Exchange Act. Parent will advise the Company and HL promptly after it receives notice thereof, of: (A) the time when the preliminary Proxy Statement/Prospectus has been filed; (B) in the event the preliminary Proxy Statement/Prospectus is not reviewed by the SEC, the expiration of the waiting period in Rule 14a-6(a) under the Exchange Act; (C) in the event the preliminary Proxy Statement/Prospectus is reviewed by the SEC, receipt of oral or written notification of the completion of the review by the SEC; (D) the filing of any supplement or amendment to the Proxy Statement/Prospectus; (E) any request by the SEC for amendment of the Proxy Statement/Prospectus; (F) any comments from the SEC relating to the Proxy Statement/Prospectus and responses thereto; and (G) requests by the SEC for additional information.

(b) As soon as practicable following the SEC declaring the Form F-4 effective (the "<u>SEC Approval Date</u>"), HL shall (i) distribute the Proxy Statement/Prospectus to the HL Shareholders, (ii) having, prior to the SEC Approval Date, established the record date therefor, duly call, give notice of, convene and hold the HL Special Meeting in accordance with the BVI Companies Act and subject to the other provisions of this Agreement, and (iii) subject to the other provisions of this Agreement, solicit proxies from such holders to vote in favor of the HL Shareholder Matters. Notwithstanding the foregoing provisions of this <u>Section 8.1(b)</u>, if on a date for which the HL Special Meeting is scheduled, HL reasonably determines that the HL Merger cannot be consummated for any reason, HL shall have the right to make one or more successive postponements or adjournments of the HL Special Meeting, provided that HL continues to satisfy its obligations under <u>Section 8.1(d)</u> below.

(c) Parent, HL, and the Company shall each comply with all applicable provisions of and rules under the Securities Act, Exchange Act, all applicable provisions of the Corporate Law, as applicable, in the preparation, filing and distribution of the Form F-4 and the Proxy Statement/Prospectus, the solicitation of proxies thereunder, and the calling and holding of the HL Special Meeting. Without limiting the foregoing, Parent, HL, and the Company shall each ensure that each of the Proxy Statement/Prospectus does not, as of the date on which it is first distributed to HL Shareholders and as of the date of the HL Special Meeting, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements made in light of the circumstances under which they were made, not misleading (provided that no Party shall be responsible for the accuracy or completeness of any information relating to another Party or any other information furnished by another Party for inclusion in the Proxy Statement/Prospectus).

(d) HL, acting through its board of directors, shall include in the Proxy Statement/Prospectus the recommendation of its board of directors that the HL Shareholders vote in favor of the HL Shareholder Matters, and shall otherwise use reasonable best efforts to obtain approval of the HL Shareholder Matters. Subject to the fiduciary obligations of HL's board of directors, neither HL's board of directors nor any committee or agent or representative thereof shall withdraw (or modify in a manner adverse to the Company), or propose to withdraw (or modify in a manner adverse to the Company) the HL board of director's recommendation that the HL Shareholders vote in favor of the adoption of the HL Shareholder Matters.

8.2 <u>Conversion of Parent to Public Limited Company</u>; <u>Directors and Officers of Parent After Transactions</u>. Prior to the Closing, the Parent shall revise its Memorandum and Articles of Association to (i) change its name to Fusion Fuel PLC, or such other name as is acceptable to the Company Shareholders, (ii) convert from a private limited company to a public limited company, (iii) incorporate the terms set forth on <u>Schedule 8.2</u>, and (iv) establish a board of directors consistent with the terms set forth on <u>Schedule 8.2</u>; and HL hereby agrees to such revisions and will cooperate with Parent and the Company as reasonably requested to implement such revisions. Except as otherwise agreed in writing by the Company and HL prior to the Closing, the Parties shall take all necessary action so that all of the members of the board of directors of HL and all officers of HL resign effective as of the Closing unless such member or officer is included on <u>Schedule 8.2</u>.

### 8.3 Intentionally Omitted.

### 8.4 Public Announcements.

(a) As promptly as practicable after execution of this Agreement (but in any event within four (4) Business Days thereafter), HL will prepare and file with the SEC a Current Report on Form 8-K pursuant to the Exchange Act to report the execution of this Agreement ("Signing Form 8-K"), the form and substance of which shall be approved in writing in advance by the Company.

(b) Promptly after the execution of this Agreement, HL and the Company shall also issue a joint press release announcing the execution of this Agreement (the "Signing Press Release").

(c) At least five (5) days prior to Closing, Parent shall prepare a draft Shell Company Report on Form 20-F announcing the Closing, together with, or incorporating by reference, the financial statements prepared by the Company and its accountant, and such other information that may be required to be disclosed with respect to the Transactions in any report or form to be filed with the SEC ("<u>Closing Form 20-F</u>"), which shall be in a form reasonably acceptable to HL and the Company. Prior to Closing, HL and the Company shall prepare a press release announcing the consummation of the Transactions hereunder ("<u>Closing Press Release</u>"), which shall be in a form reasonably acceptable to HL and the Company. Concurrently with the Closing, HL shall issue the Closing Press Release. Concurrently with the Closing, or as soon as practicable thereafter (but in any event within four (4) Business Days thereafter), Parent shall file the Closing Form 20-F with the SEC.

### 8.5 Required Information.

(a) In connection with the preparation of the Signing Form 8-K, the Signing Press Release, the Form F-4, the Proxy Statement/Prospectus, the Closing Form 20-F, the Closing Press Release or any other statement, filing, notice, or application made by or on behalf of HL, Parent or the Company to any Governmental Entity or other third party in connection with Transactions (each, a "<u>Reviewable Document</u>"), and for such other reasonable purposes, each of HL, the Company and Parent shall, upon request by the other, use commercially reasonable efforts (subject to applicable Legal Requirements and contractual restrictions) to promptly furnish the other with all information concerning themselves, their Subsidiaries, and each of their and their Subsidiaries' respective directors, officers, and shareholders (including the directors of Parent to be elected effective as of the Closing pursuant to <u>Section 8.2</u> hereof) and such other matters as may be reasonably necessary or advisable in connection with the Transactions and the preparation of such Reviewable Document.

(b) At a reasonable time prior to the filing, issuance, or other submission or public disclosure of a Reviewable Document by HL, Parent or the Company, the other Parties shall each be given a reasonable opportunity to review and comment upon such Reviewable Document and give its consent to the form thereof, such consent not to be unreasonably withheld, and each Party shall accept and incorporate all reasonable comments from the other Party to any such Reviewable Document prior to filing, issuance, submission or disclosure thereof. Furthermore, HL, Parent and the Company shall cooperate and mutually agree upon (such agreement not to be unreasonably withheld or delayed) any response to any SEC comments on any Reviewable Document and shall otherwise use commercially reasonable efforts to cause the Form F-4 to be declared effective by the SEC as promptly as practicable, and keep the Form F-4 effective as long as is necessary to consummate the Transactions.

(c) Any language included in a Reviewable Document that reflects the comments of the reviewing Party, as well as any text as to which the reviewing Party has not commented upon after being given a reasonable opportunity to comment, shall be deemed to have been approved by the reviewing Party and may henceforth be used by other Party in other Reviewable Documents and in other documents distributed by the other Party in connection with the Transactions without further review or consent of the reviewing Party.

(d) Prior to the Closing Date, the Company and HL shall notify each other as promptly as reasonably practicable (i) upon obtaining knowledge of any event or circumstance which should be described in an amendment of, or supplement to, a Reviewable Document that has been filed with or submitted to any Governmental Entity, and (ii) after the receipt by it of any written or oral comments of any Governmental Entity on, or of any written or oral request by any Governmental Entity for amendments or supplements to, any such Reviewable Document, and shall promptly supply the other with copies of all correspondence between it or any of its representatives and such Governmental Entity with respect to any of the foregoing filings or submissions. HL and the Company shall use their respective commercially reasonable best efforts, after consultation with each other, to resolve all such requests or comments with respect to any Reviewable Document as promptly as reasonably practicable after receipt of any comments of any Governmental Entity. All correspondence and communications to any Governmental Entity made by HL or the Company with respect to the Transactions or any agreement ancillary hereto shall, to extent permitted by applicable Legal Requirements, be considered to be Reviewable Documents subject to the provisions of this <u>Section 8.5</u>.

8.6 <u>No Securities Transactions</u>. Neither the Company, Company Shareholders, Parent, Merger Sub nor HL, nor any of its respective controlled Affiliates, directly or indirectly, shall engage in any purchases or sales of the securities of HL, the Company, or Parent, respectively, prior to the Effective Time without the consent of the other Party. Each of the Company and HL shall use its commercially reasonable efforts to require each of its Affiliates that it controls to comply with the foregoing requirement.

8.7 No Claim Against Trust Fund. Notwithstanding anything else in this Agreement, the Company acknowledges that it has read HL's final prospectus dated June 27, 2018 ("Final Prospectus") and understands that HL has established the Trust Fund for the benefit of HL's public shareholders and that HL may disburse monies from the Trust Fund only (a) to HL's public shareholders in the event they elect to convert their shares into cash in accordance with HL's Charter Documents and/or the liquidation of HL or (b) to HL after, or concurrently with, the consummation of a business combination. The Company further acknowledges that, if the Transactions, or, upon termination of this Agreement, another business combination, are not consummated by July 2, 2020, or such later date as shall be set forth in an amendment to HL's Amended and Restated Memorandum and Articles of Association for the purpose of extending the date by which HL must complete a business combination. HL will be obligated to return to its shareholders the amounts being held in the Trust Fund. Accordingly, the Company, on behalf of itself and its Affiliates, hereby waives all rights, title, interest or claim of any kind against HL to collect from the Trust Fund any monies that may be owed to them by HL for any reason whatsoever, including but not limited to a breach of this Agreement by HL or any negotiations, agreements or understandings with HL (whether in the past, present or future), and will not seek recourse against the Trust Fund at any time for any reason whatsoever; provided that: (a) nothing herein shall serve to limit or prohibit the Company's right to pursue a claim against HL pursuant to this Agreement for legal relief against monies or other assets of HL held outside the Trust Fund or for specific performance or other equitable relief in connection herewith and (b) nothing herein shall serve to limit or prohibit any claims that the Company may have in the future pursuant to this Agreement against HL's assets or funds that are not held in the Trust Fund. Furthermore, HL shall not execute any definitive agreement related to such business combination that (i) attempts to prevent HL from so filing or pursuing any such claim, or (ii) permits the Person that survives such combination not to assume HL's obligation for damages in connection with this Agreement and the Transactions. This paragraph will survive the termination of this Agreement for any reason and will not expire and will not be altered in any way without the express written consent of HL.

8.8 <u>Disclosure of Certain Matters</u>. Each of HL and the Company will provide the others with prompt written notice of any event, development or condition of which it obtains knowledge that (a) would cause, if existing or known on the date hereof, such Party's representations and warranties to be or become untrue or misleading or which would prevent it from consummating the Transactions, (b) had it existed or been known on the date hereof would have been required to be discussed under this Agreement, (c) gives such Party any reasonable basis to believe that any of the conditions to the obligations of the other Party set forth in <u>Article IX</u> will not be satisfied, (d) is of a nature that is or may be materially adverse to the operations, prospects, or condition (financial or otherwise) of Parent, or (e) would require any amendment or supplement to the Form F-4 or the Proxy Statement/Prospectus. The Parties shall have the obligation to supplement or amend the Company Schedule and HL Schedule being delivered concurrently with the execution of this Agreement with respect to any matter hereafter arising or discovered which, if existing or known at the date of this Agreement, would have been required to be set forth or described in such schedules. The obligations of the Parties to amend or supplement the Company Schedule and HL Schedule, respectively, shall terminate on the Closing Date.

8.9 Securities Listing. Prior to the Effective Time, HL and Parent shall prepare and submit to Nasdaq, in accordance with all applicable laws and the policies of Nasdaq, an application for the listing on Nasdaq of the Parent Class A Ordinary Shares and HL Parent Warrants, and use commercially reasonable efforts to cause the Parent Class A Ordinary Shares and HL Parent Warrants to be approved for listing on Nasdaq prior to the Effective Time, subject to official notice of issuance. The Company shall cooperate with HL and Parent as reasonably requested by HL to cause the listing application to be approved by Nasdaq and shall promptly furnish to HL all information concerning the Company that may be required or reasonably requested in connection therewith.

### 8.10 Charter Protections; Directors' and Officers' Liability Insurance.

(a) All rights to indemnification for acts or omissions occurring through the Closing Date now existing in favor of the current directors and officers of the Company under applicable Legal Requirement or as provided in the Charter Documents of the Company or in any indemnification agreements shall survive and shall continue in full force and effect in accordance with their terms.

(b) Without limiting any additional rights that any Person may have under any other agreement, for a period of six (6) years after the Closing Date, each of Parent, the Surviving BVI Company, and the Company shall, jointly and severally, indemnify and hold harmless each present (as of immediately prior to the Closing Date) and former officer, director, employee, fiduciary or agent of HL, the Company, the Parent or any Subsidiary (each, a "D&O Indemnified Person") against all claims, losses, liabilities, damages, judgments, inquiries, fines and reasonable fees, costs and expenses, including attorneys' fees and disbursements, incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to (i) the fact that the D&O Indemnified Person is or was an officer, director, employee, manager, managing member, partner (general or limited), fiduciary or agent of HL, the Company or a Subsidiary or (ii) matters existing or occurring at or prior to the Closing Date (including this Agreement and the transactions and actions contemplated hereby), whether asserted or claimed on or after the Closing Date, to the fullest extent permitted under applicable law. In the event of any such claim, action, suit, proceeding or investigation, (x) each D&O Indemnified Person will be entitled to advancement of expenses incurred in the defense of any claim, action, suit, proceeding or investigation from Parent within ten (10) business days of receipt by Parent from the D&O Indemnified Person of a request therefor; provided that any Person to whom expenses are advanced provides an undertaking, if and only to the extent required by applicable law and by the Charter Documents of HL, the Company, or the Subsidiary, as applicable, (y) neither Parent, the Surviving BVI Company, the Company, nor any of its or their Affiliates shall settle, compromise or consent to the entry of any judgment in any proceeding or threatened action, suit, proceeding, investigation or claim in which indemnification could be sought by such D&O Indemnified Person hereunder, unless such settlement, compromise or consent includes an unconditional release of such D&O Indemnified Person from all liability arising out of such action, suit, proceeding, investigation or claim (including all attorney's fees and expenses) or such D&O Indemnified Person otherwise consents and (z) Parent, the Surviving BVI Company, the Company, and its and their Affiliates shall cooperate in the defense of any such matter.

(c) For a period of six (6) years after the Closing Date, Parent shall not and shall not permit the Surviving BVI Company, the Company or any Subsidiary to amend, repeal or otherwise modify any provision in its respective Charter Documents relating to the exculpation or indemnification (including fee advancement) of any officers or directors (unless required by law), it being the intent of the parties that the D&O Indemnified Persons shall continue to be entitled to such exculpation and indemnification (including fee advancement) to the full extent of the law. Parent shall, and shall cause the Surviving BVI Company, the Company, and each Subsidiary to, honor and perform under all indemnification obligations owed to any of the D&O Indemnified Persons.

(d) Upon the Closing, Parent shall purchase a prepaid insurance policy (i.e., "tail coverage") which policy provides liability insurance coverage for the D&O Indemnified Persons on no less favorable terms (including in amount and scope) as the policy or policies maintained by HL or the Company immediately prior to the Closing for the benefit of such individuals for an aggregate period of not less than six (6) years with respect to claims arising from acts, events or omissions that occurred at or prior to the Closing, including with respect to the Transactions. The cost of such policy shall be borne by Parent. Such policy shall be from an insurance carrier with the same or better credit rating as the current insurance carrier(s) of HL and the Company with respect to directors' and officers' liability insurance.

(e) Upon the Closing, Parent shall purchase a directors' and officers' insurance policy in an amount to be mutually agreed by the Company and HL which policy provides liability insurance coverage with respect to claims arising from acts, events or omissions that occur after the Closing. The cost of such policy shall be borne by Parent. Such policy shall be from an insurance carrier with the same or better credit rating as the current insurance carrier(s) of HL and the Company with respect to directors' and officers' liability insurance.

(f) If Parent or any of its successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving entity of such consolidation or merger, or (ii) transfers or conveys all or substantially all of its properties and assets to any Person, then, in each such case, to the extent necessary, proper provision will be made so that the successors and assigns of Parent assume the obligations set forth in this <u>Section 8.10</u>.

(g) The provisions of this <u>Section 8.10</u> are intended to be for the benefit of, and will be enforceable by, each of the D&O Indemnified Persons and may not be changed after Closing without the consent of the HL Representative and a majority of those D&O Indemnified Persons serving on Parent's board of directors after the Closing Date.

8.11 <u>Insider Loans</u>. The Company shall cause each Insider of the Company or its Subsidiaries to, at or prior to Closing (i) repay to the Company any loan by the Company to such Insider and any other amount owed by such Insider to the Company; and (ii) cause any guaranty or similar arrangement pursuant to which the Company has guaranteed the payment or performance of any obligations of such Insider to a third party to be terminated.

8.12 <u>HL Borrowings</u>. Through the Closing, HL shall be allowed to borrow funds from its directors, officers, shareholders, and each of their Affiliates to meet its reasonable capital requirements, with any such loans to be made only as reasonably required by the operation of HL in due course on a non-interest bearing basis and be repayable in cash at Closing (or convertible into securities of HL as described in the Final Prospectus) (such aggregate amount of outstanding borrowings, the "<u>HL Borrowings</u>").

8.13 Trust Fund Disbursement. Upon satisfaction or waiver of the conditions set forth in <u>Article IX</u> and provision of notice to Continental (which notice HL shall provide in accordance with the terms of the Trust Agreement), in accordance with and pursuant to the Trust Agreement, at the Closing, HL shall cause the documents, opinions, and notices required to be delivered to Continental pursuant to the Trust Agreement to be so delivered, including providing Continental with the trust termination letter. The trust termination letter shall include an updated Trust Disbursement Schedule to reflect the amount of all payments to be made out of the Trust Fund (or to Parent for further disbursement thereof) at or prior to the Closing, accrued to the date of the Closing, which shall be prepared by HL with the assistance of the Company, which up-dated schedule shall constitute the Trust Disbursement Schedule. HL shall use its best efforts to cause Continental to, and Continental shall thereupon be obligated to, distribute the Trust Fund as directed in the trust termination letter, including all amounts payable as set forth in the Trust Disbursement Schedule and all remaining funds to Parent. Thereafter, the Trust Fund shall terminate, except as otherwise provided therein. All liabilities and obligations of HL due and owing or incurred at or prior to the Closing Date shall be paid as and when due, including all amounts payable as set forth herein.

8.14 Intended Tax Treatment. On or after the date hereof, none of the Parties shall take any action, or fail to take any action, and following the Transactions the Parent shall prevent the Company and Surviving BVI Company from taking any action or failing to take any action, which action or failure to act would reasonably be expected to prevent or impede the Transactions from qualifying as a reorganization governed by Section 368 of the Code or an exchange governed by Section 351 of the Code, as applicable. The Parties will in all Tax Returns report the Transactions in a manner consistent with such tax treatment, and no Party will take a position inconsistent with that treatment, unless required to do otherwise pursuant to a final determination as defined in Section 1313(a) of the Code (or pursuant to any similar provision of applicable state, local or foreign Legal Requirements). For the avoidance of doubt, the receipt of any portion of the Contingent Consideration shall be treated and reported for income tax purposes as being received as additional merger consideration as part of the "reorganization" in accordance with the Code.

8.15 Incentive Equity Plan. Within six (6) months of the Closing Date, Parent shall (a) cause the Parent Plan to be adopted, the form and terms of which shall be prepared by the Board of Directors of Parent, which shall, among other things, allow Parent to make grants to its employees, officers, directors and other recipients thereunder in payments of cash or in equity that shall vest over a term to be set forth in such grants, (b) reserve for issuance such number of Parent Class A Ordinary Shares as may be required to satisfy grants made pursuant to the Parent Plan, and (c) file with the SEC a registration statement on Form S-8 (or any successor form or comparable form in another relevant jurisdiction) relating to Parent Class A Ordinary Shares issuable pursuant to the Parent Plan. Parent shall use commercially reasonable efforts to maintain the effectiveness of such registration statement for so long as any awards issued under the Parent Plan remain outstanding.

8.16 <u>Employment Agreements</u>. Prior to the Closing Date, the Company shall enter into employment agreements, in a form which is reasonably acceptable to the Company and HL, with the Company executives listed in <u>Schedule 8.16</u> on terms reasonably acceptable to such executives (the "<u>Employment Agreements</u>").

8.17 <u>Third Party Consents</u>. Each Party shall, and shall cause its Subsidiaries and Affiliates to (a) use commercially reasonable efforts to assemble, prepare, and file any information (and, as needed, to supplement such information) as may be reasonably necessary to obtain as promptly as practicable all governmental and regulatory consent required to be obtained in connection with the Transactions, (b) use commercially reasonable efforts to obtain all consents and approvals of third parties that such Party or its Subsidiaries or Affiliates is required to obtain in order to consummate the Transactions, and (c) take such other action as may reasonably be necessary or as the other Party may reasonably request to satisfy the conditions set forth in <u>Article IX</u> or otherwise to comply with this Agreement and to consummate the Transactions as soon as practicable.

8.18 <u>Amendment to the Stock Escrow Agreement</u>. HL shall enter into an amendment to the Stock Escrow Agreement to add Parent as a party, reflect the forfeiture of HL Ordinary Shares by the Sponsors pursuant to <u>Section 1.6</u>, and change all references of HL Ordinary Shares to Parent Class A Ordinary Shares.

8.19 <u>Amendment to the Warrant Agreement</u>. Parent and HL shall (a) enter into an assignment and assumption agreement pursuant to which HL will assign to Parent all of its rights, interests, and obligations in and under the Warrant Agreement and (b) amend the Warrant Agreement to change all references of HL Warrants to HL Parent Warrants (and all references of HL Ordinary Shares underlying such warrants to Parent Class A Ordinary Shares), which shall cause each outstanding HL Warrant to represent the right to receive, on the Closing Date, that number of whole Parent Class A Ordinary Shares equal to the number of HL Ordinary Shares that were issuable upon exercise of such HL Warrant immediately prior to the Effective Time, and add the Parent Warrants to be issued to the Company Shareholders in the Share Exchange (including any Parent Warrants issuable as Contingent Consideration) as warrants covered by the Warrant Agreement (collectively, the "<u>Amended Warrant Agreement</u>"). Certificates representing the HL Warrants may at any time surrender do Parent certificate(s) representing such HL Warrants and request replacement certificates representing the HL Parent Warrant received in exchange therefor, which shall not affect the interest of any such warrants within ten (10) Business Days of its receipt of a written request from the holder of an HL Warrant.

8.20 <u>Transfer Restrictions</u>. The Company Shareholders shall not transfer the Parent Class B Ordinary Shares to be received as Closing Consideration hereunder (including the Parent Class B Ordinary Shares included in the Escrow Fund) or the Parent Class A Ordinary Shares issuable as Contingent Consideration or upon exercise of Parent Warrants (if issued prior to the end of the following periods), except to Permitted Transferees, for a period ending on the one-year anniversary of the Closing Date, or earlier if, subsequent to the Closing Date, Parent consummates a liquidation, merger, stock exchange or other similar transaction which results in all of Parent's shareholders having the right to exchange their ordinary shares for cash, securities or other property. The book entry positions or certificates evidencing Parent Class A Ordinary Shares and Parent Class B Ordinary Shares issued hereunder will each include prominent disclosure or bear a prominent legend evidencing the fact that such shares are subject to such transfer restrictions.

8.21 <u>Sublease Agreement</u>. At or prior to the Closing, the Company shall execute and deliver a sublease agreement with MagP Inovação, S.A. as sublessor, and obtain all requisite approvals in connection therewith, on terms no less favorable to the Company than would be obtained from an independent third party and otherwise reasonably acceptable to HL.

8.22 <u>Registration Rights Agreement</u>. At or prior to the Closing, Parent shall execute and deliver a Registration Rights Agreement in a form to be mutually agreed upon in substance reasonable and customary for transactions of a similar nature, pursuant to which, among other things, Parent will register for resale under the Securities Act Parent Class A Ordinary Shares issued or issuable pursuant to this Agreement (including the Parent Class A Ordinary Shares issued hereunder, the Parent Class A Ordinary Shares issuable upon exercise of the Parent Warrants, any and all earned Contingent Consideration and the Parent Class A Ordinary Shares issued be upon exercise of the Parent Warrants, any and all earned Contingent Consideration and the Parent Class A Ordinary Shares issued be upon exercise of the Parent Warrants that are issued as Contingent Consideration, and the Parent Class A Ordinary Shares issued pursuant to the PIPE Subscription Agreements). Parent and HL shall enter into an amendment to that certain Registration Rights Agreement between HL and certain holders of HL Ordinary Shares and HL Warrants to add Parent as a party, reflect the forfeiture of HL Ordinary Shares by the Sponsors pursuant to <u>Section 1.6</u>, and change all references therein of HL Ordinary Shares to Parent Class A Ordinary Shares and all references therein of HL Warrants to HL Parent Warrants.

8.23 <u>Shareholders Agreement</u>. At or prior to the Closing, Parent, the Company Shareholders and the Sponsor shall enter into an agreement ("<u>Shareholders Agreement</u>") pursuant to which, among other things, the Company Shareholders will have the right to nominate up to four members of Parent's board of directors, and the Sponsors will have the right to nominate up to three members of Parent's board of directors, following the consummation of the Transactions.

8.24 <u>Reservation of Shares</u>. At all times while any HL Parent Warrants or Parent Warrants issuable hereunder are outstanding, the Parent shall authorize and reserve for issuance, and will at all times keep reserved for issuance, the maximum number of Parent Class A Ordinary Shares as will be issuable upon the exercise of all outstanding warrants to purchase Parent Class A Ordinary Shares from time to time. All such Parent Class A Ordinary Shares that shall be so issuable shall be duly and validly issued, fully paid and non-assessable and will be free from preemptive rights, all taxes, liens and charges in respect of the issue thereof (other than taxes in respect of any transfer occurring contemporaneously or otherwise specified herein).

## 8.25 Extension Proxy Statement.

(a) At such time as mutually agreed by HL and the Company, HL shall prepare (with the Company's reasonable cooperation) and file with the SEC, a mutually acceptable proxy statement (such proxy statement, together with any amendments or supplements thereto, the "<u>Extension Proxy</u> <u>Statement</u>") to amend HL's Amended and Restated Memorandum and Articles of Association to extend the time period for HL to consummate a Business Combination (as defined in the Amended and Restated Memorandum and Articles of Association) from July 2, 2020 to a date mutually agreed upon by HL and the Company (the "<u>Extension Proposal</u>"). HL shall use its best efforts to cause the Extension Proxy Statement to comply with the rules and regulations promulgated by the SEC, and to have the Extension Proxy Statement cleared by the SEC as promptly as practicable after such filing.

(b) Each of HL and the Company shall ensure that none of the information supplied by or on its behalf for inclusion or incorporation by reference in the Extension Proxy Statement will, at the date it is first mailed to the HL Shareholders and at the time of the Extension Shareholders' Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. If at any time prior to the Extension Shareholders' Meeting any information relating to HL, the Company, the Parent, Merger Sub, or any of their respective Subsidiaries, Affiliates, directors or officers is discovered by HL or the Company, which is required to be set forth in an amendment or supplement to the Extension Proxy Statement, so that such document would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which discovers such information shall promptly notify the other Parties and an appropriate amendment or supplement describing such information shall be filed promptly with the SEC and, to the extent required by applicable Legal Requirements, disseminated to the HL Shareholders.

(c) At such time as mutually agreed by HL and the Company, HL shall (i) cause the Extension Proxy Statement to be disseminated to holders of HL Ordinary Shares in compliance with applicable Legal Requirements, (ii) duly give notice of and convene and hold a meeting of its shareholders (the "<u>Extension Shareholders' Meeting</u>") in accordance with its Amended and Restated Memorandum and Articles of Association and the Nasdaq Listing Rules, (iii) solicit proxies from the HL Shareholders to vote in favor of each of the Extension Proposal, and (iv) provide the HL Shareholders with the opportunity to elect to effect a share redemption in accordance with the terms of HL's Amended and Restated Memorandum and Articles of Association. HL shall, through its board of directors, recommend to its shareholders the approval of the Extension Proposal (the "<u>HL Extension Approval</u>"), and include such recommendation in the Extension Proxy Statement. The board of directors of HL shall not withdraw, amend, qualify or modify its recommendation to the HL Shareholders that they vote in favor of the Extension Proposal. To the fullest extent permitted by applicable Legal Requirements, (x) HL's obligations to establish a record date for, duly call, give notice of, convene and hold the Extension Shareholders' Meeting shall not be affected by any modification in the recommendation of its board of directors with respect to the Extension Proposal, (y) HL shall establish a record date for, duly call, give notice of, convene and hold the Extension Proposal and (z) if the HL Extension Approval has not been obtained at any such Extension Shareholders' Meeting and submit for approval the Extension Proposal and (z) if the HL Extension Approval has not been obtained at any such Extension Shareholders' Meeting and submit for approval the Extension Proposal and (z) if the HL Extension Approval has not been obtained at any such Extension Shareholders' Meeting and submit for approval the Extension Proposal and (z) if the HL Extension Approval has not be

# ARTICLE IX

### CONDITIONS TO THE TRANSACTION

9.1 <u>Conditions to Obligations of Each Party to Effect the Transactions</u>. The respective obligations of each Party to this Agreement to effect the Transactions shall be subject to the satisfaction as of the Closing Date of the following conditions, any one or more of which may be waived (if legally permitted) in writing by all of such parties:

(a) <u>HL Shareholder Matters</u>. The HL Shareholder Matters shall have been duly approved and adopted by the affirmative vote of the HL Shareholders required under HL's Charter Documents and the applicable provisions of the BVI Companies Act.

(b) <u>HL Net Tangible Assets</u>. HL shall have at least \$5,000,001 of net tangible assets immediately prior to or upon the Closing after taking into account the exercise by Converting Shareholders of their right to convert their shares into a pro rata share of the Trust Fund in accordance with HL's Charter Documents.

(c) <u>No Order</u>. No Governmental Entity shall have enacted, issued, promulgated, enforced or entered any statute, rule, regulation, executive order, decree, injunction or other order (whether temporary, preliminary or permanent) which is in effect and which has the effect of making the Transactions illegal or otherwise prohibiting consummation of the Transactions, substantially on the terms contemplated by this Agreement.

(d) <u>Form F-4</u>. The Form F-4 (including the Proxy Statement/Prospectus) shall have become effective in accordance with the provisions of the Securities Act, no stop order shall have been issued by the SEC which remains in effect with respect to the Form F-4, and no proceeding seeking such a stop order shall have been threatened or initiated by the SEC which remains pending.

(e) Listing. The Parent Class A Ordinary Shares shall have been approved for listing on Nasdaq, subject only to official notice thereof.

(f) <u>Valuation</u>. HL shall have received a valuation of the Company from a third-party valuation firm with expertise in valuing companies in the alternative energy or Green Hydrogen industry, determining that the fair market value of the Company as of the date of this Agreement is equal to at least 80% of the balance in the Trust Account (excluding taxes payable).

(g) <u>Composition Agreement/SEAS</u>. The Parent has entered into a composition agreement with the Irish Revenue Commissioners and a Special Eligibility Agreement for Securities with The Depository Trust Company in respect of the Parent Class A Ordinary Shares, the Parent Warrants, and the HL Parent Warrants, as applicable, both of which are in full force and effect and are enforceable in accordance with their terms.

(h) <u>Amended Memorandum and Articles of Association</u>. The Memorandum and Articles of Association of Parent shall have been amended to implement the revisions set forth in <u>Section 8.2</u>.

(i) <u>PIPE Subscriptions</u>. Parent shall have entered into subscription agreements with accredited investors ("<u>PIPE Subscription</u> <u>Agreements</u>") for the private placement of no less than 2,450,000 Parent Class A Ordinary Shares, which PIPE Subscription Agreements (y) require such investors to deposit the purchase price for the Parent Class A Ordinary Shares sold pursuant to the PIPE Subscription Agreements into an escrow account with DNB Bank ASA within 3 business days of the establishment of such escrow account, and (z) close simultaneously with or immediately after a Closing and only allow for the return of escrowed funds to investors in the event that there is no Closing. For the avoidance of doubt, pursuant to the PIPE Subscription Agreements, an aggregate amount equal to or in excess of \$25,112,500 shall be available in an escrow account with DNB Bank ASA as of the Closing Date.

9.2 <u>Additional Conditions to Obligations of the Company, Company Shareholders, Parent and Merger Sub</u>. The obligations of the Company, the Company Shareholders, Parent and Merger Sub to consummate and effect the Transactions shall be subject to the satisfaction as of the Closing Date of each of the following conditions, any of which may be waived, in writing, exclusively by the Company:

(a) <u>Representations and Warranties</u>. Each representation and warranty of HL contained in this Agreement that is (i) qualified as to materiality or Material Adverse Effect shall have been true and correct (A) as of the date of this Agreement and (B) on and as of the Closing Date with the same force and effect as if made on the Closing Date (except to the extent such representation and warranty expressly relates to an earlier date, and in such case, shall be true and correct as of such earlier date) and, (ii) not qualified as to materiality or Material Adverse Effect shall have been true and correct in all material respects (A) as of the date of this Agreement, (B) on and as of the Closing Date, and (C) with respect to the representation and warranty contained in Section 5.3(g) only, also as of the SEC Approval Date with the same force and effect as if made on the Closing Date (except to the extent such representation and warranty expressly relates to an earlier date, and in such case, shall be true and correct as of such earlier date, and in such case, shall be true and correct as of such earlier date, and in such case, shall be true and correct as of such earlier date). The Company shall have received a certificate with respect to the foregoing signed on behalf of HL by an authorized officer of HL ("<u>HL Closing Certificate</u>").

(b) <u>Agreements and Covenants</u>. HL shall have performed or complied with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date, except to the extent that any failure to perform or comply (other than a willful failure to perform or comply or failure to perform or comply in all material respects with an agreement or covenant reasonably within the control of HL) does not, or is not reasonably expected to, constitute a Material Adverse Effect with respect to HL (or, after the Closing, Parent or the Surviving BVI Company), and the HL Closing Certificate shall include a provision to such effect.

(c) <u>Secretary Certificate</u>. The Company shall have received a certificate of the secretary or equivalent officer of HL certifying: (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors of HL authorizing the execution, delivery, and performance of this Agreement and the transactions contemplated hereby, and that all such resolutions are in full force and effect and are all of the resolutions of the board of directors adopted in connection with the transactions contemplated hereby, (ii) that attached thereto are certificates of good standing, or the equivalent, of HL from the Registrar of the British Virgin Islands, and from each state where it is qualified or licensed to do business as a foreign corporation, (iii) that attached thereto is evidence that the HL Shareholder Matters have been approved, and (iv) the names and signatures of the officers of HL authorized to sign this Agreement and the other documents to be delivered hereunder.

(d) <u>No Litigation</u>. No action, suit or proceeding shall be pending or threatened by any Governmental Entity which is reasonably likely to (i) prevent consummation of any of the Transactions, (ii) cause any of the Transactions to be rescinded following consummation, or (iii) affect materially and adversely the right of Parent or the Surviving BVI Company to own, operate or control any of the assets and operations of HL following the Transactions, and no order, judgment, decree, stipulation or injunction to any such effect shall be in effect.

(e) <u>Material Adverse Effect</u>. No Material Adverse Effect with respect to HL shall have occurred since the date of this Agreement, and the HL Closing Certificate shall include a provision to such effect.

(f) <u>SEC Compliance</u>. Immediately prior to Closing, HL shall be in compliance with the reporting requirements under the Securities Act and Exchange Act, and the HL Closing Certificate shall include a provision to such effect.

(g) Shareholders Agreement. The Shareholders Agreement shall have been executed and delivered and shall be in full force and effect.

(h) Sponsor Agreement. The Sponsor Agreement shall have been executed and delivered and shall be in full force and effect.

9.3 Additional Conditions to the Obligations of HL. The obligations of HL to consummate and effect the Transactions shall be subject to the satisfaction as of the Closing Date of each of the following conditions, any of which may be waived, in writing, exclusively by HL:

(a) <u>Representations and Warranties</u>. Each representation and warranty of the Company, the Company Shareholders, Parent, and Merger Sub contained in this Agreement that is (i) qualified as to materiality or Material Adverse Effect shall have been true and correct (A) as of the date of this Agreement and (B) on and as of the Closing Date with the same force and effect as if made on the Closing Date (except to the extent such representation and warranty expressly relates to an earlier date, and in such case, shall be true and correct as of such earlier date) and, (ii) not qualified as to materiality or Material Adverse Effect shall have been true and correct in all material respects (A) as of the date of this Agreement and (B) on and as of the Closing Date (except to the extent such representation and warranty expressly relates to an earlier date, and in such case, shall be true and correct as of such earlier date of this Agreement and (B) on and as of the Closing Date (except to the extent such representation and warranty expressly relates to an earlier date, and in such case, shall be true and correct as of such earlier date). HL shall have received a certificate with respect to the foregoing signed on behalf of the Company, Parent, and Merger Sub by an authorized officer of each, and signed on behalf of the Company Shareholders by the Company Shareholders Representative ("<u>Company Closing Certificate</u>").

(b) <u>Agreements and Covenants</u>. The Company, the Company Shareholders, Parent, and Merger Sub shall have performed or complied with all agreements and covenants required by this Agreement to be performed or complied with by them at or prior to the Closing Date except to the extent that any failure to perform or comply (other than a willful failure to perform or comply or failure to perform or comply in all material respects with an agreement or covenant reasonably within the control of the Company) does not, or is not reasonably expected to constitute a Material Adverse Effect respect to the Company (or, after the Closing, Parent), and the Company Closing Certificate shall include a provision to such effect.

(c) <u>Secretary Certificate</u>. HL shall have received a certificate of the secretary or equivalent officer of the Company certifying: (i) that attached thereto are true and complete copies of all resolutions adopted by the board of directors (or equivalent governing body) of each of the Company, Parent, and Merger Sub authorizing the execution, delivery, and performance of this Agreement and the transactions contemplated hereby, and that all such resolutions are in full force and effect and are all of the resolutions of the board of directors (or equivalent governing body) adopted in connection with the transactions contemplated hereby, (ii) that attached thereto are certificates of good standing, or the equivalent (if applicable), of the Company, Parent, and Merger Sub from the each state, province, or other jurisdiction where it is organized, or qualified or licensed to do business as a foreign corporation, and (iii) the names and signatures of the officers of the Company, Parent, and Merger Sub authorized to sign this Agreement and the other documents to be delivered hereunder.

(d) <u>No Litigation</u>. No action, suit or proceeding shall be pending or threatened by any Governmental Entity which is reasonably likely to (i) prevent consummation of any of the Transactions, (ii) cause any of the Transactions to be rescinded following consummation or (iii) affect materially and adversely the right of Parent to own, operate or control any of the assets and operations of the Company following the Transactions and no order, judgment, decree, stipulation or injunction to any such effect shall be in effect.

(e) <u>Share Transfer Documents</u>. The Company Shareholders shall have delivered or caused to be delivered to Parent duly completed and executed share transfer forms in relation to all of the Company Ordinary Shares and Company Class A Shares outstanding at the Effective Time, and any waivers, consents, or other documents required to vest in Parent the full legal and beneficial ownership of all of the outstanding Company Ordinary Shares and Company Class A Shares.

(f) <u>Material Adverse Effect</u>. No Material Adverse Effect with respect to the Company, Parent, or Merger Sub shall have occurred since the date of this Agreement, and the Company Closing Certificate shall include a provision to such effect.

(g) Indemnification Escrow Agreement. The Indemnification Escrow Agreement shall have been executed and delivered and shall be in full force and effect.

(h) <u>Insider Loans</u>. All outstanding loans or other indebtedness by the Company to any Insider shall have been repaid in full and all outstanding guaranties and similar arrangements pursuant to which the Company has guaranteed the payment or performance of any obligations of any Insider to a third party shall have been terminated.

(i) Amended Warrant Agreement. The Amended Warrant Agreement shall have been executed and delivered and shall be in full force and

effect.

#### ARTICLE X

#### **TERMINATION**

10.1 Termination. This Agreement may be terminated at any time prior to the Closing:

(a) by mutual written agreement of HL and the Company at any time;

(b) by either HL or the Company if the Transactions shall not have been consummated by January 2, 2021 (the "<u>Outside Date</u>"); provided, however, that the right to terminate this Agreement under this <u>Section 10.1(b)</u> shall not be available to any Party whose action or failure to act has been a principal cause of or primarily resulted in the failure of the Transactions to occur on or before such date and such action or failure to act constitutes a breach of this Agreement;

(c) by either HL or the Company if a Governmental Entity shall have issued an order, decree, judgment or ruling or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the Transactions, which order, decree, ruling or other action is final and nonappealable;

(d) by the Company, upon a material breach of any representation, warranty, covenant or agreement on the part of HL set forth in this Agreement, or if any representation or warranty of HL shall have become untrue, in either case such that the conditions set forth in <u>Article IX</u> would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become untrue, provided, that if such breach by HL is curable by HL prior to the Closing Date, then the Company may not terminate this Agreement under this <u>Section 10.1(d)</u> for thirty (30) days after delivery of written notice from the Company may not terminate this Agreement pursuant to this <u>Section 10.1(d)</u> if it shall have materially breached this Agreement or if such breach by HL is cured during such thirty (30)-day period);

(e) by HL, upon a material breach of any representation, warranty, covenant or agreement on the part of the Company, Company Shareholders, Parent, or Merger Sub set forth in this Agreement, or if any representation or warranty of the Company, Company Shareholders, Parent, or Merger Sub shall have become untrue, in either case such that the conditions set forth in <u>Article IX</u> would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become untrue, provided, that if such breach is curable by the Company, Company Shareholders, Parent, or Merger Sub prior to the Closing Date, then HL may not terminate this Agreement under this <u>Section 10.1(e)</u> for thirty (30) days after delivery of written notice from HL to the Company of such breach, provided the Company, Company Shareholders, Parent, and Merger Sub continue to exercise commercially reasonable efforts to cure such breach (it being understood that HL may not terminate this Agreement pursuant to this <u>Section 10.1(e)</u> if it shall have materially breached this Agreement or if such breach by the Company, Company Shareholders, Parent, or Merger Sub is cured during such thirty (30)-day period);

(f) by either HL or the Company, if, at the HL Special Meeting (including any adjournments thereof), the HL Shareholder Matters shall fail to be approved by the affirmative vote of the HL Shareholders required under HL's Charter Documents and the BVI Companies Act; or

(g) by either HL or the Company if, immediately prior to or upon consummation of the Transactions, HL will have less than \$5,000,001 of net tangible assets after taking into account the exercise by the Converting Shareholders of their rights to convert the HL Ordinary Shares held by them into cash in accordance with HL's Charter Documents.

## 10.2 Notice of Termination; Effect of Termination.

(a) Any termination of this Agreement under <u>Section 10.1</u> above will be effective immediately upon (or, if the termination is pursuant to <u>Section 10.1(d)</u> or <u>Section 10.1(e)</u> and the proviso therein is applicable, thirty (30) days after) the delivery of written notice of the terminating Party to the other Parties.

(b) In the event of the termination of this Agreement as provided in <u>Section 10.1</u>, this Agreement shall be of no further force or effect and the Transactions shall be abandoned, except for and subject to the following: (i) <u>Sections 7.2(a)</u>, <u>8.7</u>, <u>10.2</u> and <u>10.3</u>, and <u>Article XII</u> (General Provisions), shall survive the termination of this Agreement, and (ii) nothing herein shall relieve any Party from liability for any intentional and willful breach of this Agreement by such Party occurring prior to such termination. Without limiting the foregoing, and except as provided in this <u>Section 10.2</u>, and subject to the right to seek injunctions, specific performance or other equitable relief in accordance with <u>Section 12.6</u>, the Parties' sole right prior to Closing with respect to any breach of any representation, warranty, covenant or other agreement contained in this Agreement by another Party or with respect to the Transactions shall be the right, if applicable, to terminate this Agreement pursuant to <u>Section 10.1</u>.

10.3 <u>Fees and Expenses</u>. Except as otherwise set forth herein, all fees and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid as a Trust Disbursement if the Transactions are consummated or shall be paid by the Party incurring such expenses if the Transactions are not consummated.

## ARTICLE XI

## INDEMNIFICATION

## 11.1 Indemnification

(a) Subject to the terms and conditions of this <u>Article XI</u> (including without limitation the limitations set forth in <u>Section 11.4</u>), HL and each of its respective representatives, successors, and permitted assigns (the "<u>HL Indemnitees</u>") will be indemnified, defended, and held harmless by the Company Shareholders, jointly and severally, but only to the extent of the Escrow Fund, from and against all Losses asserted against, resulting to, imposed upon, or incurred by any HL Indemnitee by reason of, arising out of or resulting from:

(i) the inaccuracy or breach of any representation or warranty of the Company or the Company Shareholders contained in or made pursuant to this Agreement, any Schedule or any certificate delivered by the Company or the Company Shareholders pursuant to this Agreement in connection with the Closing;

(ii) the non-fulfillment or breach of any covenant or agreement of the Company contained in this Agreement; and

(iii) any claim, cause of action, right or remedy, or any action, asserted at any time by any Company Shareholder relating to the allocation or entitlement to its Pro Rata Portion of Company Consideration pursuant to this Agreement, including any assertion of contractual, employment or other rights to receive such Pro Rata Portion of Company Consideration and any assertion of rights to own or acquire any such consideration, in each case, in excess of the amount or number of securities allocable to such Company Shareholder pursuant to this Agreement.

(b) Subject to the terms and conditions of this <u>Article XI</u> (including without limitation the limitations set forth in <u>Section 11.4</u>), the Company and its representatives, successors, and permitted assigns (the "<u>Company Indemnitees</u>" and together with the HL Indemnitees, the "<u>Indemnitees</u>") will be indemnified, defended, and held harmless by the Parent, but only to the extent of the Indemnification Pool, from and against all Losses asserted against, resulting to, imposed upon, or incurred by any Company Indemnitee by reason of, arising out of or resulting from:

(i) the inaccuracy or breach of any representation or warranty of HL contained in or made pursuant to this Agreement, any Schedule or any certificate delivered by any of HL pursuant to this Agreement in connection with the Closing; and

(ii) the non-fulfillment or breach of any covenant or agreement of any of HL contained in this Agreement.

(c) As used in this <u>Article XI</u>, the term "<u>Losses</u>" includes all losses, liabilities, damages, judgments, awards, orders, penalties, settlements, costs and expenses (including, without limitation, interest, penalties, court costs and reasonable legal fees and expenses) including those arising from any demands, claims, suits, actions, costs of investigation, notices of violation or noncompliance, causes of action, proceedings and assessments whether or not made by third parties or whether or not ultimately determined to be valid. In no event shall "Losses" recoverable under this Agreement be deemed to include (i) indirect, incidental or consequential damages, lost profits, business interruption, special or punitive damages and, in particular, no "multiple of profits" or "multiple of cash flow" or similar valuation methodology shall be used in calculating the amount of any Losses, (ii) any damages caused by a change in law, administrative practice, the making of any legislation, the withdrawal of extra statutory concession previously made by any taxation or revenue authority or published interpretation of the law, after the date of this Agreement including any increase in the rates of taxation, or any imposition of Taxation or any withdrawal of relief from taxation not in effect at the date of this Agreement whether or not the change purports to be effective retrospectively in whole or part, (iii) any damages caused by any change in GAAP or IFRS, or (iv) any damages caused by any change in the accounting policies or practices of any Party after the date of this Agreement.

11.2 Claims. Any action on account of a Loss (a "Claim") may be asserted by the HL Representative on behalf of any HL Indemnitee or by the Company Shareholder Representative on behalf of any Company Indemnitee by giving Parent written notice (a "Notice of Claim") which sets forth (i) a brief description of the nature of the Claim and (ii) the total amount of the actual out-of-pocket Loss or the anticipated potential Loss (including any costs or expenses which have been or may be reasonably incurred in connection therewith). Failure to give prompt Notice of Claim or to provide copies of relevant available documents or to furnish relevant available data will not constitute a defense (in whole or in part) to any Claim and will not affect the Company Shareholders' or Parent's, as applicable in the capacity of indemnitors (the "Indemnitors"), duties or obligations under this Article XI, except to the extent (and only to the extent that) such failure has adversely affected the ability of the Indemnitors to defend against or reduce their liability or caused or increased such liability or otherwise caused the damages for which the Indemnitors are obligated to be greater than such damages would have been had the Indemnitee given Parent prompt notice hereunder. Parent will have twenty (20) calendar days after its receipt of the Notice of Claim to investigate the matter or circumstance alleged to give rise to the Claim, and the HL Representative or Company Shareholder Representative, as applicable, shall reasonably assist such investigation by giving such information and access to Persons or records as Parent may reasonably request. If Parent does not respond to the Notice of Claim within such twenty (20) calendar day period, Parent will be deemed to have irrevocably accepted the Claim, and such acceptance will be binding on the Indemnitor. If Parent rejects the Claim, it shall, within such twenty (20) calendar day period, notify the HL Representative or Company Shareholder Representative, as applicable, in writing of its rejection, specifying the factual or legal basis therefor, and Parent and the HL Representative or Company Shareholder Representative, as applicable, shall negotiate in good faith to resolve the Claim. If the parties are unable to reach an agreement within ten (10) calendar days after receipt by the HL Representative or Company Shareholder Representative, as applicable, of such rejection notice, then HL Representative or Company Shareholder Representative, as applicable, may submit the Claim to JAMS Worldwide for binding arbitration under the Comprehensive Arbitration Rules and Procedures, which arbitration will be conducted by a single arbitrator who shall be mutually agreed by Parent and the HL Representative or Company Shareholder Representative, as applicable. If the parties are unable to agree on the arbitrator, the arbitrator shall be appointed by JAMS. The arbitration shall be held in New York, New York, in the English language. The arbitrator's fees will be split equally between the parties to the arbitration and each party to the arbitration will be responsible for the payment of its own costs, attorneys' fees, expert fees and all of its other fees, costs and expenses in connection with any arbitration. The arbitrator's decision will be final and binding as to all matters of substance and procedure and may be enforced by an ex parte petition to the Supreme Court of the State of New York, County of New York, or any court having jurisdiction over the non-moving party.

11.3 Insurance Effect. To the extent that any Losses that are subject to indemnification pursuant to this <u>Article XI</u> are covered by insurance of the Company, HL, Parent, or its or their Subsidiaries, Parent shall use commercially reasonable efforts to obtain the maximum recovery under such insurance; however the Indemnitee (or HL Representative on behalf of an Indemnitee) will nevertheless be entitled to bring a claim for indemnification under this <u>Article XI</u> in respect of such Losses and the time limitations set forth in <u>Section 11.4</u> for bringing a claim of indemnification under this Agreement will be tolled during the pendency of such insurance claim. The existence of a claim by an Indemnite for monies from an insurer or against a third party in respect of any Loss will toll any disbursement of the Escrow Fund, if applicable, pursuant to the indemnification provisions contained in this <u>Article XI</u> and the Indemnification Escrow Agreement, if applicable. To the extent an Indemnitee receives proceeds from insurance or other amounts in respect of such Loss, such proceeds shall reduce the amount of Losses subject to indemnification hereunder. Notwithstanding any other provisions of this Agreement, it is the intention of the Parties that no insurer or any other third party will be (i) entitled to a benefit it would not be entitled to receive in the absence of the foregoing indemnification provisions, or (ii) relieved of the responsibility to pay any claims for which it is obligated.

## 11.4 Limitations on Indemnification.

(a) The representations, warranties, covenants and agreements in this Agreement or in any writing delivered by the Parties in connection with this Agreement will survive the Closing until on the tenth (10th) Business Day after Parent files its annual report for the year ending December 31, 2021 (the "<u>Survival Period</u>"). Any claim set forth in a Notice of Claim sent prior to the expiration of the Survival Period will survive until final resolution thereof. Except as set forth in the immediately preceding sentence, no claim for indemnification under this <u>Article XI</u> may be brought after the end of the Survival Period.

(b) In no event shall the Company Shareholders' aggregate indemnification obligations under this Agreement exceed the Escrow Fund.

(c) In no event shall Parent's aggregate indemnification obligation under this Agreement exceed the Indemnification Pool.

(d) No amount shall be payable under this <u>Article XI</u> unless and until the aggregate amount of Losses otherwise payable by the Company, on the one hand, or the Parent, on the other hand, exceeds  $\notin$ 750,000 (the "<u>Threshold</u>"), in which case the Company or Parent, as applicable, shall be liable for all Losses that exceed the Threshold.

11.5 <u>Adjustments to Consideration</u>. Amounts paid for indemnification under <u>Article XI</u> will be deemed to be an adjustment to the Company Consideration or Merger Consideration, as applicable, except as otherwise required by applicable Legal Requirements.

### 11.6 Payments; Application of Escrow Fund.

(a) Once a Loss payable to an HL Indemnitee is agreed to by the HL Representative and Parent, or finally adjudicated to be payable pursuant to this <u>Article XI</u>, then the HL Representative and Parent shall issue joint instructions to Continental to effect the forfeiture and cancellation of all or a portion of the Escrow Fund, as appropriate, to satisfy such Losses (it being acknowledged that Continental will hold the remaining portion of the Escrow Fund until final resolution of all claims for indemnification or disputes relating thereto). The Parent Class B Ordinary Shares to be cancelled in satisfaction of Losses will be valued at a price per share equal to the five-day trailing average of the mean of the high and low trading prices of Parent Class A Ordinary Shares on Nasdaq as of the five trading days immediately preceding the satisfaction of such Losses.

(b) Once a Loss payable to a Company Indemnitee is agreed to by the Company Shareholder Representative and Parent, or finally adjudicated to be payable pursuant to this <u>Article XI</u>, then the Parent shall promptly issue to the Company Indemnitee such number of Parent Class A Ordinary Shares from the Indemnification Pool equal to the dollar value of the Loss, with such Parent Class A Ordinary Shares valued at a price per share equal to the five-day trailing average of the mean of the high and low trading prices of Parent Class A Ordinary Shares on Nasdaq as of the five trading days immediately preceding the satisfaction of such Losses.

11.7 Exclusive Remedy. The Parties hereby acknowledge and agree that, from and after the Closing, the sole remedy of the Parties with respect to any and all claims for money damages arising out of or relating to this Agreement shall be pursuant and subject to the requirements of the indemnification provisions set forth in this <u>Article XI</u>. Notwithstanding any of the foregoing, nothing contained in this <u>Article XI</u> shall in any way impair, modify, or otherwise limit a Party's right to bring any claim, demand, or suit against any Person based upon actual fraud, it being understood that a mere breach of a representation and warranty does not constitute fraud.

## ARTICLE XII

### GENERAL PROVISIONS

12.1 <u>Notices</u>. All notices and other communications among the Parties shall be in writing and shall be deemed to have been duly given (i) when delivered in person, (ii) when delivered after posting in the United States mail having been sent registered or certified mail return receipt requested, postage prepaid, (iii) when delivered by FedEx or other nationally recognized overnight delivery service or (iv) when e-mailed during normal business hours (and otherwise as of the immediately following Business Day), addressed as follows:

if to HL, to:

HL Acquisitions Corp. 499 Park Avenue, 12th Floor New York, NY 10022 Attention: Jeffrey E. Schwarz E-mail: jschwarz@metrocap.net

with a copy to:

Graubard Miller The Chrysler Building 405 Lexington Avenue, 11<sup>th</sup> Floor New York, New York 10174 Attention: David Alan Miller / Jeffrey M. Gallant E-mail: dmiller@graubard.com / jgallant@graubard.com

if to the Company to:

Fusion Welcome – Fuel, S.A. Ex-Siemens Facilities Rua da Fabrica, S/N, Sabugo 2715-376, Almargem do Bispo Portugal Attention: Joao Teixeira Wahnon Email: jwahnon@fusion-fuel.eu

Feinberg Hanson LLP 855 Boylston Street, 8th Floor Boston, Massachusetts 02116 Attention: David H. Feinberg, Esq. Email: dfeinberg@feinberghanson.com

12.2 Interpretation. The definitions of the terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context shall require, any pronoun shall include the corresponding masculine, feminine and neuter forms. When a reference is made in this Agreement to an Exhibit or Schedule, such reference shall be to an Exhibit or Schedule to this Agreement unless otherwise indicated. When a reference is made in this Agreement to Sections or subsections, such reference shall be to a Section or subsection of this Agreement. Unless otherwise indicated the words "include," "includes" and "including" when used herein shall be deemed in each case to be followed by the words "without limitation." The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Reference to the Subsidiaries of an entity shall be deemed to include all direct and indirect Subsidiaries of such entity. References to a document or item of information having been "made available" will be deemed to include the posting of such document or item of information in an electronic data room accessible by HL or any of its representatives.

12.3 <u>Counterparts; Electronic Delivery</u>. This Agreement and each other document executed in connection with the Transactions, and the consummation thereof, may be executed in one or more counterparts, all of which shall be considered one and the same document and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Parties, it being understood that all Parties need not sign the same counterpart. Delivery by electronic transmission to counsel for the other Party of a counterpart executed by a Party shall be deemed to meet the requirements of the previous sentence.

12.4 Entire Agreement; Third Party Beneficiaries. This Agreement and the documents and instruments and other agreements among the Parties as contemplated by or referred to herein, including the Exhibits and Schedules hereto, and the Confidentiality Agreement (which will terminate at the Closing) (a) constitute the entire agreement among the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the Parties and any of their respective Affiliates with respect to the Transactions; and (b) are not intended to confer upon any other Person any rights or remedies hereunder (except as specifically provided in this Agreement, including Section 8.10). No representations, warranties, covenants, understandings, agreements, oral or otherwise, relating to the Transactions exist between the Parties except as expressly set forth or referenced in this Agreement and the Confidentiality Agreement.

12.5 <u>Severability</u>. In the event that any provision of this Agreement, or the application thereof, becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement will continue in full force and effect and the application of such provision to other Persons or circumstances will be interpreted so as reasonably to effect the intent of the Parties. The Parties further agree to replace such void or unenforceable provision of this Agreement will achieve, to the extent possible, the economic, business and other purposes of such void or unenforceable provision.

12.6 Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a Party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy. The Parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity. Each Party agrees that it will not oppose the granting of specific performance and other equitable relief on the basis that the other parties have an adequate remedy at law or that an award of specific performance is not an appropriate remedy for any reason at law or equity. The Parties acknowledge and agree that any Party seeking an injunction to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this Section 12.6 shall not be required to provide any bond or other security in connection with any such injunction.

12.7 Governing Law. This Agreement shall be governed by and construed in accordance with the internal law of the State of New York regardless of the law that might otherwise govern under applicable principles of conflicts of law thereof.

12.8 <u>Consent to Jurisdiction; WAIVER OF TRIAL BY JURY</u>. Each of the Parties irrevocably consents to the exclusive jurisdiction and venue of the state courts of the State of New York, located in the County of New York (or, in the case of claims to which the federal courts have exclusive subject matter jurisdiction, any federal courts of the United States of America sitting in the Southern District of New York) in connection with any matter based upon or arising out of this Agreement or the Transactions (other than disputes relating to indemnification Claims, which shall be settled in accordance with the procedures set forth in <u>Section 12.4</u>), agrees that process may be served upon them in any manner authorized by the laws of the State of New York for such Persons and waives and covenants not to assert or plead any objection which they might otherwise have to such jurisdiction, venue and manner of service of process. Each Party hereby agrees not to commence any legal proceedings relating to or arising out of this Agreement or the transactions contemplated hereby in any jurisdiction or courts other than as provided herein. EACH OF THE PARTIES IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY ACTION BASED UPON, ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

12.9 <u>Rules of Construction</u>. The Parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document.

12.10 <u>Assignment</u>. No Party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other Parties. Subject to the first sentence of this <u>Section 12.10</u>, this Agreement shall be binding upon and shall inure to the benefit of the Parties and their respective successors and permitted assigns.

12.11 <u>Amendment</u>. This Agreement may be amended by the Parties at any time only by execution of an instrument in writing signed on behalf of each of the Parties. The approval of this Agreement by the shareholders of any Party shall not restrict the ability of the board of directors of such Party to terminate this Agreement in accordance with <u>Section 10.1</u> or to cause such Party to enter into an amendment to this Agreement pursuant to this <u>Section 12.11</u>.

12.12 Extension; Waiver. At any time prior to the Closing, any party hereto may, to the extent legally allowed, (i) extend the time for the performance of any of the obligations or other acts of the other Parties, (ii) waive any inaccuracies in the representations and warranties made to such Party contained herein or in any document delivered pursuant hereto and (iii) waive compliance with any of the agreements or conditions for the benefit of such Party contained herein. Any agreement on the part of a Party to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such Party. Delay in exercising any right under this Agreement shall not constitute a waiver of such right.

12.13 <u>Currency</u>. Unless otherwise specified, all references to currency amounts in this Agreement shall mean United States dollars. Any United States dollar amounts that need to be converted into Euros shall use the current prevailing exchange rate on the applicable date quoted by the Financial Times (or such other source as the Company and the HL Representative shall reasonably agree in writing).

12.14 <u>Schedules</u>. The information furnished in the Schedules is arranged in sections corresponding to the Sections of this Agreement, and the disclosures in any section of the Schedules shall qualify (a) the corresponding Section of this Agreement and (b) other Sections of this Agreement to the extent (notwithstanding the absence of a specific cross-reference), that it is reasonably apparent on its face that such disclosure is also applicable to such other Sections of this Agreement. The Schedules and the information and disclosures contained in such Schedules are intended only to qualify and limit the representations and warranties of the parties contained in this Agreement and shall not be deemed to expand in any way the scope of any such representation or warranty. The inclusion of any information in the Schedules shall not be deemed to be an admission or acknowledgment that such information is material or outside the ordinary course of business. The inclusion of any fact or information in a Schedule is not intended to be construed as an admission or concession as to the legal effect of any such fact or information in any proceeding between any party and any Person who is not a party.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the date first written above.

HL ACQUISITIONS CORP.

- By: /s/ Jeffrey Schwarz Name: Jeffrey Schwarz Title: Chief Executive Officer
- FUSION WELCOME FUEL, S.A.
- By: /s/ João Teixeira Wahnon Name: João Teixeira Wahnon Title: Director

FUSION FUEL ATLANTIC LIMITED

By: /s/ Frederico Figueira de Chaves Name: Frederico Figueira de Chaves Title: Director

FUSION FUEL GREEN LIMITED

By: <u>/s/ João Teixeira Wahnon</u> Name: João Teixeira Wahnon Title: Director

# COMPANY SHAREHOLDERS:

# FUSION WELCOME, S.A.

By: /s/ João Teixeira Wahnon Name: João Teixeira Wahnon Title: Director

# FALCFIVE, LDA.

By: /s/ Vicente Falcão e Cunha Name: Vicente Falcão e Cunha Title: Director

NUMBERBUBBLE, S.A.

By: /s/ João Teixeira Wahnon Name: João Teixeira Wahnon Title: Director

## MAGNO EFEITO, S.A.

By: /s/ Márcia Vicente Name: Márcia Vicente Title: Director

KEY FAMILY HOLDING INVESTIMENTOS E CONSULTORIA DE GESTAO, LDA.

By: /s/ Frederico Figueira de Chaves Name: Frederico Figueira de Chaves Title: Director

#### Exhibit A Certain Definitions

"<u>Affiliate</u>" means, as applied to any Person, any other Person directly or indirectly controlling, controlled by or under direct or indirect common control with, such Person. For purposes of this definition, "control" (including with correlative meanings, the terms "controlling," "controlled by" and "under common control with"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

"Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by law to close.

"<u>Closing Consideration</u>" means (a) an aggregate of 2,125,000 Parent Class B Ordinary Shares, (b) Parent Warrants to purchase an aggregate of 2,125,000 Parent Class A Ordinary Shares, which Parent Warrants are (i) not redeemable by Parent and (ii) may be exercised for cash or on a cashless basis at the holder's option, in either case as long as the Parent Warrants are held by the Company Shareholders or their Affiliates and Permitted Transferees.

"Company Class A Shares" means the Class A shares of the Company, with nominal value of €1 per share.

"Company Ordinary Shares" means the common shares of the Company, with nominal value of €1 per share.

"Company Consideration" means an aggregate of the Closing Consideration and the Contingent Consideration issuable to Company Shareholders pursuant to this Agreement.

"Company Intellectual Property" means any Intellectual Property that is owned by the Company, including software and software programs developed by the Company.

"Company Product" means all products or service offerings of the Company and its Subsidiaries.

"Company Registered Intellectual Property" means all of the Company Intellectual Property that is the subject of an application, certificate, filing, registration or other document issued, filed with, or recorded by any government or other legal authority.

"Company Securities" means the Company Ordinary Shares, the Company Class A Shares and each other equity security of the Company issued and outstanding prior to the Effective Time.

"<u>Contingent Consideration</u>" means up to an aggregate of 1,137,000 Parent Class A Ordinary Shares and Parent Warrants to purchase up to an aggregate of 1,137,000 Parent Class A Ordinary Shares, issuable in accordance with the terms as set forth in <u>Section 2.4</u>. For the avoidance of doubt, at any time that Contingent Consideration is issuable hereunder, each one Parent Class A Ordinary Share shall be issued with one Parent Warrant, which Parent Class A Ordinary Share and Parent Warrant together shall have an agreed value of  $\notin 10.73$ .

"Copyrights" means all copyrights, copyrights registrations and applications therefor, and all other rights corresponding thereto throughout the world.

"<u>Environmental Law</u>" means any federal, state, local or foreign law, regulation, order, decree, permit, authorization, opinion, common law or agency requirement relating to: (i) the protection, investigation or restoration of the environment, health and safety (in relation to exposure to Hazardous Substances), or natural resources; (ii) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance or (iii) noise, odor, wetlands, pollution, contamination or any injury or threat of injury to persons or property.

"<u>Export Control Laws</u>" means (i) all U.S. import and export laws (including those laws under the authority of U.S. Departments of Commerce (Bureau of Industry and Security) codified at 15 CFR, Parts 700-799; Homeland Security (Customs and Border Protection) codified at 19 CFR, Parts 1-199; State (Directorate of Defense Trade Controls) codified at 22 CFR, Parts 103, 120-130; and Treasury (Office of Foreign Assets Control) codified at 31 CFR, Parts 500-599) and (ii) all comparable applicable laws outside the United States.

"Form F-4" shall mean the registration statement on Form F-4 of Parent with respect to registration of the Parent Class A Ordinary Shares and HL Parent Warrants to be issued as HL Merger Consideration.

"Governmental Entity" means any federal, state, provincial, municipal, foreign, or other court, judicial body, administrative agency, commission, governmental or regulatory authority or similar body.

"Green Hydrogen" means hydrogen produced with renewable energy that results in no carbon emissions.

"<u>Hazardous Substance</u>" means any substance that is: (i) listed, classified or regulated pursuant to any Environmental Law; (ii) any petroleum product or byproduct, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, radioactive materials or radon; or (iii) any other substance which is the subject of regulatory action by any Governmental Entity pursuant to any Environmental Law.

"<u>HL Merger Consideration</u>" means the Parent Class A Ordinary Shares and HL Parent Warrants issuable to securityholders of HL pursuant to this Agreement.

"HL Ordinary Shares" means the ordinary shares of HL, par value \$0.0001 per share.

"HL Preferred Shares" means the preferred shares of HL, par value \$0.0001 per share.

"HL Securities" means the HL Ordinary Shares, HL Warrants, HL Rights, HL's units, and each other equity security of HL issued and outstanding immediately prior to the Effective Time.

"HL Shareholders" means the holders of HL Ordinary Shares.

"HL UPOs" means those certain unit purchase options to purchase an aggregate of 250,000 units of HL issued in HL's initial public offering.

"IFRS" means the International Financial Reporting Standard as adopted by the European Union.

"<u>Information Privacy and Security Laws</u>" means all applicable laws concerning the privacy, data protection, transfer, or security of Personal Confidential Information, including, to the extent applicable, the General Data Protection Regulation (EU), other state, and federal, data security laws, data breach notification laws, and consumer protection laws.

"Indemnification Pool" means 212,500 newly issued Parent Class A Ordinary Shares to be issued by Parent in accordance with the provisions of <u>Article XI</u>.

"Insider" means any individual who is an officer, director or employee of the Company or any of its Subsidiaries.

"Insurance Policies" means all material insurance policies and material fidelity and surety bonds covering the assets, business, equipment, properties, operations, employees, officers and directors.

"<u>Intellectual Property</u>" means any or all of the following and all worldwide common law and statutory rights in, arising out of, or associated therewith: (i) Patents; (ii) inventions (whether patentable or not), invention disclosures, improvements, trade secrets, proprietary information, know how, technology, technical data and customer lists, and all documentation relating to any of the foregoing; (iii) Copyrights; (iv) software and software programs; (v) domain names, uniform resource locators and other names and locators associated with the Internet (vi) industrial designs and any registrations and applications therefor; (vii) Trademarks; (viii) all databases and data collections and all rights therein; (ix) all moral and economic rights of authors and inventors, however denominated, and (x) any similar or equivalent rights to any of the foregoing (as applicable).

"<u>knowledge</u>" means actual knowledge or awareness as to a specified fact or event (i) in the case of the Company or Parent, of Joao Wahnon or Frederico Figueira de Chaves, and (ii) in the case of HL or Merger Sub, Jeffrey Schwarz or Greg Drechsler.

"Legal Requirements" means any federal, state, local, municipal, foreign or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, regulation, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Entity.

"Lien" means any mortgage, pledge, security interest, encumbrance, lien, restriction or charge of any kind (including, without limitation, any conditional sale or other title retention agreement or lease in the nature thereof, any sale with recourse against the seller or any Affiliate of the seller, or any agreement to give any security interest).

"Material Adverse Effect" when used in connection with the Company or HL, as the case may be, means any change, event, occurrence or effect, individually or when aggregated with other changes, events, occurrences or effects, that has a materially adverse effect on the business or financial condition of the Company and its Subsidiaries, taken as whole, or HL, as applicable, provided however that none of the following (or the effect of any of the following) alone or in combination shall be deemed, in and of itself, to constitute, or be taken into account in determining whether there has been or will be, a Material Adverse Effect: any changes, events, occurrences or effects arising out of, resulting from or attributable to (i) acts of war, sabotage or terrorism, or any escalation or worsening of any such acts of war, sabotage or terrorism, (ii) earthquakes, hurricanes, tornados or other natural or man-made disasters, acts of God or other force majeure events, (iii) any proposal, enactment or change in interpretation of, or other change in, applicable Legal Requirements, IFRS, U.S. GAAP (or equivalent accounting practice in any other jurisdiction) or governmental policy or any development or effect of any investigation, audit or review of the Company or any of its Subsidiaries by any Governmental Entity commencing from and after the date hereof, (iv) general conditions in the industries in which the Company or any of its Subsidiaries operate, (v) the failure, in and of itself, of the Company or any of its Subsidiaries to meet any internal or published projections, forecasts, estimates or predictions in respect of revenue, earnings or other financial or operating metrics before, on or after the date of this Agreement (it being understood that the underlying facts giving rise or contributing to such failure or change may be taken into account in determining whether there has been a Material Adverse Effect if otherwise contemplated by this definition), (vi) changes attributable to the public announcement or pendency of the Transactions or the performance of this Agreement, including the impact thereof on relationships, contractual or otherwise, with customers, suppliers, licensors, distributors, partners, providers and employees, (vii) any change in interest rates or economic, political, business, financial, commodity, currency or market conditions generally, or (viii) any action taken or omitted to be taken by the Company or its Subsidiaries at HL's direction or written request, on the one hand, or by HL at the Company's direction or written request, on the other hand, including in either case actions not taken as a result of the failure of the other Party to consent to any action requiring such Party's consent, or any action otherwise required or permitted to be taken or omitted to be taken by this Agreement or to which the other Party has consented in writing; provided, however, in the case of the foregoing clauses (i), (ii), (iii), (iv) and (vii), in the event that the Company and its Subsidiaries, taken as a whole, are materially and disproportionately affected by such change, event, occurrence or effect relative to other participants in the business and industries in which the Company and its Subsidiaries operate, the extent (and only the extent) of such adverse effect, relative to such other participants, on the Company or any of its Subsidiaries may be taken into account in determining whether there has been a Material Adverse Effect.

"Parent Class A Ordinary Shares" means the Class A ordinary shares of Parent, par value \$0.0001 per share.

"Parent Class B Ordinary Shares" means the Class B ordinary shares of Parent, par value \$0.0001 per share.

"Parent Plan" means a management incentive equity plan to be adopted by Parent.

"Parent Warrants" means the warrants to purchase Parent Class A Ordinary Shares.

"Patents" means all patents and applications therefor and all reissues, divisions, renewals, extensions, provisionals, continuations and continuations-in-part thereof.

"<u>Permitted Liens</u>" means (i) statutory Liens for Taxes, assessments or other governmental charges, in each case, not yet delinquent or the amount or validity of which is being contested in good faith and for which adequate reserves have been established in accordance with U.S. GAAP or IFRS, as applicable, (ii) mechanics', carriers', workers', repairers' and similar Liens arising or incurred in the ordinary course of business, (iii) zoning, entitlement and other land use and environmental regulations promulgated by any Governmental Entity, (iv) covenants, conditions, restrictions, easements, rights of way, encumbrances, defects, imperfections, irregularities of title or other Liens, if any, that would not reasonably be expected to have a Material Adverse Effect, (v) with respect to any leased real property, (a) the interests and rights of the respective lessors with respect thereto and (b) any Lien permitted under the applicable lease agreement and any ancillary documents thereto, (vi) Liens created by HL or its successors and assigns, (vii) Liens disclosed in the Company Schedule or the HL Schedule, (viii) Liens (other than monetary liens) incurred in the ordinary course of business since the date of the most recent Financial Statement, (ix) licenses to Intellectual Property granted in the ordinary course of business, (x) Liens securing the Company's and its Subsidiaries' existing credit facilities, (xii) statutory or contractual Liens of lessors or Liens on the lessor's or prior lessor's interest, and (xiii) Liens of public record.

"<u>Permitted Transferee</u>" means (i) Parent or an Affiliate of Parent, (ii) if the transferor is an entity, (x) its shareholders, partners, or members upon the transferee's liquidation or (y) an entity, if such entity's equity securities are 100% owned by the transferor or its shareholders, partners, or members, or (iii) if the transferor is an individual, (x) a member of the transferor's immediate family or a trust, the beneficiary of which is the transferor or a member of the transferor's immediate family, who receives Parent Securities from the transferor by bona fide gift for estate planning purposes, (y) a Person who receives Parent Securities from the transferor by virtue of the laws of descent and distribution upon the death of the transferor, or (z) a Person who receives Parent Securities from the transferor pursuant to a qualified domestic relations order binding on the transferor. As used herein, "immediate family," means a spouse, parent, lineal descendants, the spouse of any lineal descendent, brothers or sisters, or a trust, all of whose current beneficiaries are members of the immediate family of the transferor.

"<u>Person</u>" means any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization, entity or Governmental Entity.

"<u>Personal Confidential Information</u>" means any information, in any form, that could reasonably be used to identify, contact, or locate a single person, that is governed, regulated, or protected by one or more Information Privacy and Security Laws.

"Power Purchase Agreement" means a contract between the Company and a third-party purchaser pursuant to which the Company shall supply Green Hydrogen to such third-party purchaser.

"<u>Pro Rata Percentage of Closing Consideration</u>" means the percentage interest held by a Company Shareholder calculated by dividing a the number of Company Ordinary Shares held by such Company Shareholder immediately prior to the Effective Time by the total number of issued and outstanding Company Ordinary Shares prior to the Effective Time.

"<u>Pro Rata Percentage of Contingent Consideration</u>" means the percentage interest held by a Contingent Consideration Shareholder calculated by dividing the number of Company Class A Shares held by such Contingent Consideration Shareholder immediately prior to the Effective Time by the total number of issued and outstanding Company Class A Shares prior to the Effective Time

"<u>Pro Rata Portion of Closing Consideration</u>" means (i) the number of Parent Class B Ordinary Shares calculated by multiplying a Company Shareholder's Pro Rata Percentage of Closing Consideration by 2,125,000; and (ii) the number of Parent Warrants equal to the number of Parent Class B Ordinary Shares calculated pursuant to Section (i) of this definition.

"Pro Rata Portion of Company Consideration" means a Shareholder's Pro Rata Portion of Closing Consideration and/or Pro Rata Portion of Contingent Consideration, as applicable.

"<u>Pro Rata Portion of Contingent Consideration</u>" means (i) the number of Parent Class A Ordinary Shares calculated by multiplying a Contingent Consideration Shareholder's Pro Rata Percentage of Contingent Consideration by the number of Parent Class A Ordinary Shares issuable as Contingent Consideration in accordance with <u>Section 2.4</u>, and (ii) the number of Parent Warrants equal to the number of Parent Class A Ordinary Shares calculated pursuant to Section (i) of this definition issuable as Contingent Consideration in accordance with <u>Section 2.4</u>.

"<u>Proxy Statement/Prospectus</u>" means the proxy statement/prospectus included in the Form F-4, including the proxy statement filed by HL on Schedule 14A with respect to the HL Special Meeting to approve the HL Shareholder Matters and the registration statement registering the HL Merger Consideration, relating to the transactions contemplated by this Agreement which shall constitute a proxy statement of HL to be used for the HL Special Meeting to approve the HL Shareholder Matters (which shall also provide the Converting Shareholders with the opportunity to have their HL Ordinary Shares converted to cash in conjunction with a shareholder vote on the Merger Proposal) and a prospectus with respect to the Parent Class A Ordinary Shares and HL Parent Warrants to be offered and issued in the Merger in all cases in accordance with and as required by HL's Charter Documents and the Corporate Law.

"SEC" means the U.S. Securities and Exchange Commission.

"Sponsors" means Metropolitan Capital Partners V, LLC, HL Acquisitions Holdings LLC, and Jeffrey Schwarz Children's Trust, and, in the discretion of HL, any other Persons holding HL Ordinary Shares or HL Warrants.

"<u>Tax</u>" or "<u>Taxes</u>" refers to any and all federal, state, local and foreign taxes, including, without limitation, gross receipts, income, profits, sales, use, occupation, value added, ad valorem, transfer, franchise, withholding, payroll, recapture, employment, excise and property taxes, assessments, governmental charges in the nature of a tax and duties together with all interest, penalties and additions imposed with respect to any such amounts and including any liability of a predecessor entity for any such amounts, by contract or otherwise.

"Trademarks" means trade names, logos, common law trademarks and service marks, trademark and service mark registrations and applications therefor.

"<u>Transaction Expenses</u>" means all financial advisory, legal, accounting, underwriting, finder's, brokerage, agent's and other commissions, fees and expenses incurred by any Party in connection with the preparation and execution of this Agreement, the documents ancillary hereto, the compliance herewith and therewith, and the transactions contemplated hereby and thereby.

### AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

This AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "Amended and Restated Registration Rights Agreement") is entered into as of December 10, 2020, by and among HL Acquisitions Corp., a British Virgin Islands company ("<u>HL</u>"), Fusion Fuel Green PLC, a public limited company incorporated in Ireland ("<u>Parent</u>"), the Fusion Fuel Shareholders (as defined below), the individuals and entities listed under HL Investors on the signature pages hereto (each, an "<u>HL Investor</u>" and collectively, the "<u>HL Investors</u>"), the individuals and entities listed under HL Affiliates on the signature pages hereto (each, an "<u>HL Affiliate</u>" and collectively, the "<u>HL Affiliates</u>"), EarlyBirdCapital, Inc. ("<u>EBC</u>") and the designees of EBC listed under EBC Designees on the signature pages hereto (collectively, the "<u>EBC Designees</u>") and the individuals listed under Directors on the signature pages hereto, either in their individual capacities or on behalf of an entity controlled by them (each, a "<u>Director</u>" and collectively, the "<u>INvestors</u>" and collectively, the "<u>Investors</u>" and collectively, the "<u>Directors</u>" and collectively, the "<u>Directors</u>" and together with the Fusion Fuel Shareholders, HL Investors, the HL Affiliates EBC and the EBC Designees, the "<u>Investors</u>" and each an "<u>Investor</u>"), amends and restates in its entirety that certain Registration Rights Agreement by and among HL and the HL Investors dated June 27, 2018 ("<u>Prior Agreement</u>")

WHEREAS, the Prior Agreement provides for certain rights of registration of the securities of HL held by the Investors;

WHEREAS, HL has entered into a Business Combination Agreement, dated as of June 6, 2020 (as amended and restated on August 25, 2020, and as may be further amended from time to time, the "Business Combination Agreement"), with Parent, Fusion Fuel Atlantic Limited, a British Virgin Islands business company and wholly owned subsidiary of Parent ("Merger Sub"), Fusion Welcome - Fuel, S.A., a public limited company domiciled in Portugal, sociedade anónima ("Fusion Fuel"), and the shareholders of Fusion Fuel ("Fusion Fuel Shareholders") and, as a result of the transactions contemplated by the Business Combination Agreement, among other things, (i) each outstanding ordinary share of HL will be converted into one Class A ordinary share of Parent ("Parent Class A Ordinary Shares") except that holders of HL ordinary shares sold in HL's initial public offering will be entitled to elect instead to receive a pro rata portion of HL's trust account, as provided in HL's amended and restated memorandum and articles of association ("<u>M&A</u>"), (ii) each outstanding right of HL will be exchanged for one-tenth of one ordinary share of HL immediately prior to the effective time of the Merger (as defined in the Business Combination Agreement), and each such ordinary share of HL will be converted into one Parent Class A Ordinary Share, (iii) each outstanding warrant of HL will remain outstanding and will be automatically adjusted to entitle the holder to purchase one Parent Class A Ordinary Share at a price of \$11.50 per share ("HL Parent Warrant"); (iv) Parent will privately issue to the Fusion Fuel Shareholders (a) 2,125,000 Class B ordinary shares of Parent ("Class B Ordinary Shares"), each such Class B Ordinary Share to be convertible at any time into one Parent Class A Ordinary Share at the option of the holder and all outstanding Class B Ordinary Shares to be automatically converted into an equal number of Parent Class A Ordinary Shares on December 31, 2023, (b) warrants to purchase 2,125,000 Parent Class A Ordinary Shares (the "FF Parent Warrants"), and (c) the right to receive upon achievement of certain milestones 1,137,000 Parent Class A Ordinary Shares (the "Contingent Class A Ordinary Shares") and warrants to purchase 1,137,000 Parent Class A Ordinary Shares (the "Contingent FF Parent Warrants");

WHEREAS, pursuant to Section 1.6 of the Business Combination Agreement, certain of the Investors have agreed to the forfeiture and cancellation of an aggregate of 125,000 ordinary shares of HL and 125,000 warrants of HL ("Forfeited Securities");

WHEREAS, pursuant to Section 1.3(d) of the Business Combination Agreement, EBC, on behalf of itself and the EBC Designees, has agreed to exchange outstanding purchase options of HL for an aggregate of 50,000 ordinary shares of HL, which ordinary shares of HL shall be converted into an equal number of Parent Class A Ordinary Shares ("UPO Shares");

WHEREAS, pursuant to Section 8.22 of the Business Combination Agreement, HL, Parent and Fusion Fuel agreed to amend the Prior Agreement such that, after giving effect to the transactions contemplated by the Business Combination Agreement and the forfeiture of the Forfeited Securities, the Parent securities held by the Investors (other than the Forfeited Securities) shall bear the same registration rights as currently held by the HL Investors party to the Prior Agreement;

WHEREAS, HL, Parent, and the Investors desire to amend and restate the Prior Agreement as set forth in this Amended and Restated Registration Rights Agreement;

WHEREAS, each of the parties to the Prior Agreement is a signatory to this Amended and Restated Registration Rights Agreement, satisfying the requirements for amendments as set forth in Section 6.7 of the Prior Agreement.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. CERTAIN DEFINITIONS. The following capitalized terms used herein have the following meanings:

"<u>As-Converted Basis</u>" means as of any date of determination, (a) with respect to the Parent Class A Ordinary Shares, all issued and outstanding Parent Class A Ordinary Shares, (b) with respect to the Class B Ordinary Shares, the number of Parent Class A Ordinary Shares issuable upon the conversion thereof, (c) with respect to any of the HL Parent Warrants, FF Parent Warrants, or Contingent FF Parent Warrants, the number of Parent Class A Ordinary Shares issuable upon the exercise thereof, or (d) with respect to any other type, class, or series of securities, all Parent Class A Ordinary Shares issuable upon the exercise or conversion thereof as of such date, whether or not exercisable or convertible at such time.

"Closing" means the closing of the Business Combination Agreement.

"Commission" means the Securities and Exchange Commission, or any other federal agency then administering the Securities Act or the Exchange Act.

"<u>Director Shares</u>" means the Parent Class A Ordinary Shares granted to Jeffrey Schwarz and the Directors as compensation for their service as a director of Parent, which Director Shares are subject to the transfer restrictions as set forth in the Director Agreement entered into by Parent and such Investors.

"Escrow Shares" means the Parent Class A Ordinary Shares owned by the HL Investors and held in escrow pursuant to that certain Amended and Restated Stock Escrow Agreement dated on or about the date hereof by and among HL, Parent, the HL Investors, Continental Stock Transfer & Trust Company, and the other parties thereto.

"Fusion Fuel Securities" means (i) all of the FF Parent Warrants, (ii) all of the Parent Class A Ordinary Shares issuable upon the exercise of the FF Parent Warrants, (iii) all of the Parent Class A Ordinary Shares issuable upon the conversion of Class B Ordinary Shares, (iii) all of the Contingent Class A Ordinary Shares, (iv) all of the Contingent FF Parent Warrants, and (v) all of the Parent Class A Ordinary Shares issuable upon the exercise of Contingent FF Parent Warrants.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission promulgated thereunder, all as the same shall be in effect at the time.

"<u>HL Securities</u>" means (i) all of the Parent Class A Ordinary Shares issued to the HL Affiliates, EBC, and the EBC Designees, including the Escrow Shares, (ii) all of the HL Parent Warrants held by the HL Affiliates, (iii) the HL Parent Warrants held by any HL Affiliates or HL Investors following the automatic adjustment of certain outstanding warrants of HL created upon conversion of convertible working capital loans made by an HL Affiliate or HL Investor to HL prior to the Closing, and (iv) all of the Parent Class A Ordinary Shares underlying all HL Parent Warrants (which, for the avoidance of doubt, includes HL Parent Warrants held by the HL Investors and HL Affiliates).

"Lockup Expiration Date" means the date that is the one-year anniversary of the Closing of the Business Combination Agreement.

"<u>PIPE Investors</u>" means those certain investors who entered into subscription agreements with the Parent, dated on or around August 25, 2020, for the purchase of Parent Class A Ordinary Shares in the aggregate amount of \$25,112,500.

"<u>PIPE Registration Statement</u>" means a Registration Statement to be filed within thirty (30) days of the closing of the Business Combination Agreement as contemplated by those certain subscription agreements between the Parent and the PIPE Investors.

"<u>Register</u>," "<u>Registered</u>" and "<u>Registration</u>" mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"<u>Registrable Securities</u>" means (i) the HL Securities, (ii) the UPO Shares, (iii) the Fusion Fuel Securities and (iv) Director Shares, in each case beneficially owned or held by the Investors. Registrable Securities include any warrants, units, shares of capital stock or other securities of Parent issued as a dividend or other distribution with respect to or in exchange for or in replacement of the securities listed in items (i) through (iv) of the previous sentence (and underlying securities). As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when: (w) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (x) such securities shall have been otherwise transferred, new certificates for them not bearing a legend restricting further transfer shall have been delivered by Parent and subsequent public distribution of them shall not require registration under the Securities Act; (y) such securities shall have ceased to be outstanding, or (z) the Registrable Securities are freely saleable under Rule 144 under the Securities Act without volume limitations.

"<u>Registration Statement</u>" means a registration statement filed by Parent with the Commission in compliance with the Securities Act and the rules and regulations promulgated thereunder for a public offering and sale of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities (other than a registration statement on Form F-4 or Form S-8, or their successors, or any registration statement covering only securities proposed to be issued in exchange for securities or assets of another entity).

"<u>Release Date</u>" means the date on which the Escrow Shares are disbursed from escrow pursuant to Section 3 of that certain Amended and Restated Stock Escrow Agreement dated on or about the date hereof by and among HL, Parent, the Investors and Continental Stock Transfer & Trust Company.

"Securities Act of 1933, as amended, and the rules and regulations of the Commission promulgated thereunder, all as the same shall be in effect at the time.

"<u>Triggering Date</u>" means (i) with respect to the Escrow Shares, the date falling three months prior to the Release Date; (ii) with respect to the Director Shares, the date that the transfer restrictions lapse with respect to such Director Shares; (iii) with respect to the Fusion Fuel Securities, the Lockup Expiration Date, and (iv) with respect to all other Registrable Securities, the date hereof.

"Underwritter" means a securities dealer who purchases any Registrable Securities as principal in an underwritten offering and not as part of such dealer's market-making activities.

## 2. <u>REGISTRATION RIGHTS</u>.

## 2.1 Demand Registration.

2.1.1 <u>Request for Registration</u>. At any time and from time to time on or after the Triggering Date, the holders of a majority-in-interest of all the outstanding Registrable Securities held by any of (i) the HL Investors, EBC, and the EBC Designees in respect of the Registrable Securities held by them, (ii) the Fusion Fuel Shareholders in respect of the Registrable Securities held by them, as the case may be (but not including Escrow Shares to the extent the Release Date has not occurred, the Fusion Fuel Securities to the extent the Lockup Expiration Date has not occurred, the Director Shares to the extent any contractual transfer restrictions other than restrictions imposed by securities laws have not lapsed, or the UPO Shares, which are covered by <u>Section 2.1.2</u>), may make a written demand for registration under the Securities Act of all or part of their Registrable Securities (a "Demand Registration"). Any demand for a Demand Registration shall specify the type and number of Registrable Securities proposed to be sold and the intended method(s) of distribution thereof. Within ten (10) days of Parent's receipt of the Demand Registrable Securities in such registration, a "Demanding Holder") shall so notify Parent within ten (10) days after the receipt by the holder of the notice from Parent. Upon any such request, the Demanding Holders shall be entitled to have their Registrable Securities included in the Demand Registration, subject to Section 2.1.5 and the provisos set forth in Section 3.1.1. Parent shall not be obligated to effect more than an aggregate of two (2) Demand Registrations under this Section 2.1.1 in respect of all Registrable Securities.

2.1.2 <u>UPO Shares Request for Registration</u>. At any time and from time to time on or after the Triggering Date, the holders of a majorityin-interest of the UPO Shares may make a written demand for registration under the Securities Act of all or part of their UPO Shares (a "<u>UPO Demand Registration</u>"). Any demand for a UPO Demand Registration shall specify the number of UPO Shares proposed to be sold and the intended method(s) of distribution thereof. Within ten (10) days of Parent's receipt of the UPO Demand Registration, Parent will notify all holders of Registrable Securities of the demand and each Demanding Holder shall notify Parent of its wish to include all or a portion of such holder's Registrable Securities in the UPO Demand Registration within fifteen (15) days after the receipt by the holder of the notice from Parent. Upon any such request, the Demanding Holders shall be entitled to have their Registrable Securities included in the UPO Demand Registration, subject to Section 2.1.5 and the provisos set forth in Section 3.1.1. Parent shall not be obligated to effect more than an aggregate of one (1) UPO Demand Registration under this Section 2.1.2 in respect of all UPO Shares.

2.1.3 Effective Registration. A registration will not count as a Demand Registration or a UPO Demand Registration until the Registration Statement filed with the Commission with respect to such Demand Registration or UPO Demand Registration has been declared effective and Parent has complied with all of its obligations under this Amended and Restated Registration Rights Agreement with respect thereto; provided, however, that if, after such Registration Statement has been declared effective, the offering of Registrable Securities pursuant to a Demand Registration or UPO Demand Registration is interfered with by any stop order or injunction of the Commission or any other governmental agency or court, the Registration Statement with respect to such Demand Registration or UPO Demand Registration will be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders, on an As-Converted Basis, thereafter affirmatively elect to continue the offering and notify Parent in writing, but in no event later than five (5) days of such election; provided, further, that Parent shall not be obligated to file a second Registration Statement until a Registration Statement that has been filed is counted as a Demand Registration or UPO Demand Registration or uPO Demand Registration statement that has been filed is counted as a Demand Registration or UPO Demand Registration or is terminated.

2.1.4 <u>Underwritten Offering</u>. If a majority-in-interest of the Demanding Holders, on an As-Converted Basis, so elect and such holders so advise Parent as part of their written demand for a Demand Registration or UPO Demand Registration, the offering of such Registrable Securities pursuant to such Demand Registration or UPO Demand Registration shall be in the form of an underwritten offering. In such event, the right of any holder to include its Registrable Securities in such registration shall be conditioned upon such holder's participation in such underwriting and the inclusion of such holder's Registrable Securities in the underwriting to the extent provided herein. All Demanding Holders proposing to distribute their Registrable Securities through such underwriting shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for such underwriting by a majority-in-interest of the holders initiating the Demand Registration or UPO Demand Registration, as applicable, on an As-Converted Basis.

2.1.5 <u>Reduction of Offering</u>. If the managing Underwriter or Underwriters for a Demand Registration or UPO Demand Registration that is to be an underwritten offering advises Parent and the Demanding Holders in writing that the dollar amount or number of any Registrable Securities which the Demanding Holders desire to sell, taken together with all other Parent securities, if any, as to which registration has been requested pursuant to written contractual piggy-back registration rights held by other shareholders of Parent who desire to sell (the "<u>Outside Registrable Securities</u>" and together with the Registrable Securities, the "<u>Offering Securities</u>") and together with Parent Class A Ordinary Shares, warrants of Parent, or other equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, which Parent desires to sell for its own account and/or for shareholders of Parent for their account(s) (the "<u>Parent Offering Securities</u>"), exceeds the maximum dollar amount or maximum number of shares, warrants and/or other securities that can be sold in such offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of Securities, as applicable, the "<u>Maximum Number of Securities</u>"), then Parent shall include in such registration: (i) first, the amount of Offering Securities that can be sold without exceeding the Maximum Number of securities, such that the number of each type, series or class of Offering Security registrable by a holder in a Demand Registration or UPO Demand Registration shall be determined on a pro rata basis in respect of the total number of the same type, series or class of Offering Securities that can be sold Offering Security registrable by a holder in a Demand Registration or Securities has not been reached under the foregoing clause (i), the Parent Offering Securities that can be sold without exceed

2.1.6 <u>Withdrawal</u>. If a majority-in-interest of the Demanding Holders, on an As-Converted Basis, disapprove of the terms of any underwriting or are not entitled to include all of their Registrable Securities in any offering, such majority-in-interest of the Demanding Holders may elect to withdraw from such offering by giving written notice to Parent and the Underwriter or Underwriters of their request to withdraw prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Demand Registration. If the majority-in-interest of the Demanding Holders, on an As-Converted Basis, withdraws from a proposed offering relating to a Demand Registration or UPO Demand Registration, then such registration shall not count as a Demand Registration provided for in Section 2.1.1 or 2.1.2, as applicable.

### 2.2 Piggy-Back Registration.

2.2.1 <u>Piggy-Back Rights</u>. If at any time on or after the Closing Parent proposes to file a Registration Statement under the Securities Act with respect to an offering of Parent Offering Securities (including, without limitation, pursuant to Section 2.1 and the PIPE Registration Statement), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to Parent's existing shareholders, (iii) for an offering of debt that is convertible into equity securities of Parent or (iv) for a dividend reinvestment plan, then Parent shall (x) give written notice of such proposed filing to the holders of all Registrable Securities (but in each case only if such offering is proposed to be made after the applicable Triggering Date), as soon as practicable but in no event less than ten (10) days before the anticipated filing date, which notice shall describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, of the offering, and (y) offer to the holders of Registrable Securities in such notice the opportunity to register the sale of such number of shares of Registrable Securities as such holders may request in writing within five (5) days following receipt of such managing Underwriter or Underwriters of a proposed underwriten offering to permit the Registrable Securities requested to be included in a Piggy-Back Registration"). Parent shall cause such Registrable Securities of Parent and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All holders of Registrable Securities proposing to distribute their securities through a Piggy-Back Registration that involves an Underwriter or Underwriters shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for suc

2.2.2 <u>Reduction of Offering</u>. If the managing Underwriter or Underwriters for a Piggy-Back Registration that is to be an underwritten offering advises Parent and the holders of Registrable Securities in writing that the dollar amount or number of Parent Offering Securities, taken together with the Outside Registrable Securities, the Registrable Securities as to which registration has been requested under this Section 2.2 (the "<u>Piggy-Back</u> <u>Registrable Securities</u>"), and the Parent securities, if any, as to which registration has been requested pursuant to the written contractual piggy-back registration rights of other shareholders of Parent (the "<u>Outside Piggy-Back Registrable Securities</u>"), exceeds the Maximum Number of Securities, then Parent shall include in any such registration:

(a) If the registration is undertaken for Parent's account: (i) the Parent Offering Securities that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Piggy-Back Registrable Securities as to which registration has been requested, Pro Rata, that can be sold without exceeding the Maximum Number of Securities; and (iii) third, to the extent the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Outside Piggy-Back Registrable Securities as to which registration has been requested, Pro Rata, that can be sold without exceeding the Maximum Number of Securities; and

(b) If the registration is a "demand" registration undertaken at the demand of holders of Outside Registrable Securities, (i) first, the Outside Registrable Securities that can be sold without exceeding the Maximum Number of securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Piggy-Back Registrable Securities as to which registration has been requested, Pro Rata, that can be sold without exceeding the Maximum Number of Securities; (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Outside Piggy-Back Registrable Securities, Pro Rata, that can be sold without exceeding the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), extended the foregoing clauses (i), fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), extended the foregoing clauses (i), fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), extended the foregoing clauses (i), fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), fourth, the foregoing clause (i), fourth, the foregoing clause (

2.2.3 <u>Withdrawal</u>. Any holder of Registrable Securities may elect to withdraw such holder's request for inclusion of Registrable Securities in any Piggy-Back Registration by giving written notice to Parent of such request to withdraw prior to the effectiveness of the Registration Statement. Parent (whether on its own determination or as the result of a withdrawal by persons making a demand pursuant to written contractual obligations) may withdraw a Registration Statement at any time prior to the effectiveness of such Registration Statement. Notwithstanding any such withdrawal, Parent shall pay all expenses incurred by the holders of Registrable Securities in connection with such Piggy-Back Registration as provided in Section 3.3.

2.2.4 <u>Unlimited Piggy Back Registration Rights</u>. For purposes of clarity, any registration effected pursuant to Section 2.2 hereof shall not be counted as a Demand Registration or UPO Demand Registration effected under Section 2.1 hereof.

2.3 <u>Registrations on Form F-3</u>. The holders of Registrable Securities may at any time and from time to time after the applicable Triggering Date, request in writing that Parent register the resale of any or all of such Registrable Securities on Form F-3 or any similar short-form registration which may be available at such time ("<u>Form F-3</u>"); provided, however, that Parent shall not be obligated to effect such request through an underwritten offering. Upon receipt of such written request, Parent shall promptly give written notice of the proposed registration to all other holders of Registrable Securities, and each Demanding Holder shall notify Parent of its wish to include all or a portion of such holder's Registrable Securities in such Form F-3 within ten (10) days after the receipt by the holder of the notice from Parent, and, as soon as practicable thereafter, but not more than twelve (12) days after Parent's initial receipt of such written request for a registration, Parent shall effect the registrable Securities or other securities of Parent, if any, of any other holder or holders joining in such request; provided, however, that Parent shall not be obligated to effect any such registration pursuant to this Section 2.3: (i) if Form F-3 is not available for such offering; or (ii) if the holders of the Registrable Securities, together with the holders of any other securities of Parent entitled to inclusion in such registration, propose to sell Registrable Securities and such other securities (if any) at any aggregate price to the public of less than \$500,000. Registrations effected pursuant to this Section 2.3 shall not be counted as Demand Registrations or UPO Demand Registration effected pursuant to Section 2.1.

## 3. REGISTRATION PROCEDURES.

3.1 <u>Filings; Information</u>. Whenever Parent is required to effect the registration of any Registrable Securities pursuant to Section 2, Parent shall use its best efforts to effect the registration and sale of such Registrable Securities in accordance with the intended method(s) of distribution thereof as expeditiously as practicable, and in connection with any such request:

3.1.1 <u>Filing Registration Statement</u>. Parent shall use its best efforts to, as expeditiously as possible and in any event within sixty (60) days after receipt of a request for a Demand Registration or UPO Demand Registration pursuant to Section 2.1, prepare and file with the Commission a Registration Statement on any form for which Parent then qualifies or which counsel for Parent shall deem appropriate and which form shall be available for the sale of all Registrable Securities to be registered thereunder in accordance with the intended method(s) of distribution thereof, and shall use its best efforts to cause such Registration Statement to become effective and use its best efforts to keep it effective for the period required by Section 3.1.3; provided, however, that Parent shall have the right to defer any Demand Registration or UPO Demand Registration for up to thirty (30) days, and any Piggy-Back Registration for such period as may be applicable to deferment of any demand registration to which such Piggy-Back Registration relates, in each case if Parent shall furnish to the holders a certificate signed by the Chairman of the Board of Directors or President of Parent stating that, in the good faith judgment of the Board of Directors of Parent, it would be materially detrimental to Parent and its shareholders for such Registration Statement to be effected at such time; provided further, however, that Parent shall not have the right to exercise the right set forth in the immediately preceding proviso more than once in any 365-day period in respect of a Demand Registration or UPO Demand Registration hereunder.

3.1.2 <u>Copies</u>. Parent shall, prior to filing a Registration Statement or prospectus, or any amendment or supplement thereto, furnish without charge to the holders of Registrable Securities included in such registration, and such holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the prospectus included in such Registration Statement (including each preliminary prospectus), and such other documents as the holders of Registrable Securities included in such registration or legal counsel for any such holders may request in order to facilitate the disposition of the Registrable Securities owned by such holders.

3.1.3 <u>Amendments and Supplements</u>. Parent shall prepare and file with the Commission such amendments, including post-effective amendments, and supplements to such Registration Statement and the prospectus used in connection therewith as may be necessary to keep such Registration Statement effective and in compliance with the provisions of the Securities Act until all Registrable Securities and other securities covered by such Registration Statement have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement or such securities have been withdrawn.

3.1.4 Notification. After the filing of a Registration Statement, Parent shall promptly, and in no event more than two (2) business days after such filing, notify the holders of Registrable Securities included in such Registration Statement of such filing, and shall further notify such holders promptly and confirm such advice in writing in all events within two (2) business days of the occurrence of any of the following: (i) when such Registration Statement becomes effective; (ii) when any post-effective amendment to such Registration Statement becomes effective; (iii) the issuance or threatened issuance by the Commission of any stop order (and Parent shall take all actions required to prevent the entry of such stop order or to remove it if entered); and (iv) any request by the Commission for any amendment or supplement to such Registration Statement or any prospectus relating thereto or for additional information or of the occurrence of an event requiring the preparation of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of the securities covered by such Registration Statement, such prospectus will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and promptly make available to the holders of Registrable Securities included in such Registration Statement any such supplement or amendment; except that before filing with the Commission a Registration Statement or prospectus or any amendment or supplement thereto, including documents incorporated by reference, Parent shall furnish to the holders of Registrable Securities included in such Registration Statement and to the legal counsel for any such holders, copies of all such documents and comment thereon, and Parent shall not file any Registration Statement or prospectus or amendment or supplement thereto, including documents incorporated by reference, to which such holders or their legal coun

3.1.5 Securities Laws Compliance. Parent shall use its best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities or securities exchanges, including the Nasdaq Capital Market, as may be necessary by virtue of the business and operations of Parent and do any and all other acts and things that may be necessary or advisable to enable the holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that Parent shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this paragraph or subject itself to taxation in any such jurisdiction.

3.1.6 <u>Agreements for Disposition</u>. Parent shall enter into customary agreements (including, if applicable, an underwriting agreement in customary form) and take such other actions as are reasonably required in order to expedite or facilitate the disposition of such Registrable Securities. No holder of Registrable Securities included in such registration statement shall be required to make any representations or warranties in the underwriting agreement except as reasonably requested by the Underwriters and, if applicable, with respect to such holder's organization, good standing, authority, title to Registrable Securities, lack of conflict of such sale with such holder's material agreements and organizational documents, and with respect to written information relating to such holder that such holder has furnished in writing expressly for inclusion in such Registration Statement.

3.1.7 <u>Cooperation</u>. The principal executive officer of Parent, the principal financial officer of Parent, the principal accounting officer of Parent and all other officers and members of the management of Parent shall cooperate fully in any offering of Registrable Securities hereunder, which cooperation shall include, without limitation, the preparation of the Registration Statement with respect to such offering and all other offering materials and related documents, and participation in meetings with Underwriters, attorneys, accountants and potential investors.

3.1.8 <u>Records</u>. Parent shall make available for inspection by the holders of Registrable Securities included in such Registration Statement, any Underwriter participating in any disposition pursuant to such registration statement and any attorney, accountant or other professional retained by any holder of Registrable Securities included in such Registration Statement or any Underwriter, all financial and other records, pertinent corporate documents and properties of Parent, as shall be necessary to enable them to exercise their due diligence responsibility, and cause Parent's officers, directors and employees to supply all information requested by any of them in connection with such Registration Statement.

3.1.9 <u>Opinions and Comfort Letters</u>. Parent shall furnish to each holder of Registrable Securities included in any Registration Statement a signed counterpart, addressed to such holder, of (i) any opinion of counsel to Parent delivered to any Underwriter and (ii) any comfort letter from Parent's independent public accountants delivered to any Underwriter. In the event no legal opinion is delivered to any Underwriter, Parent shall furnish to each holder of Registrable Securities included in such Registration Statement, at any time that such holder elects to use a prospectus, an opinion of counsel to Parent to the effect that the Registration Statement containing such prospectus has been declared effective and that no stop order is in effect.

3.1.10 <u>Earnings Statement</u>. Parent shall comply with all applicable rules and regulations of the Commission and the Securities Act, and make available to its shareholders, as soon as reasonably practicable, an earnings statement covering a period of twelve (12) months, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder.

3.1.11 Listing. Parent shall use its best efforts to cause all Registrable Securities included in any registration to be listed on such exchanges or otherwise designated for trading in the same manner as similar securities issued by Parent are then listed or designated or, if no such similar securities are then listed or designated, in a manner satisfactory to the holders of a majority of the Registrable Securities included in such registration, on an As-Converted Basis.

3.1.12. <u>Transfer Agent</u>. Parent shall provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of the registration statement.

3.1.13. <u>Misstatements</u>. Parent shall notify the Investors at any time when a prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the prospectus included in such Registration Statement, as then in effect, includes an untrue statement of a material fact or an omission to state a material fact required to be stated in a registration statement or prospectus, or necessary to make the statements therein in the light of the circumstances under which they were made not misleading (a "<u>Misstatement</u>"), and then to correct such Misstatement.

3.2 <u>Obligation to Suspend Distribution</u>. Upon receipt of any notice from Parent of the happening of any event of the kind described in Section 3.1.4(iv), or, in the case of a resale registration on Form F-3 pursuant to Section 2.3 hereof, upon any suspension by Parent, pursuant to a written insider trading compliance program adopted by Parent's Board of Directors, of the ability of all "insiders" covered by such program to transact in Parent's securities because of the existence of material non-public information, each holder of Registrable Securities included in any registration shall immediately discontinue disposition of such Registrable Securities pursuant to the Registration Statement covering such Registrable Securities until such holder receives the supplemented or amended prospectus contemplated by Section 3.1.4(iv) or the restriction on the ability of "insiders" to transact in Parent's securities is removed, as applicable, and, if so directed by Parent, each such holder will deliver to Parent all written copies, other than permanent file copies then in such holder's possession, of the most recent prospectus covering such Registrable Securities at the time of receipt of such notice.

3.3 <u>Registration Expenses</u>. Parent shall bear all costs and expenses incurred in connection with any Demand Registration or UPO Demand Registration pursuant to Section 2.1, any Piggy-Back Registration pursuant to Section 2.2, and any registration on Form F-3 effected pursuant to Section 2.3, and all expenses incurred in performing or complying with its other obligations under this Amended and Restated Registration Rights Agreement, whether or not the Registration Statement becomes effective, including, without limitation: (i) all registration and filing fees and fees of any securities exchange on which the Parent Class A Ordinary Shares, HL Parent Warrants and/or FF Parent Warrants are then listed; (ii) fees and expenses of compliance with securities or "blue sky" laws (including fees and disbursements of counsel of the Underwriters in connection with blue sky qualifications of the Registrable Securities); (iii) printing expenses; (iv) Parent's internal expenses (including, without limitation, all salaries and expenses of is officers and employees); (v) the fees and expenses or costs associated with the delivery of any opinions or comfort letters requested pursuant to Section 3.1.11; (vi) Financial Industry Regulatory Authority fees; (vii) fees and disbursements of counsel for Parent and fees and expenses for independent certified public accountants retained by Parent (including the expenses or costs associated with the delivery of any opinions or comfort letters requested pursuant to Section 3.1.9); (viii) the fees and expenses of a majority-in-interest of the Registrable Securities being sold by the holders thereof, which underwriting discounts or selling commissions attributable to the Registrable Securities being sold by the holders thereof, which underwriting discounts or selling commissions attributable to the Registrable Securities being sold by the holders and Parent shall hear the expenses of the Underwriter pro rata in proportion to the respective amount of shares each is selling in such offering.

3.4 <u>Information</u>. The holders of Registrable Securities shall provide such information as may reasonably be requested by Parent, or the managing Underwriter, if any, in connection with the preparation of any Registration Statement, including amendments and supplements thereto, in order to effect the registration of any Registrable Securities under the Securities Act pursuant to Section 2 and in connection with Parent's obligation to comply with federal and applicable state securities laws.

3.5 <u>Requirements for Participation in Underwritten Offerings</u>. No person may participate in any underwritten offering for equity securities of Parent pursuant to a registration initiated by Parent hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by Parent and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.

3.6 Suspension of Sales; Adverse Disclosure. Upon receipt of written notice from Parent that a registration statement or prospectus contains a Misstatement, each of the holders of Registrable Securities shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended prospectus correcting the Misstatement (it being understood that Parent hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until it is advised in writing by Parent that the use of the prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any registration at any time would require Parent to make an Adverse Disclosure (as defined below) or would require the inclusion in such Registration Statement of financial statements that are unavailable to Parent for reasons beyond Parent's control, Parent may, upon giving prompt written notice of such action to the holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than thirty (30) days, determined in good faith by Parent to be necessary for such purpose. In the event Parent exercises its rights under the preceding sentence, the Investors agree to suspend, immediately upon their receipt of the notice referred to above, their use of the prospectus relating to any registration in connection with any sale or offer to sell Registrable Securities. Parent shall immediately notify the Investors of the expiration of any period during which it exercised its rights under this Section 3.6. As used herein, "Adverse Disclosure" means any public disclosure of material non-public information, which disclosure, in the good faith judgment of the principal executive officer or principal financial officer of Parent, after consultation with coursel to Parent, (i) would be required to be made in any Registration Statement or prospectus in order for the applicable Registration Statement or prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) Parent has a bona fide business purpose for not making such information public.

3.7 <u>Reporting Obligations</u>. As long as any Investor shall own Registrable Securities, Parent, at all times while it shall be reporting under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by Parent after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Investors with true and complete copies of all such filings. Parent will be deemed to have furnished the Investors with copies of such filings upon the appearance of such filings on EDGAR and Parent's website. Parent further covenants that it shall take such further action as any Investor may reasonably request, all to the extent required from time to time to enable such holder to sell Registrable Securities held by such Investor without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act, including providing any legal opinions. Upon the request of any Investor, Parent shall deliver to such Investor a written certification of a duly authorized officer as to whether it has complied with such requirements.

### 4. INDEMNIFICATION AND CONTRIBUTION.

4.1 Indemnification by Parent. Subject to the Irish Companies Act of 2014, in particular and without limitation, Section 82 thereunder, Parent agrees to indemnify and hold harmless each Investor and each other holder of Registrable Securities, and each of their respective officers, employees, affiliates, directors, partners, members, attorneys and agents, and each person, if any, who controls an Investor and each other holder of Registrable Securities (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) (each, an "Investor Indemnified Party"), from and against any expenses, losses, judgments, claims, damages or liabilities, whether joint or several, arising out of or based upon any untrue statement (or allegedly untrue statement) of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to such Registration Statement, or arising out of or based upon any omission (or alleged omission) to state a material fact required to be stated therein or necessary to make the statements therein not misleading, or any violation by Parent of the Securities Act or any rule or regulation promulgated thereunder applicable to Parent and relating to action or inaction required of Parent in connection with any such registration; and Parent shall promptly reimburse the Investor Indemnified Party for any legal and any other expenses reasonably incurred by such Investor Indemnified Party in connection with investigating and defending any such expense, loss, judgment, claim, damage, liability or action whether or not any such person is a party to any such claim or action and including any and all legal and other expenses incurred in giving testimony or furnishing documents in response to a subpoena or otherwise; provided, however, that Parent will not be liable in any such case to the extent that any such expense, loss, claim, damage or liability arises out of or is based upon any untrue statement or allegedly untrue statement or omission or alleged omission made in such Registration Statement, preliminary prospectus, final prospectus, or summary prospectus, or any such amendment or supplement, in reliance upon and in conformity with information furnished to Parent, in writing, by such selling holder expressly for use therein. Parent also shall indemnify any Underwriter of the Registrable Securities, their officers, affiliates, directors, partners, members and agents and each person who controls such Underwriter on substantially the same basis as that of the indemnification provided above in this Section 4.1.

4.2 Indemnification by Holders of Registrable Securities. Subject to the limitations set forth under Section 4.4.3 hereof, each selling holder of Registrable Securities will, in the event that any registration is being effected under the Securities Act pursuant to this Amended and Restated Registration Rights Agreement of any Registrable Securities held by such selling holder, indemnify and hold harmless Parent, each of its directors and officers and each Underwriter (if any), and each other selling holder and each other person, if any, who controls another selling holder or such Underwriter within the meaning of the Securities Act, against any losses, claims, judgments, damages or liabilities, whether joint or several, insofar as such losses, claims, judgments, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement or allegedly untrue statement of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to the Registration Statement, or arise out of or are based upon any omission or the alleged omission to state a material fact required to be stated therein or necessary to make the statement therein not misleading, if the statement or omission was made in reliance upon and in conformity with information furnished in writing to Parent by such selling holder expressly for use therein, and shall reimburse Parent, its directors and officers, and each other selling holder or controlling person for any legal or other expenses reasonably incurred by any of them in connection with investigation or defending any such loss, claim, damage, liability or action. Each selling holder's indemnification obligations hereunder shall be several and not joint and shall be limited to the amount of any net proceeds actually received by such selling holder. Each selling holder of Registrable Securities shall indemnify any Underwriter of the Registrable Securities, the Underwriter's officers, affiliates, directors, partners, members, and agents, and each person who controls such Underwriter to the same extent as provided herein with respect to Parent.

4.3 Conduct of Indemnification Proceedings. Promptly after receipt by any person of any notice of any loss, claim, damage or liability or any action in respect of which indemnity may be sought pursuant to Section 4.1 or 4.2, such person (the "Indemnified Party") shall, if a claim in respect thereof is to be made against any other person for indemnification hereunder, notify such other person (the "Indemnifying Party") in writing of the loss, claim, judgment, damage, liability or action; provided, however, that the failure by the Indemnified Party to notify the Indemnifying Party shall not relieve the Indemnifying Party from any liability which the Indemnifying Party may have to such Indemnified Party hereunder, except and solely to the extent the Indemnifying Party is actually prejudiced by such failure. If the Indemnified Party is seeking indemnification with respect to any claim or action brought against the Indemnified Party, then the Indemnifying Party shall be entitled to participate in such claim or action, and, to the extent that it wishes, jointly with all other Indemnifying Parties, to assume control of the defense thereof with counsel satisfactory to the Indemnified Party. After notice from the Indemnifying Party to the Indemnifying Party of its election to assume control of the defense of such claim or action, the Indemnifying Party shall not be liable to the Indemnified Party for any legal or other expenses subsequently incurred by the Indemnified Party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that in any action in which both the Indemnified Party and the Indemnifying Party are named as defendants, the Indemnified Party shall have the right to employ separate counsel (but no more than one such separate counsel) to represent the Indemnified Party and its controlling persons who may be subject to liability arising out of any claim in respect of which indemnity may be sought by the Indemnified Party against the Indemnifying Party, with the fees and expenses of such counsel to be paid by such Indemnifying Party if, based upon the written opinion of counsel of such Indemnified Party, representation of both parties by the same counsel would be inappropriate due to actual or potential differing interests between them. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, consent to entry of judgment or effect any settlement of any claim or pending or threatened proceeding in respect of which the Indemnified Party is or could have been a party and indemnity could have been sought hereunder by such Indemnified Party, unless such judgment or settlement includes an unconditional release of such Indemnified Party from all liability arising out of such claim or proceeding.

### 4.4 Contribution.

4.4.1 If the indemnification provided for in the foregoing Sections 4.1, 4.2 and 4.3 is unavailable to any Indemnified Party in respect of any loss, claim, damage, liability or action referred to herein, then each such Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such loss, claim, damage, liability or action in such proportion as is appropriate to reflect the relative fault of the Indemnified Parties and the Indemnifying Parties in connection with the actions or omissions which resulted in such loss, claim, damage, liability or action, as well as any other relevant equitable considerations. The relative fault of any Indemnified Party and any Indemnifying Party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by such Indemnified Party or such Indemnifying Party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.

4.4.2 The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 4.4 were determined by pro rata allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the immediately preceding Section 4.4.1.

4.4.3 The amount paid or payable by an Indemnified Party as a result of any loss, claim, damage, liability or action referred to in the immediately preceding paragraph shall be deemed to include, subject to the limitations set forth above, any legal or other expenses incurred by such Indemnified Party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this Section 4.4, no holder of Registrable Securities shall be required to contribute any amount in excess of the dollar amount of the net proceeds (after payment of any underwriting fees, discounts, commissions or taxes) actually received by such holder from the sale of Registrable Securities which gave rise to such contribution obligation. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) with respect to any action shall be entitled to contribution in such action from any person who was not guilty of such fraudulent misrepresentation.

4.5 <u>Survival</u>. The indemnification provided for under this Amended and Restated Registration Rights Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the Indemnified Party or any officer, director or controlling person of such Indemnified Party and shall survive the transfer of securities.

5. <u>RULE 144</u>. Parent covenants that it shall file any reports required to be filed by it under the Securities Act and the Exchange Act and shall take such further action as the holders of Registrable Securities may reasonably request, all to the extent required from time to time to enable such holders to sell Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 under the Securities Act, as such Rules may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission.

#### 6. MISCELLANEOUS

6.1 <u>Other Registration Rights</u>. Except for the registration rights granted to the PIPE investors under each of their respective subscription agreements, dated on or around August 25, 2020, Parent represents and warrants that no person other than the holders of the Registrable Securities has any right to require Parent to register any shares of Parent's capital stock for sale or to include shares of Parent's capital stock in any registration filed by Parent for the sale of shares of capital stock for its own account or for the account of any other person. Further, except with respect to the registration rights granted to the PIPE Investors, Parent represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Amended and Restated Registration Rights Agreement, the terms of this Amended and Restated Registration Rights Agreement shall prevail.

6.2 <u>Assignment; No Third Party Beneficiaries</u>. This Amended and Restated Registration Rights Agreement and the rights, duties and obligations of Parent hereunder may not be assigned or delegated by Parent in whole or in part. This Amended and Restated Registration Rights Agreement and the rights, duties and obligations of the holders of Registrable Securities hereunder may be freely assigned or delegated by such holder of Registrable Securities in conjunction with and to the extent of any transfer of Registrable Securities by any such holder. This Amended and Restated Registration Rights Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties, to the permitted assigns of the Investors or holder of Registrable Securities or of any assignee of the Investors or holder of Registrable Securities. This Amended and Restated Registration Rights Agreement is not intended to confer any rights or benefits on any persons that are not party hereto other than as expressly set forth in Article 4 and this Section 6.2. No assignment by any party hereto of such party's rights, duties, and obligations hereunder shall be binding upon or obligate Parent unless and until Parent shall have received (i) written notice of such assignment and (ii) the written assignment of the assignee, in a form reasonably satisfactory to Parent, to be bound by the terms and provisions of this Amended and Restated Registration Rights Agreement (which may be accomplished by an addendum or certificate of joinder to this Amended and Restated Registration Rights Agreement).

6.3 Notices. All notices, demands, requests, consents, approvals or other communications (collectively, "Notices") required or permitted to be given hereunder or which are given with respect to this Amended and Restated Registration Rights Agreement shall be in writing and shall be personally served, delivered by reputable air courier service with charges prepaid, or transmitted by hand delivery, telegram, telex or facsimile, addressed as set forth below, or to such other address as such party shall have specified most recently by written notice. Notice shall be deemed given on the date of service or transmission if personally served or transmitted by telegram, telex or facsimile; provided, that if such service or transmission is not on a business day or is after normal business hours, then such notice shall be deemed given on the next business day. Notice otherwise sent as provided herein shall be deemed given.

To Parent, to:

Fusion Fuel Green PLC 10 Earlsfort Terrace Dublin 2, D02 T380, Ireland Attention: Frederico Figueira de Chaves, Chief Financial Officer Email: frederico@keyfh.com with a copy, which shall not constitute notice, to:

Arthur Cox 10 Earlsfort Terrace Dublin 2, D02 T380, Ireland Attn: Connor Manning, Esq. Email: connor.manning@arthurcox.com

and

Feinberg Hanson LLP 855 Boylston Street, 8th Floor Boston, MA 02116 Attn: David H. Feinberg, Esq. Email: dfeinberg@feinberghanson.com

If to the Fusion Fuel Shareholders:

Fusion Welcome, S.A. Ex-Siemens Facilities Rua da Fabrica, S/N, Sabugo 2715-376, Almargem do Bispo Portugal Attn: Frederico Figueira de Chaves Email: frederico@keyfh.com

with a copy, which shall not constitute notice, to:

Feinberg Hanson LLP 855 Boylston Street, 8th Floor Boston, MA 02116 Attn: David H. Feinberg, Esq. Email: dfeinberg@feinberghanson.com

If to the HL Investors, to:

Jeffrey Schwarz c/o Metropolitan Capital Advisors, Inc. 499 Park Avenue, 12th Floor New York, NY 10022 Email: jschwarz@metrocap.com

with a copy, which shall not constitute notice, to:

Graubard Miller The Chrysler Building 405 Lexington Ave, 11th Floor New York, NY 10174 Attn: David Alan Miller, Esq. / Jeffrey M. Gallant, Esq. Email: dmiller@graubard.com / jgallant@graubard.com

EarlyBirdCapital, Inc. 366 Madison Avenue New York, NY 10017 Attn: Steven Levine, CEO Email: slevine@ebcap.com

with a copy, which shall not constitute notice, to:

Graubard Miller The Chrysler Building 405 Lexington Ave, 11th Floor New York, NY 10174 Attn: David Alan Miller, Esq. / Jeffrey M. Gallant, Esq.

6.4 <u>Severability</u>. This Amended and Restated Registration Rights Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Amended and Restated Registration Rights Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Amended and Restated Registration Rights Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible that is valid and enforceable.

6.5 <u>Counterparts</u>. This Amended and Restated Registration Rights Agreement may be executed in multiple counterparts, each of which shall be deemed an original, and all of which taken together shall constitute one and the same instrument. Delivery of a signed counterpart of this Amended and Restated Registration Rights Agreement by facsimile or email/pdf transmission shall constitute valid and sufficient delivery thereof.

6.6 Entire Agreement. This Amended and Restated Registration Rights Agreement (including all agreements entered into pursuant hereto and all certificates and instruments delivered pursuant hereto and thereto) constitute the entire agreement of the parties with respect to the subject matter hereof and supersede all prior and contemporaneous agreements, representations, understandings, negotiations and discussions between the parties, whether oral or written.

6.7 Modifications and Amendments. Upon the written consent of Parent and the holders of at least sixty-six and two-thirds percent (66-2/3%) of the Registrable Securities, on an As-Converted Basis, at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Amended and Restated Registration Rights Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one holder of Registrable Securities or group of holders of Registrable Securities, solely in its or their capacity(ies) as a holder(s) of Registrable Securities, in a manner that is materially different from the other holders of Registrable Securities (in such capacity) shall require the consent of the holder(s) so affected; for the sake of clarity and by way of example, any such amendment or waiver that adversely affects the Fusion Fuel Shareholders in a manner that is materially different from the other holders of Registrable Securities shall require the consent of the holders of a majority in interest of the Fusion Fuel Securities, on an As-Converted Basis. No course of dealing between any holders of Registrable Securities or Parent in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any other rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.

6.8 <u>Titles and Headings</u>. Titles and headings of sections of this Amended and Restated Registration Rights Agreement are for convenience only and shall not affect the construction of any provision of this Amended and Restated Registration Rights Agreement.

6.9 Waivers and Extensions. Any party to this Amended and Restated Registration Rights Agreement may waive any right, breach or default which such party has the right to waive, provided that such waiver will not be effective against the waiving party unless it is in writing, is signed by such party, and specifically refers to this Amended and Restated Registration Rights Agreement. Waivers may be made in advance or after the right waived has arisen or the breach or default waived has occurred. Any waiver may be conditional. No waiver of any breach of any agreement or provision herein contained shall be deemed a waiver of any preceding or succeeding breach thereof nor of any other agreement or provision herein contained. No waiver or extension of time for performance of any obligations or acts shall be deemed a waiver or extension of the time for performance of any other obligations or acts.

6.10 <u>Remedies Cumulative</u>. In the event that Parent fails to observe or perform any covenant or agreement to be observed or performed under this Amended and Restated Registration Rights Agreement, the Investor or any other holder of Registrable Securities may proceed to protect and enforce its rights by suit in equity or action at law, whether for specific performance of any term contained in this Amended and Restated Registration Rights Agreement or for an injunction against the breach of any such term or in aid of the exercise of any power granted in this Amended and Restated Registration Rights Agreement or to enforce any other legal or equitable right, or to take any one or more of such actions, without being required to post a bond. None of the rights, powers or remedies conferred under this Amended and Restated Registration Rights Agreement shall be mutually exclusive, and each such right, power or remedy shall be cumulative and in addition to any other right, power or remedy, whether conferred by this Amended and Restated Registration Rights Agreement or now or hereafter available at law, in equity, by statute or otherwise.

6.11 <u>Governing Law</u>. This Amended and Restated Registration Rights Agreement shall be governed by, interpreted under, and construed in accordance with the internal laws of the State of New York applicable to agreements made and to be performed within the State of New York, without giving effect to any choice-of-law provisions thereof that would compel the application of the substantive laws of any other jurisdiction. Parent and each Investor irrevocably submits to the nonexclusive jurisdiction of any New York State or United States Federal court sitting in The City of New York, Borough of Manhattan, over any suit, action or proceeding arising out of or relating to this Amended and Restated Registration Rights Agreement. Parent and each Investor irrevocably waives, to the fullest extent permitted by law, any objection that he, she, or it may now or hereafter have to the laying of venue of any such suit, action or proceeding brought in such a court and any claim that any such suit, action or proceeding brought in such a court and any claim that any such suit, action or proceeding brought in such a court and any claim that any such suit, action or proceeding brought in such a court and any claim that any such suit, action or proceeding brought in such a court has been brought in an inconvenient forum.

6.12 <u>Waiver of Trial by Jury</u>. EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES THE RIGHT TO A TRIAL BY JURY IN ANY ACTION, SUIT, COUNTERCLAIM OR OTHER PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF, CONNECTED WITH OR RELATING TO THIS AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREBY, OR THE ACTIONS OF THE INVESTOR IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties have caused this Amended and Restated Registration Rights Agreement to be executed and delivered by their duly authorized representatives as of the date first written above.

### HL

HL ACQUISITIONS CORP.

By: /s/ Jeffrey E. Schwarz Name: Jeffrey E. Schwarz Title: Chief Executive Officer

PARENT

### FUSION FUEL GREEN PLC

By: /s/ Frederico Figueira de Chaves Name: Frederico Figueira de Chaves Title: Chief Executive Officer

#### FUSION FUEL SHAREHOLDERS

#### FUSION WELCOME, S.A.

By: /s/ João Teixeira Wahnon Name: João Teixeira Wahnon Title: Director

### FALCFIVE, LDA

By: /s/ Vicente Falcão e Cunha Name: Vicente Falcão e Cunha Title: Director

### NUMBERBUBBLE, S.A.

By: /s/ João Teixeira Wahnon Name: João Teixeira Wahnon Title: Director

#### MAGNO EFEITO, S.A.

#### By: /s/ Jaime Silva

Name: Jaime Silva Title: Director

KEY FAMILY HOLDING INVESTIMENTOS E CONSULTORIA DE GESTAO, LDA

By: /s/ Frederico Figueira de Chaves Name: Frederico Figueira de Chaves Title: Director

## HL AFFILIATES:

/s/ Jeffrey E. Schwarz Jeffrey E. Schwarz

HL INVESTORS

/s/ Wendy Schwarz Wendy Schwarz

/s/ Joel Greenblatt Joel Greenblatt

/s/ Jonathan Guss Jonathan Guss

/s/ Stephanie Guss Stephanie Guss

/s/ Greg Drechsler Greg Drechsler

## LUNDE3 HOLDING AS

By: /s/ Rune Magnus Lundetrae Name: Rune Magnus Lundetrae Title: Director

/s/ Ajay Khandelwal

Ajay Khandelwal

## /s/ Karen Finerman

Karen Finerman

/s/ Craig Effron

Craig Effron

/s/ Curtis Schenker Curtis Schenker

/s/ Benjamin Schwarz Benjamin Schwarz

## JEFFREY SCHWARZ CHILDREN'S TRUST

By: /s/ Craig Frank Name: Craig Frank Title: Trustee

## STERN YOI LIMITED PARTNERSHIP

By: /s/ Yoav Stern Name: Yoav Stern Title: Partner

#### Alla Jezmir

### EBC DESIGNEES:

EARLYBIRDCAPITAL, INC.

By: /s/ Michael Powell Name: Michael Powell Title: Managing Director

/s/ Michael Powell Michael Powell

/s/ Edward Kovary Edward Kovary

/s/ Marc Van Tricht Marc Van Tricht

/s/ Gregory Stoupnitzky

Gregory Stoupnitzky

/s/ Robert Gladstone

Robert Gladstone

/s/ Mauro Conijeski Mauro Conijeski

/s/ Eileen Moore Eileen Moore

/s/ Gleeson Cox

Gleeson Cox

/s/ Jillian Carter Jillian Carter

Amy Kaufmann Amy Kaufmann

I-BANKERS SECURITIES, INC.

By: /s/ Mike McCrory Name: Mike McCrory Title: Chief Executive Officer

### DIRECTORS:

/s/ António Augusto Gutierrez Sá da Costa António Augusto Gutierrez Sá da Costa

/s/ Rune Magnus Lundetrae Rune Magnus Lundetrae

/s/ Alla Jezmir Alla Jezmir

#### **INDEMNIFICATION ESCROW AGREEMENT**

This **INDEMNIFICATION ESCROW AGREEMENT** (this "<u>Agreement</u>") is made and entered into as of December 10, 2020 by and among Fusion Fuel Green PLC (formerly known as Fusion Fuel Green Limited and Dolya Holdco 3 Limited) ("<u>Parent</u>"), Fusion Welcome – Fuel, S.A. (the "<u>Company</u>"), Fusion Welcome, S.A. ("<u>Company Shareholder Representative</u>") as the representative of the Company Shareholders (as defined in the Business Combination Agreement), HL Acquisitions Corp. ("<u>HL</u>"), Jeffrey Schwarz ("<u>HL Representative</u>") as the representative of the former shareholders of HL, and Continental Stock Transfer & Trust Company (the "<u>Escrow Agent</u>"). Parent, the Company Shareholder Representative, HL, and the HL Representative are collectively referred to in this Agreement as the "<u>Escrow Parties</u>". The Escrow Parties and the Escrow Agent are collectively referred to in this Agreement as the "<u>Parties</u>" and each individually as a "<u>Party</u>".

#### RECITALS

A. WHEREAS, Parent, HL, the Company, the Company Shareholders, and the other parties thereto have entered into a Business Combination Agreement dated as of June 6, 2020, as amended and restated on August 25, 2020 (as may be further amended from time to time, the "Business Combination Agreement"), pursuant to which (i) Fusion Fuel Atlantic Limited, a British Virgin Islands business company and wholly owned subsidiary of Parent, will merge with and into HL with HL being the surviving entity of such merger (the "Merger") and becoming a wholly-owned subsidiary of Parent, followed immediately by (ii) the acquisition by Parent of all of the issued and outstanding shares of the Company (the "Share Exchange", and together with the Merger, the "Transactions").

**B.** WHEREAS, upon consummation of the Share Exchange, each Company Shareholder will receive its pro rata portion of an aggregate of 2,125,000 Class B ordinary shares of Parent ("Parent Class B Ordinary Shares") and warrants to purchase an aggregate of 2,125,000 Class A ordinary shares of Parent ("Parent Class A Ordinary Shares"), and the Company Shareholders will be allotted the right to receive a portion of up to an aggregate of an additional 1,137,000 Parent Class A Ordinary Shares and warrants to purchase up to an aggregate of 1,137,000 Parent Class A Ordinary Shares upon the satisfaction of certain earnout targets.

C. WHEREAS, the Business Combination Agreement contemplates the execution and delivery of this Agreement and the deposit by Parent with the Escrow Agent of 212,500 Parent Class B Ordinary Shares (the "Escrow Shares"), allocated among the Company Shareholders in the same proportions as the total Company Consideration is allocated among them, to satisfy any Losses that are recoverable against the Company Shareholders pursuant to and in accordance with the Business Combination Agreement.

D. WHEREAS, capitalized terms not defined herein shall have the meaning ascribed to such terms in the Business Combination Agreement.

#### AGREEMENT

NOW, THEREFORE, in consideration of the premises and the mutual promises herein made, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Escrow Parties and Escrow Agent, intending to be legally bound, agree as follows:

#### Section 1. Escrow.

1.1 <u>Appointment; Shares Placed in Escrow</u>. The Escrow Parties hereby appoint the Escrow Agent as their escrow agent for the purposes set forth herein, and the Escrow Agent hereby accepts such appointment and agrees to act as escrow agent in accordance with the terms and conditions set forth herein. Simultaneously with the execution and delivery of this Agreement, Parent shall instruct the Escrow Agent to include an escrow legend and stop transfer order against the book entry positions representing the Escrow Shares, and to issue such restricted book entry positions in the name of the Company Shareholders and in the amounts as set forth on <u>Exhibit A</u> hereto, which, for the avoidance of doubt, have been calculated in the same proportions as the Parent Class B Ordinary Shares are allocated among the Company Shareholders, together with an assignment separate from each such certificate executed in blank by each such Company Shareholder, with medallion signature guarantees.

1.2 Escrow Account. The Escrow Agent will issue its written confirmation of the receipt of the Escrow Shares and, upon delivery, shall hold the Escrow Shares in an account established with the Escrow Agent, with a separate account for each Company Shareholder's portion of the Escrow Shares, subject to the terms of Section 3 below (collectively, the "Escrow Account").

**1.3** <u>Trust Fund</u>. The Escrow Shares shall be held, by way of security, on bare trust and shall not be subject to any lien, attachment, trustee process or any other judicial process of any creditor of any Escrow Party or any of its respective affiliates. The Escrow Agent shall hold and safeguard the Escrow Shares until the Termination Date (as defined in Section 5) or earlier release in accordance with this Agreement.

1.4 <u>Rights as Shareholders</u>. Except as otherwise provided in this Agreement, the Company Shareholders shall retain all of their rights as shareholders of Parent with respect to the Escrow Shares while such remain in the Escrow Account, including, without limitation, the right of the Company Shareholders to vote their Parent Class B Ordinary Shares included in the Escrow Shares.

1.5 <u>Dividends</u>. All dividends payable in cash with respect to the Escrow Shares held in the Escrow Account shall be paid to the Company Shareholders in the same proportions as the Parent Class B Ordinary Shares are allocated among them, but all dividends payable in stock or other non-cash property ("<u>Non-Cash Dividends</u>") shall be delivered to the Escrow Agent to hold in accordance with the terms hereof. As used herein, the term "<u>Escrow</u> Fund" shall be deemed to include the Non-Cash Dividends distributed thereon, if any.

1.6 <u>Transfers</u>. The Company Shareholder Representative acknowledges on behalf of the Company Shareholders that the Escrow Shares are subject to lock-up provisions set forth in the Business Combination Agreement. In the event a Company Shareholder makes a transfer prior to the end of the lock-up period as set forth in the Business Combination Agreement with respect to his, her, or its Escrow Shares, the Company Shareholder Representative shall deliver to the Escrow Agent, on behalf of the transferring Company Shareholder, (i) an assignment separate from certificate executed by the transferring Company Shareholder, with medallion signature guaranty, evidencing the transfer of the Escrow Shares, (ii) an assignment separate from certificate executed in blank by the transferree with medallion signature guaranty, with respect to the transferred Escrow Shares and (iii) a written acknowledgement of the transferree that the transferred Escrow Shares are subject to this Agreement and, if applicable, the transfer has been duly stamped by the Irish Revenue Commissioners. Upon receipt of such documents, the Escrow Agent shall deliver to Parent's transfer agent the original certificate(s) representing the Escrow Shares which have been transferred, together with the executed assignment separate from certificate executed by the transferring Company Shareholder, and shall request that Parent issue new certificates representing the number of Escrow Shares, if any, that continue to be owned by the transferring Company Shareholder and the number of Escrow Shares and by the transferring Company Shareholder and the number of Escrow Shares are shall cooperate in all respects with the Escrow Agent in documenting each such transfer and in effectuating the result intended to be accomplished thereby.

#### Section 2. Release of Escrow.

The Escrow Shares held pursuant to this Agreement are intended to provide an exclusive remedy in respect of indemnification claims brought pursuant to the Business Combination Agreement. The Parties shall act in accordance with, and the Escrow Agent shall hold as bare trustee, and release the Escrow Shares, as provided in, this <u>Section 2</u>.

### 2.1 Indemnification Related Claims.

(a) At any time and from time to time on or prior to the Escrow Termination Date, the HL Representative, acting on behalf of any HL Indemnitee, may make a claim for indemnification pursuant to and in accordance with Article XI of the Business Combination Agreement (a "<u>Claim</u>") by delivering to the Escrow Agent and Parent a Notice of Claim setting forth (i) the information required to be included in a Notice of Claim pursuant to the Business Combination Agreement and (ii) the estimated number of Escrow Shares to be forfeited and cancelled in satisfaction of the Claim.

(b) If the Escrow Agent has not received a written objection to such Claim or portion thereof or the amount of such Claim from Parent within twenty (20) calendar days following the Escrow Agent's and Parent's receipt of the Notice of Claim, the HL Representative shall promptly calculate the Fair Market Value (as defined below) of the Escrow Shares to be reviewed and confirmed by Parent, and Parent and the HL Representative shall deliver a joint instruction ("Forfeiture Instruction") to the Escrow Agent to release for forfeiture and cancellation such number of Escrow Shares equal to the *quotient* of (i) the amount of Loss as set forth in the Notice of Claim *divided by* (ii) the Fair Market Value. Upon the Escrow Agent's receipt of the Forfeiture Instruction, the Escrow Agent shall release for forfeiture and cancellation the number of Escrow Shares as set forth in the Forfeiture Instruction from the accounts maintained on behalf of each Company Shareholder in the same proportion that the total Company Consideration is allocated among them. In no event shall the Escrow Agent be required to calculate Fair Market Value or make a determination of the number of Escrow Shares to be forfeited and cancelled in satisfaction of any Claim in the aggregate or from the account maintained on behalf of any particular Company Shareholder. The Escrow Parties agree that the foregoing right to satisfy Claims by the cancellation and forfeiture of any Escrow Shares may be made notwithstanding any other agreements restricting or limiting the ability of any Company Shareholder to sell or transfer any Escrow Shares of Parent or otherwise.

(c) If Parent in good faith delivers to the Escrow Agent and HL Representative a written objection (a "Dispute Notice") to any Claim or portion thereof or to the amount of such Claim within twenty (20) calendar days following both the Escrow Agent's and Parent's receipt of the Notice of Claim, then the Escrow Agent shall not release any of the Escrow Shares that are the subject of the Dispute Notice until the Escrow Agent receives either (i) a Forfeiture Instruction signed by Parent and the HL Representative authorizing the release of such number of Escrow Shares for cancellation and forfeiture in satisfaction of the Claim or (ii) a Final Determination directing the release for cancellation and forfeiture of such number of Escrow Shares as set forth in the Final Determination or, if the Final Determination determines the dollar value of the Claim but does not calculate the number of Escrow Shares to be forfeited and cancelled, in a Forfeiture Instruction delivered by the HL Representative and Parent together with the Final Determination. Notwithstanding the foregoing, if Parent objects only in part to the amount of the Claim, then after the lapse of the aforementioned twenty (20) calendar day period the HL Representative and Parent shall submit a Forfeiture Instruction to the Escrow Agent setting forth the portion of the Claim not objected to by Parent, pursuant to Section 2.1(b). Upon receipt of a Final Determination and/or Forfeiture Instruction, as the case may be, the Escrow Agent shall release such number of Escrow Shares in the Escrow Account in accordance with such Final Determination and/or Forfeiture Instruction.

(d) Notwithstanding anything to the contrary herein, no Claim shall be payable unless and until the aggregate amount of indemnifiable Losses exceeds €750,000 ("<u>Threshold</u>"), in which case all Losses that exceed the Threshold shall be indemnifiable.

#### 2.2 Release of Remaining Escrow.

(a) On the tenth (10th) Business Day after Parent files its annual report for the year ending December 31, 2021 ("Escrow Termination Date"), the Escrow Agent shall release the Escrow Shares remaining in the Escrow Account, less such number of Escrow Shares representing the value of any Unresolved Claims (defined below), to the Company Shareholders in accordance with Joint Release Instructions specifying the number of Escrow Shares to be distributed to each Company Shareholder (or transferee pursuant to Section 1.6 hereof).

. 4	
-	

(b) Unresolved Claims for which Parent has objected in accordance with Section 2.1 shall be administered in accordance with Section

<u>2.1(c)</u>.

(c) Upon the expiration of the twenty (20) calendar day objection period for any Unresolved Claims for which no Dispute Notice has been delivered, Parent and the HL Representative shall submit a Forfeiture Instruction to the Escrow Agent setting forth the portion of the Claim not objected to by Parent, and the Escrow Agent shall release such number of Escrow Shares for forfeiture and cancellation as set forth in the Forfeiture Instruction, pursuant to Section 2.1(b).

(d) After the resolution of each Unresolved Claim, any remaining portion of the Escrow Account not released for cancellation and forfeiture and not subject to other Unresolved Claims shall be released promptly thereafter by the Escrow Agent to the Company Shareholders in accordance with a Joint Release Instruction.

2.3 <u>Court Order</u>. Notwithstanding any other provision in this Agreement to the contrary, the Escrow Agent shall release the Escrow Account (or any portion thereof) in accordance with a notice from either Parent or the HL Representative of a Final Determination, along with a copy of the order, pursuant to which such court has determined whether and to what extent either the Escrow Shares shall be forfeited and cancelled or the Company Shareholders are entitled to the Escrow Shares, as applicable. If the Final Determination determines the dollar value of the Claim but does not calculate the number of Escrow Shares to be forfeited and cancelled, Parent and the HL Representative shall deliver a Forfeiture Instruction together with the Final Determination showing the number of Escrow Shares to be forfeited and cancelled from the accounts maintained on behalf of each Company Shareholder.

2.4 <u>Claims in Excess of Escrow</u>. If at any time during the term of this Agreement the number of Escrow Shares required to be released for cancellation and forfeiture pursuant to <u>Section 2.1</u> exceeds the number of Escrow Shares remaining in the Escrow Account, the Escrow Agent shall release the entire Escrow Account for cancellation and forfeiture and the rights of the HL Indemnitees to indemnification under Article XI the Business Combination Agreement shall be satisfied in full and extinguished.

**2.5** <u>Call Back Authorized Individuals</u>. In the event a Joint Release Instruction is delivered to the Escrow Agent, whether by e-mail, facsimile, mail, courier or otherwise, the Escrow Agent is authorized to seek confirmation of such instruction by telephone call back to the person or persons designated in Exhibits B-1 and/or B-2 annexed hereto (the "<u>Call Back Authorized Individuals</u>"), and the Escrow Agent may rely upon the confirmations of anyone purporting to be a Call Back Authorized Individual. To ensure accuracy of the instructions it receives, the Escrow Agent may record such call backs. If the Escrow Agent is unable to verify the instructions, or is not satisfied with the verification it receives, it will not execute the instruction until all such issues have been resolved. The persons and telephone numbers for call backs may be changed only in writing actually received and acknowledged by the Escrow Agent.

#### 2.6 Certain Definitions.

(a) "<u>Business Day</u>" means any day that is not a Saturday, a Sunday, or other day on which commercial banks located in New York, New York or Dublin, Ireland are obligated or authorized by applicable law to remain closed for business.

(b) "<u>Fair Market Value</u>" of the Escrow Shares means a price per share equal to the five-day trailing average of the mean of the high and low trading prices of Parent Class A Ordinary Shares on the Nasdaq Capital Market as of the five trading days immediately preceding the satisfaction of such Losses.

(c) "<u>Final Determination</u>" means a final non-appealable order of any court of competent jurisdiction which may be issued in respect of a Claim, together with (i) a certificate of the prevailing Escrow Party(ies) to the effect that such judgment is final and non-appealable and from a court of competent jurisdiction having proper authority and (ii) the written instructions of the prevailing Escrow Party(ies), provided that if the Final Determination directs the forfeiture and cancellation of Escrow Shares, Parent shall countersign the written instructions to confirm that the number of Escrow Shares to be forfeited and cancelled from each Company Shareholder's account has been verified by Parent.

(d) "Joint Release Instruction" means a joint written instruction of Parent and HL Representative, which is executed by Parent and HL Representative, to the Escrow Agent directing the Escrow Agent to release all or a portion of the Escrow Account, as applicable.

(e) "<u>Person</u>" means any individual, general or limited partnership, firm, corporation, limited liability company, association, joint stock company, trust, joint venture, unincorporated organization, or other entity, including a Governmental Authority or any department, agency, or political subdivision thereof.

(f) "<u>Unresolved Claims</u>" means, as of the Escrow Termination Date, the aggregate dollar amount of all Claims that are the subject of a Dispute Notice that have not previously been resolved or satisfied in accordance herewith, or were otherwise properly and timely asserted under this Agreement but unsatisfied as of the Escrow Termination Date, including any Claims for which a Notice of Claim has been delivered but for which the twenty (20) calendar day objection period has not expired as of the Escrow Termination Date.

#### Section 3. Fees and Expenses.

The Escrow Agent shall be entitled to receive, from time to time, fees in accordance with <u>Schedule 1</u>. In accordance with <u>Schedule 1</u>, the Escrow Agent will also be entitled to reimbursement for reasonable and documented out-of-pocket expenses incurred by the Escrow Agent in the performance of its duties hereunder and the execution and delivery of this Agreement. All fees owed to the Escrow Agent pursuant to this Agreement shall be paid for by the Parent.

#### Section 4. Limitation of Escrow Agent's Liability.

**4.1** <u>Duties and Limitation of Liability</u>. The Escrow Agent undertakes to perform such duties as are specifically set forth in this Agreement only and shall have no duty under any other agreement or document, and no implied covenants or obligations shall be read into this Agreement against the Escrow Agent. The Escrow Agent shall incur no liability with respect to any action taken by it or for any inaction on its part in reliance upon any notice, direction, instruction, consent, statement or other document believed by it in good faith to be genuine and duly authorized, nor for any other action or inaction except for its own gross negligence or willful misconduct. In all questions arising under this Agreement and/or its interpretation hereof in conjunction with the Business Combination Agreement, the Escrow Agent may rely on the advice of counsel, and for anything done, omitted or suffered in good faith by the Escrow Agent based upon such advice the Escrow Agent shall not be liable to anyone. In no event shall the Escrow Agent be liable for incidental, punitive or consequential damages.

**4.2** Indemnity. Parent will indemnify the Escrow Agent and its officers, directors, employees and agents for, and hold it and them harmless against, any loss, liability or expense (including attorney fees) incurred by the Escrow Agent without gross negligence or willful misconduct on the part of the Escrow Agent (or its officers, directors, employees or agents), arising out of or in connection with the Escrow Agent's carrying out its duties hereunder excluding any tax imposed on or calculated by reference to the net income receivable by the Escrow Agent. This right of indemnification shall survive the termination of this Agreement and the resignation of the Escrow Agent.

#### Section 5. Termination.

This Agreement shall terminate upon the release by the Escrow Agent of all of the Escrow Shares held in the Escrow Account in accordance with Section 2.

#### Section 6. Successor Escrow Agent.

In the event the Escrow Agent becomes unavailable or unwilling to continue as escrow agent under this Agreement, the Escrow Agent may resign and be discharged from its duties and obligations hereunder by giving its written resignation to the Escrow Parties. In such event, the Escrow Parties may, by mutual agreement, appoint a successor Escrow Agent. If the Escrow Parties fail to appoint a successor Escrow Agent within thirty (30) calendar days after receiving the Escrow Agent's written resignation, the Escrow Agent shall have the right to apply to a court of competent jurisdiction for the appointment of a successor Escrow Agent. The Escrow Agent's resignation will take effect upon the appointment of a successor Escrow Agent. The successor Escrow Agent shall execute and deliver to the Escrow Agent an instrument accepting such appointment, and the successor Escrow Agent shall, without further action, be vested with all the estates, property rights, powers and duties of the predecessor Escrow Agent as if originally named as Escrow Agent herein. The Escrow Agent shall act in accordance with written instructions from the Escrow Parties as to the deposit of the Escrow Shares in the Escrow Account with a successor Escrow Agent.

#### Section 7. Miscellaneous.

7.1 <u>Attorneys' Fees</u>. In any action at law or suit in equity to enforce or interpret this Agreement or the rights of any of the Escrow Parties or Escrow Agent hereunder (including disputes relating to claims for releases from the Escrow Account), the prevailing Party in such action or suit shall be entitled to recover its reasonable attorneys' fees and all other reasonable costs and expenses incurred in connection with such action or suit. Notwithstanding the foregoing, neither the HL Representative nor the Company Shareholder Representative shall have any personal responsibility for any such costs and expenses incurred by him or it in such representative capacity. The HL Representative and/or the Company Shareholder Representative may make a claim against the Escrow Account for any costs and expenses incurred by him or it, as applicable.

7.2 Notices. All notices, requests, demands, claims, and other communications hereunder will be in writing. Any notice, request, demand, claim, or other communication hereunder shall be deemed duly given (a) when delivered in person, (b) when delivered after posting in the United States mail having been sent registered or certified mail return receipt requested, postage prepaid, (c) when delivered by FedEx or other nationally recognized overnight delivery service or (d) when e-mailed during normal business hours (and otherwise as of the immediately following Business Day), addressed as follows:

if to HL or the HL Representative, to:

HL Acquisitions Corp. 499 Park Avenue, 12th Floor New York, NY 10022 Attention: Jeffrey E. Schwarz E-mail: jschwarz@metrocap.net

with a copy to:

Graubard Miller The Chrysler Building 405 Lexington Avenue, 11<sup>th</sup> Floor New York, New York 10174 Attention: David Alan Miller / Jeffrey M. Gallant E-mail: dmiller@graubard.com / jgallant@graubard.com

if to Parent, the Company, or the Company Shareholder Representative to:

Fusion Welcome – Fuel, S.A. Ex-Siemens Facilities Rua da Fabrica, S/N, Sabugo 2715-376, Almargem do Bispo Portugal Attention: Frederico Figueira de Chaves Email: frederico@keyfh.com

with a copy to:

Feinberg Hanson LLP 855 Boylston Street, 8th Floor Boston, Massachusetts 02116 Attention: David H. Feinberg, Esq. Email: dfeinberg@feinberghanson.com

If to Escrow Agent:

Continental Stock Transfer & Trust Company 1 State Street, 30<sup>th</sup> Floor New York, New York 10004 Attention: Account Administration Telephone: (212)845-4000 Facsimile: (212) 509-5150 E-mail: accountadmin@continentalstock.com

Any Party may change the address to which notices, requests, demands, claims, and other communications hereunder are to be delivered by giving the other Parties notice in the manner herein set forth. Notwithstanding the foregoing, notices addressed to the Escrow Agent shall be effective only upon receipt. If any notice or other document is required to be delivered to the Escrow Agent and any other Person, the Escrow Agent may assume without inquiry that notice or other document was received by such other Person on the date on which it was received by the Escrow Agent.

7.3 <u>Headings</u>. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.

7.4 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts (including by means of electronic mail or facsimile), each of which shall be deemed an original but all of which together will constitute one and the same instrument. Facsimile and pdf signatures shall be treated as original signatures for all purposes hereunder.

7.5 <u>Governing Law</u>. This Agreement and any claim, controversy or dispute arising out of or related to this Agreement, any of the transactions contemplated hereby, the relationship of the Parties, and/or the interpretation and enforcement of the rights and duties of the Parties, whether arising in contract, tort, equity or otherwise, shall be governed by and construed in accordance with the domestic Laws of the State of New York.

**7.6** Waiver of Jury Trial. EACH ESCROW PARTY AND THE ESCROW AGENT WAIVES ITS RESPECTIVE RIGHTS TO A TRIAL BY JURY OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OR RELATED TO THIS AGREEMENT IN ANY ACTION, PROCEEDING OR OTHER LITIGATION OF ANY TYPE BROUGHT BY ANY OF THE PARTIES HERETO AGAINST ANY OTHER PARTY OR ANY AFFILIATE OF ANY OTHER SUCH PARTY, WHETHER WITH RESPECT TO CONTRACT CLAIMS, TORT CLAIMS OR OTHERWISE. EACH ESCROW PARTY AND THE ESCROW AGENT AGREES THAT ANY SUCH CLAIM OR CAUSE OF ACTION SHALL BE TRIED BY A COURT TRIAL WITHOUT A JURY. WITHOUT LIMITING THE FOREGOING, EACH ESCROW PARTY AND THE ESCROW AGENT FURTHER AGREES THAT ITS RESPECTIVE RIGHT TO A TRIAL BY JURY IS WAIVED BY OPERATION OF THIS SECTION AS TO ANY ACTION, COUNTERCLAIM OR OTHER PROCEEDING WHICH SEEKS, IN WHOLE OR IN PART, TO CHALLENGE THE VALIDITY OR ENFORCEABILITY OF THIS AGREEMENT OR ANY PROVISION HEREOF. THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT.

7.7 <u>Succession and Assignment</u>. This Agreement shall be binding upon and shall inure to the benefit of each of the Parties hereto and each of their respective permitted successors and assigns, if any.

**7.8** <u>Amendments and Waivers</u>. Except for amendments to <u>Exhibit A</u> by the Escrow Agent in accordance with <u>Section 1.6</u> of this Agreement, no amendment of any provision of this Agreement shall be valid unless the same shall be in writing and signed by the Escrow Agent and each Escrow Party. No waiver by any Party hereto of any provision of this Agreement or any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, shall be valid unless the same shall be in writing and signed by the Party making such waiver nor shall such waiver be deemed to extend to any prior or subsequent default, misrepresentation, or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.

7.9 <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction.

7.10 No Third-Party Beneficiaries. Except as expressly provided herein, this Agreement shall not confer any rights or remedies upon any Person other than the Parties hereto and their respective successors and permitted assigns.

7.11 Entire Agreement. This Agreement and the Business Combination Agreement set forth the entire agreement among the Parties hereto relating to the subject matter hereof and supersede any prior understandings, agreements, or representations by or among the Parties hereto, written or oral, to the extent they relate in any way to the subject matter hereof. In the event of a conflict between this Agreement and the Business Combination Agreement, the Business Combination Agreement shall govern.

**7.12** <u>Cooperation</u>. The Escrow Parties shall cooperate fully with each other and the Escrow Agent and to execute and deliver such further documents, certificates, agreements, stock powers and instruments and to take such other actions as may be reasonably requested by an Escrow Party or the Escrow Agent to evidence or reflect the transactions contemplated by this Agreement and to carry out the intent and purposes of this Agreement.

### 7.13 Construction.

(a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neutral genders; the feminine gender shall include the masculine and neutral genders; and the neutral gender shall include masculine and feminine genders.

(b) The Parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement.

(c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."

(d) Except as otherwise indicated, all references in this Agreement to "Sections" and "Schedules" are intended to refer to Sections of this Agreement and Schedules to this Agreement.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the Parties hereto have duly caused this Agreement to be executed as of the day and year first above written.

#### PARENT: FUSION FUEL GREEN PLC

By: /s/ Frederico Figueira de Chaves Name: Frederico Figueira de Chaves Title: Director

COMPANY:

FUSION WELCOME – FUEL, S.A.

By: /s/ Frederico Figueira de Chaves Name: Frederico Figueira de Chaves Title: Director

<u>COMPANY SHAREHOLDER REPRESENTATIVE:</u> FUSION WELCOME, S.A.

By: /s/ João Teixeira Wahnon Name: João Teixeira Wahnon Title: Director

<u>HL:</u>

HL ACQUISITIONS CORP.

By: /s/ Jeffrey E. Schwarz Name: Jeffrey E. Schwarz Title: Chief Executive Officer

HL REPRESENTATIVE:

/s/ Jeffrey E. Schwarz Name: Jeffrey E. Schwarz

ESCROW AGENT:

CONTINENTAL STOCK TRANSFER & TRUST COMPANY, a New York corporation

By: /s/ Ana Gois Name: Ana Gois Title: Vice President

### STOCK ESCROW FEE SCHEDULE

Continental Stock Transfer & Trust ("CSST") will act as stock escrow agent as follows:

CSST will hold the shares (either via certificate or book entry) for the term of the escrow specified in the Escrow Agreement.

### Fees:

Review, execution and set up of the escrow and related required documents – one time fee	\$ 1,500
Annual fee to act as Escrow Agent (payable up front)	\$ 2,500

Additional services that may arise during the escrow term, such as voting or distributions will be billed upon appraisal of the service. Out-of-pocket charges will be billed at cost.

# <u>Exhibit A</u>

Escrow	
Shares	Company Shareholder
159,374	Fusion Welcome, S.A.
8,075	Falcfive, LDA
8,075	Numberbubble, S.A.
16,363	Magno Efeito, S.A.
20,613	Key Family Holding Investimentos e Consultoria de Gestão, LDA
212,500	TOTAL
-	